Depressive symptoms in the HIV and Hepatitis C co-infected population in Canada - An investigation in the second-generation direct acting antiviral era (2013-2020)

Gayatri Jayant Marathe, MSc, ScM

Department of Epidemiology, Biostatistics and Occupational Health

McGill University

Montreal, Quebec, Canada

April 2022

A thesis submitted to McGill University in partial fulfilment of the requirements of the

degree of Doctor of Philosophy

© Gayatri Jayant Marathe 2022

TABLE OF CONTENTS

TABLE	OF CONTENTS II
ABSTR	ACTVI
RÉSUM	IÉIX
ACKNC	WLEDGEMENTSXII
STATE	MENT OF FINANCIAL SUPPORTXV
CONTR	IBUTION OF AUTHORSXVI
STATE	MENT OF ORIGINALITYXVIII
LIST OF	TABLESXIX
LIST OF	FIGURESXXI
LIST OF	ABBREVIATIONSXXIII
1. CH	APTER 1: GENERAL INTRODUCTION 1
1.1.	Introduction1
1.2.	Research objectives
1.3.	Organization of Thesis4
2. CH	APTER 2: LITERATURE REVIEW 6
2.1.	Epidemiology of HIV infection6
2.2.	Natural history of HIV infection7
2.3.	Epidemiology of HCV infection9
2.4.	Natural history of HCV infection

4	2.5.	ΗIV	-HCV Co-infection - Epidemiology and impact on natural history	12
4	2.6.	HC	V Treatment evolution	13
4	2.7.	Dep	pression - Epidemiology, Screening, Diagnosis and Treatment	15
	2.8.	Dep	pression and HIV-HCV infection	20
2	2.9.	Dep	pressive symptoms and HCV treatment	21
3.	CH	APTI	ER 3: DATA SOURCES	24
	3.1.	Car	nadian Co-infection Cohort	24
	3.1.	1.	Participant recruitment	24
	3.1.	2.	Data collection	25
	3.2.	Foo	od Security and HIV-HCV co-infection study	26
	3.2.	1.	Participant recruitment	26
	3.2.	2.	Data collection	27
	3.3.	Ethi	ics approval	27
4.	CH	ΑΡΤΙ	ER 4: PREDICTION OF DEPRESSIVE SYMPTOMS PRESENCE IN TH	ΗE
CA	NADI	AN (CO-INFECTION COHORT	29
	4.1.	Pre	face to Manuscript 1	29
	4.1.	1.	Rationale	29
	4.1.	2.	Application	31
4	4.2.	Mar	nuscript 1: Predicting the presence of depressive symptoms in the HIV-H0	CV
(co-infe	ected	d population in Canada using supervised machine learning	33
4	4.3.	Mar	nuscript 1: Appendix	68

5.	СН	APTER 5: EFFECT OF DEPRESSIVE SYMPTOMS ON HCV TREATMENT
IN	ΙΤΙΑΤ	ION
	5.1.	Preface to Manuscript 2
	5.2.	Manuscript 2: Depressive symptoms are no longer a barrier to HCV treatment
	initiati	on in the HIV-HCV co-infected population in Canada
	5.3.	Manuscript 2: Appendix 128
6.	СН	APTER 6: IMPACT OF HCV CURE ON THE PRESENCE OF DEPRESSIVE
S١	/MPT	OMS
	6 1	Preface to Manuscript 3 136
	6.2	Manuscript 3: Impact of HCV cure on depressive symptoms in the HIV-HCV co-
	infact	ed population in Canada
	6 2	Manuacrint 2: Appandix
	0.3.	
7.	СН	APTER 7: EFFECT OF DEPRESSIVE SYMPTOMS ON HEALTH SERVICES
U	FILIZA	ATION
	7.1.	Preface to Manuscript 4180
	7.2.	Manuscript 4: Effect of depressive symptoms on health services utilization in the
	HIV a	nd Hepatitis C co-infected population in Canada
8.	СН	APTER 8: DISCUSSION
	8.1.	Summary of findings
	8.2.	Strengths and limitations
	8.3.	Implications and future directions
	8.4.	Conclusions

REFERENCES	1
------------	---

ABSTRACT

Depression is common in the human immunodeficiency virus (HIV)-Hepatitis C virus (HCV) co-infected population due to direct neurocognitive effects of HCV and HIV and/or associated stigma, discrimination, and socioeconomic burdens. HCV treatment has evolved from interferon (IFN)-based regimens to safer direct acting antivirals (DAA). Depression, a major IFN side effect, is therefore no longer a relative contraindication to HCV treatment thereby broadening access. However, it is not known if depressive symptoms will continue to have an effect on treatment initiation and if curing HCV could improve depressive symptoms.

The overall goal of this thesis was to investigate depressive symptoms in the Canadian HIV-HCV co-infected population in the second-generation DAA era (2013-2020). I used data from the Canadian HIV-HCV Co-Infection Cohort (CCC), an open multicentre prospective cohort with biannual follow-up visits. Demographic, behavioural, and clinical data were collected by standardized questionnaires and chart reviews. I also used data from the Food Security (FS) and HIV-HCV co-infection sub-study of the CCC. In the FS sub-study, depression screening was performed using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10). The scale assesses presence and severity of depressive symptoms in the past one week; scores ≥10 are indicative of being at risk for major depression.

The first objective was to predict the presence of depressive symptoms using supervised machine learning in the CCC, as depression screening was performed only in the FS sub-

VI

study. Candidate predictors (137) collected from the CCC and CES-D-10 data from the FS sub-study were used to develop a random forest algorithm to predict the CES-D-10 classes (\geq 10 vs <10) at each CCC visit. The algorithm had excellent classification performance with an area under the curve of 0.82. In the next objectives, I used these predicted depressive symptoms classes as a measure of depressive symptoms, correcting for misclassification using predictive value-based record-level correction.

In the second objective, I assessed the effect of depressive symptoms on time to HCV treatment initiation during the IFN (2003-2011) and DAA (2013-2020) eras using marginal structural Cox proportional hazards models with inverse weighting for death. The treatment initiation rate increased from 9 to 21 per 100 person-years from over the IFN to the DAA era. Treatment initiation was 19% lower among those with depressive symptoms than among those without in the IFN era, and 19% higher in DAA era.

In the third objective, I investigated the impact of sustained virologic response (SVR) or HCV cure on depressive symptoms using an interrupted time series analysis. The pre-SVR trend suggested depressive symptoms changed little over time, with no immediate level change. The post-SVR trend showed a reduction of 5% per year in depressive symptoms risk. This decline over time may reflect changes in biological pathways and/or better general health.

In the fourth objective, I examined the relationship between depressive symptoms, SVR and Health Services Utilization (HSU) using a zero-inflated negative binomial regression

VII

model. SVR was associated with 24% fewer inpatient visits. Among those who achieved SVR, no association was observed between depressive symptoms and in- or outpatient visits. However, among those who did not achieve SVR, inpatient visits were 16% higher and outpatient visits were 5% higher among those with depressive symptoms than among those without. Thus, depressive symptoms were associated with a modest increase in HSU; SVR led to attenuation of this effect.

In conclusion, a high proportion of the co-infected population had baseline CES-D-10 scores \geq 10, suggesting that depression management should be an integral part of HIV/HCV care. The doors to HCV elimination have opened in Canada in the DAA era. This body of work provides evidence that depressive symptoms are no longer a barrier to treatment initiation and that HCV cure leads to benefits beyond liver disease outcomes including reduction in depressive symptoms and HSU. This provides further rationale for treating HCV in all chronically infected persons.

RÉSUMÉ

La dépression est fréquente chez les personnes co-infectées au VIH et au virus de l'hépatite C (VHC), en raison d'effets neurocognitifs directs des deux virus et/ou de la stigmatisation, de la discrimination et du fardeau socio-économique associés à la coinfection VIH-VHC. Le traitement du VHC a évolué, passant de régimes à base d'interféron (IFN) à des antiviraux à action directe (AAD) plus sûrs. La dépression, un effet secondaire majeur de l'IFN, n'est donc plus une contre-indication relative au traitement du VHC, ce qui en élargit l'accès. Cependant, on ignore si les symptômes dépressifs continueront à avoir un effet sur l'initiation du traitement et si la guérison du VHC pourrait améliorer les symptômes dépressifs.

L'objectif général de cette thèse était d'étudier les symptômes dépressifs dans la population canadienne co-infectée aux VIH-VHC à l'ère des AAD de deuxième génération (2013-2020). J'ai utilisé les données de la cohorte canadienne de co-infection VIH-VHC (CCC), une étude prospective multicentrique ouverte avec des visites de suivi semestrielles. Les données démographiques, comportementales et cliniques ont été recueillies par questionnaires standardisés et examens de dossiers. J'ai également utilisé les données d'une sous-étude de la CCC portant sur la sécurité alimentaire (SA). Dans cette sous-étude, le dépistage de la dépression a été effectué à l'aide de l'échelle de dépression 10 du *Center for Epidemiologic Studies* (CES-D-10). Cette échelle évalue la présence et la sévérité des symptômes dépressifs au cours de la dernière semaine. Des scores ≥10 indiquent un risque de dépression clinique.

Le premier objectif était de prédire la présence de symptômes dépressifs au sein de la CCC en utilisant l'apprentissage automatique supervisé, car le dépistage de la dépression n'a été effectué que dans la sous-étude SA. Les prédicteurs candidats (137) recueillis dans la CCC et les données CES-D-10 de la sous-étude SA ont été utilisés pour développer un algorithme de forêt aléatoire afin de prédire les classes CES-D-10 (≥10 vs <10) à chaque visite de la CCC. L'algorithme a montré une excellente performance de classification avec une aire sous la courbe de 0,82. Dans les objectifs suivants, j'ai utilisé les classes de symptômes dépressifs prédites, et corrigé les erreurs de classification au niveau de l'observation en fonction des valeurs prédictives.

Dans le deuxième objectif, j'ai évalué l'effet des symptômes dépressifs sur le délai d'initiation du traitement contre le VHC pendant les périodes IFN (2003-2011) et AAD (2013-2020) en utilisant des modèles de risques proportionnels de Cox structurels marginaux avec pondération inverse pour le décès. Le taux d'initiation du traitement est passé de 9 à 21 pour 100 années-personnes entre l'ère IFN et l'ère AAD. Le taux d'initiation était inférieur de 19% chez les personnes présentant des symptômes dépressifs par rapport à celles qui n'en présentaient pas à l'ère IFN, et supérieur de 19% à l'ère AAD.

Dans le troisième objectif, j'ai étudié l'impact d'une réponse virologique soutenue (RVS) -ou guérison du VHC– sur les symptômes dépressifs en utilisant une analyse de séries chronologiques interrompues. La tendance pré-RVS suggère que les symptômes dépressifs ont peu changé dans le temps, sans rupture immédiate. Les tendances post-

Х

SVR ont montré une réduction de 5% par an du risque de symptômes dépressifs. Cette diminution dans le temps peut refléter des changements dans les processus biologiques et/ou une meilleure santé générale.

Dans le quatrième objectif, j'ai examiné la relation entre les symptômes dépressifs, la RVS et l'utilisation de services de santé (USS) à l'aide d'un modèle de régression binomial négatif à inflation de zéros. La RVS était associée à une diminution de 24 % des visites de patients hospitalisés. Parmi ceux ayant développé une RVS, aucune association n'a été observée entre les symptômes dépressifs et les visites en milieu hospitalier ou ambulatoire. Cependant, parmi ceux n'ayant pas développé de RVS, le nombre de visites en milieu hospitalier était supérieur de 16% et le nombre de visites en milieu ambulatoire supérieur de 5% chez les personnes présentant des symptômes dépressifs étaient associés à une augmentation modeste de l'USS ; la RVS a permis d'atténuer cet effet.

En conclusion, une proportion élevée de la population co-infectée présentait des scores CES-D-10 de base ≥10. La prise en charge de la dépression devrait faire partie intégrante des soins VIH/VHC. L'élimination du VHC est devenue possible au Canada à l'ère des AAD. Ces travaux fournissent des preuves que les symptômes dépressifs ne sont plus un obstacle au traitement et que la guérison entraîne des bénéfices au-delà de l'amélioration des fonctions hépatiques, y compris la réduction des symptômes dépressifs et de l'USS. Ces résultats soutiennent l'importance du traitement du VHC chez toutes les personnes chroniquement infectées.

XI

ACKNOWLEDGEMENTS

The PhD program at McGill University has been an incredible journey. I am beyond grateful to all the people involved in making this the most fruitful and engaging time of my life. First and foremost, I would like to thank my supervisors, Dr. Marina Klein, and Dr. Erica Moodie. Marina and Erica, I am deeply appreciative of your constructive approach to teaching and thank you for being the kindest supervisors I could ever hope for. Your mentorship will be invaluable for the rest of my career as a researcher. I would like to thank my thesis committee member, Dr. Marie-Josée Brouillette, for all your advice throughout the project. I would also like to thank my professors from the Department of Epidemiology, Biostatistics and Occupational Health (EBOH) for helping me figure out epidemiology.

Next, I would like to thank the participants of the Canadian Co-infection Cohort (CCC) without whom none of this research would have been possible. I would like to thank the CCC co-investigators and CCC project team, who helped collect this data and provided help with any data queries, especially, Jessica Lumia, Isabelle Robichaud, Hansi Peiris, Leo Wong, and Shouao Wang. I would also like to thank Chantal Burelle from the McGill University Health Centre and Katherine Hayden and André Yves Gagnon from EBOH for all the administrative support and help throughout.

I would like to thank the Canadian Network on Hepatitis C (CanHepC) for funding my PhD through the PhD trainee scholarship and providing me with the opportunity to meet and

XII

learn from the leading Hepatitis C researchers in Canada, and Norma Choucha for the administrative support through the CanHepC training program.

To my 2017 PhD cohort - you were all by my side from the beginning, have shared in all my homework, coding and overall grad school frustrations and have been the most supportive group. Thanks so much!!

I do not think I can thank my Montreal friend bubble enough for getting me through this PhD and the pandemic - Carla, Charlotte, Kaija, Vajini, Vanessa, Herak, and Matthieu. You have been a blessing!! Thanks for not only being the best study and work buddies, but also exploring the incredible city of Montreal and its food with me!! Kaija, thanks for being the best friend, roommate, and my Netflix binge watch partner over the last four years. Carla and Charlotte, I couldn't have done any of this without your friendship, support, and all the laughs.

Last but not the least, I want to thank my family and friends (who are basically family). Manasi and Tanvi, thanks for cheering me up and cheering me on daily through the past 15 years and being my sounding board always. My dear DBM family for making me laugh every day by being your weird, bizarre, but absolutely loving selves. Shout out to Madhura tai, Gaurav dada and Harsh for always being there for me. My dear niece, Siya, for filling my heart with more love than I thought I could have for anyone. Finally, Aai and Baba - I don't think I have enough words to thank you for always believing in me, letting me dream big, and being the biggest pillars of support even when I changed my mind a million times through the past 10 years.

This thesis would not have been possible without the encouragement from not only all the people mentioned above, but many more who may have remained unmentioned, but I am truly grateful for! Thank you so much!

STATEMENT OF FINANCIAL SUPPORT

I am extremely grateful for the financial support I received during my doctoral training from the Department of Epidemiology, Biostatistics and Occupational Health, McGill University in 2017, my supervisor Dr. Marina Klein from 2018 to 2020, the Research Institute of McGill University Heath Centre through a PhD Studentship in 2020, and the Canadian Network on Hepatitis C (CanHepC) through a PhD trainee fellowship from 2020 to 2022. Registration and travel costs for workshops and conferences were provided from funding from Dr. Klein and CanHepC. I also received a scholarship to participate in the International AIDS Society Conference on HIV science (IAS) 2021, and the new investigator scholarship to attend the virtual Conference on Retroviruses and Opportunistic Infections (CROI) 2022. The data used for the four thesis manuscripts were funded by Fonds de recherche du Québec - Santé; Réseau sida/maladies infectieuses, the Canadian Institute for Health Research (CIHR; FDN-143270); and the CIHR Canadian HIV Trials Network (CTN222 & CTN264).

CONTRIBUTION OF AUTHORS

Manuscript 1: Predicting the presence of depressive symptoms in the HIV-HCV coinfected population in Canada using supervised machine learning.

Authors: Gayatri Marathe, Erica EM Moodie, Marie-Josée Brouillette, Joseph Cox, Curtis Cooper, Charlotte Lanièce Delaunay, Brian Conway, Mark Hull, Valérie Martel-Laferrière, Marie-Louise Vachon, Sharon Walmsley, Alexander Wong, Marina B. Klein; Canadian Co-Infection Cohort.

Manuscript 2: Depressive symptoms are no longer a barrier to HCV treatment initiation in the HIV–HCV co-infected population in Canada.

Authors: Gayatri Marathe, Erica E. M. Moodie, Marie-Josée Brouillette, Joseph Cox, Charlotte Lanièce Delaunay, Curtis Cooper, Mark Hull, John Gill, Sharon Walmsley, Neora Pick, Marina B. Klein, Canadian Co-Infection Cohort.

Manuscript 3: Impact of HCV cure on depressive symptoms in the HIV-HCV coinfected population in Canada.

Authors: Gayatri Marathe, Erica EM Moodie, Marie-Josée Brouillette, Charlotte Lanièce Delaunay, Joseph Cox, Valérie Martel-Laferrière, John Gill, Curtis Cooper, Neora Pick, Marie-Louise Vachon, Sharon Walmsley, Marina B. Klein; Canadian Co-Infection Cohort.

Manuscript 4: Effect of depressive symptoms on health services utilization in the HIV and Hepatitis C co-infected population in Canada.

Authors: Gayatri Marathe, Erica E. M. Moodie, Marie-Josée Brouillette, Charlotte Lanièce Delaunay, Marina B. Klein; Canadian Co-Infection Cohort.

For all manuscripts, Gayatri Marathe, Erica E. M. Moodie, and Marina B. Klein conceived of and designed the study. Gayatri Marathe and Charlotte Lanièce Delaunay prepared the analytical dataset. Gayatri Marathe performed all statistical analyses, with input from Erica E.M. Moodie, Marina B. Klein, and Marie-Josée Brouillette. Gayatri Marathe drafted each manuscript. All authors revised manuscripts critically and gave final approval prior to submission.

Marina B. Klein is the principal investigator of the Canadian Co-infection Cohort (CCC). Erica E. M. Moodie, Joseph Cox, Curtis Cooper, Brian Conway, Mark Hull, John Gill, Neora Pick, Valérie Martel-Laferrière, Marie-Louise Vachon, Sharon Walmsley, and Alexander Wong are CCC co-investigators. The investigators provided substantive support, and/or recruited, and followed participants of the CCC. The data were extracted by Leo Wong and Shouao Wang from the CCC data management team, and Jessica Lumia (database coordinator) provided support in case of queries and assisted with data verification.

XVII

STATEMENT OF ORIGINALITY

The research presented in this thesis constitutes an original contribution in the study of depressive symptoms in the HIV and HCV co-infected population in Canada. All four manuscripts present novel findings related to depressive symptoms measurement and their association with patient outcomes. This research also provides new insights on long-term effects of HCV cure on depression in the DAA era, which will have implications on treatment strategies and mental health management in the Canadian co-infected population.

Specifically, Manuscript 1 generated the first supervised machine learning algorithms for depressive symptoms classification using available predictor data in the CCC, which were then used as a measure of depressive symptoms for the next thesis objectives. Additionally, when externally validated, these algorithms can have applications beyond the CCC. Manuscript 2 provided the first quantification of the effect of depressive symptoms on HCV treatment initiation and provided a comparison between the IFN and DAA eras. Manuscript 3 was the first longitudinal analysis describing the long-term trends of depressive symptoms post-SVR in the co-infected population. Finally, Manuscript 4 assessed the effect of depressive symptoms on health services utilization and first noted that SVR may have a modifying effect.

While I received guidance from my supervisors, committee member and co-authors on the substantive, methodological, and statistical aspects of this thesis, I declare that the conception, execution, and drafting of the work in this thesis are my own.

XVIII

LIST OF TABLES

Table 4.1: Candidate predictors used in the random forest algorithms 51
Table 4.2: Baseline characteristics of participants in the study sample (n=717)
Table 4.3: Performance evaluation in the primary analysis 53
Table 4.4: Comparison of performance evaluation measures for primary and additional
analyses
Table 4.5: Supplementary table - List of candidate predictors
Table 4.6: Supplementary table - Final RF hyperparameters for the algorithms
Table 4.7: Supplementary table - Participant characteristics in visit 1 in the study sample
(n=717) as compared to baseline visit for all CCC participants (n=2008) 89
Table 5.1: Baseline characteristics for participants included from IFN and DAA eras. 115
Table 5.2: Effect of depressive symptoms on time to HCV treatment initiation in IFN and
DAA eras
Table 5.3: Secondary analyses using measured CES-D-10 classes in a restricted subset
and baseline predicted depressive symptoms 117
Table 5.4: Supplementary table - Performance evaluation measures for the RF algorithm
Table 5.5: Supplementary table - Summary of the inverse probability weights 133
Table 6.1: Baseline characteristics of the included participants (n = 470) 155
Table 6.2: Impact of SVR on depressive symptoms in the HIV-HCV co-infected population
- primary analysis models with and without outcome misclassification correction (n=470)

Table 6.3: Sensitivity analysis to assess possible lead time bias - models with and without
outcome misclassification correction (n=332) 156
Table 6.4: Supplementary table - Performance evaluation measures for the RF algorithm
Table 6.5: Supplementary table - Model selection using QIC
Table 6.6: Supplementary table - Subgroup analyses - models without outcome
misclassification correction
Table 6.7: Supplementary table - Sensitivity analysis model for DAA non-responders177
Table 7.1: Effect of depressive symptoms and SVR on HSU in the HIV-HCV co-infected
population; all IRRs are relative to those without depressive symptoms and without (or
prior to) SVR 194

LIST OF FIGURES

Figure 2.1: Natural history of HIV infection	. 8
Figure 2.2: Evolution of HCV treatment and SVR rates	13
Figure 2.3: Model for BDI and CES-D factor structure	18
Figure 2.4: Pathways of inflammation-induced neuropsychiatric symptoms	20
Figure 3.1: Participants enrolled in the CCC from 2003 to 2022	24
Figure 4.1: Receiver Operating Characteristic (ROC) curve for the A. Full algorithm (x	(=
137) and B. Reduced algorithm (x=46)	54
Figure 4.2: Predictor importance plots: A. Full algorithm and B. Reduced algorithm	55
Figure 4.3: Supplementary figure - Calibration graphs - A. Full algorithm ($x = 137$) and	В.
Reduced algorithm (x = 46)	88
Figure 5.1: Directed Acyclic Graph for effect of depressive symptoms on HCV treatme	ent
initiation1	18
initiation	18 era
initiation	18 era 19
initiation	18 era 19 57
initiation	18 era 19 57 he
initiation	18 era 19 57 he 58
initiation 1 Figure 5.2: Flowcharts of participants included in the analytical samples - A. IFN et (2003-2011) and B. DAA era (2013-2020) 1 Figure 6.1: Flowchart of participants included in the analytical sample 1 Figure 6.2: Impact of sustained virologic response (SVR) on depressive symptoms in th 1 HIV-HCV co-infected population 1 Figure 6.3: Supplementary figure - Causal Diagram 1	18 era 19 57 he 58 72
initiation 1 Figure 5.2: Flowcharts of participants included in the analytical samples - A. IFN e (2003-2011) and B. DAA era (2013-2020) (2003-2011) and B. DAA era (2013-2020) 1 Figure 6.1: Flowchart of participants included in the analytical sample. 1 Figure 6.2: Impact of sustained virologic response (SVR) on depressive symptoms in the 1 HIV-HCV co-infected population 1 Figure 6.3: Supplementary figure - Causal Diagram 1 Figure 6.4: Supplementary figure - Impact of non-response on depressive symptoms 1	18 9ra 19 57 he 58 72 in
initiation 1 Figure 5.2: Flowcharts of participants included in the analytical samples - A. IFN et (2003-2011) and B. DAA era (2013-2020) 14 Figure 6.1: Flowchart of participants included in the analytical sample 14 Figure 6.2: Impact of sustained virologic response (SVR) on depressive symptoms in th 14 HIV-HCV co-infected population 14 Figure 6.3: Supplementary figure - Causal Diagram 17 Figure 6.4: Supplementary figure - Impact of non-response on depressive symptoms 17 DAA non-responders 17	18 era 19 57 he 58 72 in 77
initiation 17 Figure 5.2: Flowcharts of participants included in the analytical samples - A. IFN et (2003-2011) and B. DAA era (2013-2020) 17 Figure 6.1: Flowchart of participants included in the analytical sample 18 Figure 6.2: Impact of sustained virologic response (SVR) on depressive symptoms in th 18 HIV-HCV co-infected population 18 Figure 6.3: Supplementary figure - Causal Diagram 17 Figure 6.4: Supplementary figure - Impact of non-response on depressive symptoms 17 Figure 7.1: Distribution of number of self-reported inpatient and outpatient visits in the 17	18 9ra 19 57 he 58 72 in 77 he

Figure 8.1: Patient perceived barriers and facilitators to DAA treatment uptake 207

LIST OF ABBREVIATIONS

- AIDS: Acquired Immunodeficiency Syndrome
- APRI: Aspartate Aminotransferase to Platelet Ratio Index
- ART: Antiretroviral Therapy
- AST: Aspartate Aminotransferase
- AUC: Area under the curve
- **BDI: Beck Depression Inventory**
- BMI: Body Mass Index
- CAD: Canadian dollar
- cART: Combination Antiretroviral Therapy
- CCC: Canadian Co-infection Cohort
- CES-D: Center for Epidemiologic Studies Depression Scale
- CI: Confidence interval
- COVID-19: Coronavirus Disease 2019
- PH: Proportional Hazards
- DAA: Direct Acting Antivirals
- DAG: Directed Acyclic Graph
- DALY: Disability Adjusted Life Years
- DNA: Deoxyribonucleic acid
- EQ-5D: Euro-QoL 5-Dimension scale
- FS: Food Security
- GBD: Global Burden of Diseases, Injuries, and Risk Factors Study
- gbMSM: Gay and Bisexual Men who have Sex with Men

GEE: Generalized Estimating Equations HCV: Hepatitis C virus HIV: Human Immunodeficiency Virus HR: Hazard ratio HSU: Health Services Utilization IDU: Injection Drug use IFN: Interferon IL: Interleukin **IPCW:** Inverse Probability Censoring Weights **IPTW:** Inverse Probability Treatment Weights IQR: Interquartile Range IRR: Incidence rate ratios **ITS: Interrupted Time Series** LR-: Negative Likelihood Ratio LR+: Positive Likelihood Ratio LSD: Lysergic Acid Diethylamide MDA: Methylenedioxyamphetamine MSCM: Marginal Structural Cox Proportional Hazards Model MSM: Marginal Structural Model **NPV: Negative Predictive Value** OOB: Out-of-Bag PCP: Phenylcyclohexyl piperidine PHQ: Patient Health Questionnaire

PLWH: People living with HIV

PPV: Positive Predictive Value

- PWID: People Who Inject Drugs
- QIC: Quasi-likelihood under the Independence model Criterion

RF: Random Forest

- RMSE: Root Mean Squared Error
- RNA: Ribonucleic acid
- ROC: Receiver operating characteristic curve
- SD: Standard Deviation
- SES: Socio-economic Status
- SVR: Sustained Virologic Response
- TNF: Tumour Necrosis Factor
- WHO: World Health Organization
- ZINB: Zero-Inflated Negative Binomial Model

1. CHAPTER 1: GENERAL INTRODUCTION

1.1. Introduction

Globally, an estimated 37.7 million people live with Human Immunodeficiency Virus (HIV) and 71.1 million people live with a chronic Hepatitis C Virus (HCV) infection (1, 2). With the availability of antiretroviral therapy (ART) for HIV, people living with HIV (PLWH) can now live healthy, long, and productive lives, if they are adherent to treatment. However, PLWH still have to manage several co-morbid conditions including liver disease and mental illness. Due to the common modes of transmission, co-infection of HIV and HCV is common, with approximately 2.3 million people co-infected worldwide (2).

Depression is a major psychiatric co-morbidity related to both HIV and HCV and both biological and psychosocial mechanisms are at play. The reported prevalence of depression is as high as 20-30% among PLWH and almost 24% among those with chronic HCV infection (3, 4).. Depression prevalence is reported to be even higher in the co-infected population, due to the co-existence of risk factors and neuropsychiatric effects of both HIV and HCV (5).

Depression was a well described side effect of the earlier interferon (IFN)-ribavirin based HCV antiviral treatments, with studies showing more than 20% of those treated developed major depression (6). This often prevented IFN therapy from being prescribed to individuals with current or past psychiatric illness. However, after 2013, second-generation direct acting antiviral (DAA) regimens were introduced, which were safer and in real-world settings had >95% rates of cure (sustained virologic response (SVR)), even

in HIV-HCV co-infected persons (2). Importantly, there was no evidence of any significant psychiatric side effects, leading to HCV treatment guidelines being updated (7, 8). Thus, current, or past psychiatric illness is no longer a relative contraindication for HCV treatment.

The overall safety, simplicity, and high cure rates of the second generation DAA regimens have led to improved treatment access and development of elimination goals for HCV worldwide and within Canada. A study in the Canadian Co-infection cohort (CCC) suggested an almost threefold increase (from 8 to 28 per 100 person-years (PY)) in treatment initiation in the cohort after 2013; however in this study, it was also observed that treatment initiation rates were lower among people who actively inject drugs, women, and Indigenous populations compared with other co-infected people (9). Thus, despite the advantages of the new regimens, treatment access is still not uniform across subpopulations, posing a threat to HCV elimination goals. Patient- and system-level barriers to treatment and linkage to care still exist; the presence of depression could be one such barrier. In fact, the presence of significant depressive symptoms could have an impact on patient outcomes, even when not meeting diagnostic criteria for major depression. However, there is a lack of evidence on effect that depressive symptoms may still have on treatment initiation.

Furthermore, the changes in prescribing practices provide us with the opportunity to assess the potential impact of HCV treatment on depressive symptoms over time. We may expect lower depressive symptoms post-cure through biological pathways due to

HCV viral clearance. However, co-infected populations continue to face challenges including discrimination, socioeconomic burdens, and/or substance use. These could impact psychosocial and health outcomes such as overdoses and suicide, mitigating the benefits of HCV cure. It is therefore important to examine longitudinally whether HCV cure leads to a change in the level of depressive symptoms and moreover whether this change persists over time.

Finally, depression and both HIV and HCV infections are associated with increased health services utilization (HSU) (10, 11). There is evidence of high economic burden on health systems due to mental health disorders, HIV, HCV, and related risk factors including substance use (12, 13). Thus, HCV cure could impact this outcome due to possible overall improvements in mental as well as physical well-being. The effect of depressive symptoms on HSU and how it is affected by HCV cure requires investigation in the vulnerable HIV-HCV co-infected population.

1.2. Research objectives

The overall goal of my doctoral thesis was to investigate depressive symptoms in the HIV-HCV co-infected population in Canada.

The specific objectives were as follows:

1. To development an algorithm to predict the presence of depressive symptoms in participants of the CCC.

- 2. To examine the effect of depressive symptoms on HCV treatment initiation in the HIV-HCV co-infected population in Canada during the IFN (2003-2011) and secondgeneration DAA (2013-2020) eras.
- **3.** To evaluate the impact of SVR after HCV treatment on the presence of depressive symptoms in the HIV-HCV co-infected population in Canada during the second-generation DAA era (2013-2020).
- **4.** To examine the effect of depressive symptoms HSU in the HIV-HCV co-infected population in Canada during the second-generation DAA era (2013-2020).

1.3. Organization of Thesis

The format of the thesis is manuscript-based. It includes four original manuscripts, corresponding to the four thesis objectives. Chapter 2 provides the relevant thesis background and specific literature review to support the thesis objectives. Chapter 3 provides a detailed introduction to the data sources used in this thesis, including the CCC and its sub-study, the food security and HIV-HCV co-infection sub-study (FS sub-study). Chapter 4 comprises of Manuscript 1, which describes the development of supervised machine learning algorithms for predicting the presence of depressive symptoms in the HIV-HCV co-infected population in Canada. Chapter 5 contains Manuscript 2, which examines the effect of depressive symptoms on HCV treatment initiation in the IFN (2003-2011) and second-generation DAA (2013-2020) eras. Chapter 6 is Manuscript 3, which evaluates the impact of HCV cure on the presence of depressive symptoms in the HIV-HCV co-infected population in Canada.

examines the effect of depressive symptoms on HSU in the Canadian HIV-HCV coinfected population. Chapter 8 summarizes the key findings of the thesis, discusses the strengths and limitations, the implications of this work and future directions, and provides concluding remarks. The references for each manuscript are included in the corresponding chapters. The references for chapters 1, 2, 3, 8, and the manuscript prefaces are included at the end of this thesis.

2. CHAPTER 2: LITERATURE REVIEW

2.1. Epidemiology of HIV infection

The World Health Organization (WHO) estimated 37.7 million people were living with HIV worldwide in 2020, with Sub-Saharan Africa most severely affected and accounting for more than two-thirds of the PLWH (25.4 million) globally (1). In Canada, the number of PLWH was estimated to be around 62,000 (range 54,600-70,500) in 2018; an estimated 13% of these individuals were unaware of their infection (14). Globally, there has been a 29% decline in the number of new HIV infections from 2.1 million in 2010 to 1.5 million in 2020. That same year, there were 680,000 HIV-related deaths, representing a decrease of 47% from 2010 (1). Thus, although HIV continues to be a major global health concern, there have been major improvements over time.

The main reason for these improvements was the development of the ART regimens. The first medication for HIV, zidovudine, was available for use by 1987 and in 1995, the first protease inhibitor was approved for use (15). Treatment for HIV has since evolved over time and more than twenty-five drugs have been developed. Combination ART (cART) consists of a combination of drugs from different classes targeting the HIV life cycle with the aim of stopping HIV replication and preserving or restoring immune function (16). Although a cure for HIV is not yet available, this combination of drugs taken every day can maintain an undetectable viral load, which has been shown to prevent transmission (17). Guidelines recommend treating patients as soon as they are diagnosed and adherence to treatment leads to improved patient health outcomes and survival (18, 19). PLWH can thus live healthy, long, and productive lives if they are adherent to treatment.

In Canada, of those who were diagnosed, an estimated 45,000 people (85%) were on ART in 2018, 94% of whom had a suppressed viral load (14).

HIV is transmitted by body fluids via three main routes: sexual transmission through semen and vaginal secretions; mother-to-child transmission during pregnancy, delivery or through breastmilk; and parenteral transmission via contaminated blood and blood products, accidental needle sticks, and sharing contaminated needles (1, 15). The key populations who are disproportionately affected by HIV include gay and bisexual men who have sex with men (gbMSM), people who inject drugs (PWID), people in prisons and other closed settings, sex workers, and transgender people (1). In Canada, 50% of all new HIV infections were among gbMSM and 14% were in PWID; Indigenous communities showed an increase in HIV incidence from 2016 to 2018, making up 14% of all new infections in 2018 despite accounting for only 4.9% of the Canadian population (14, 20).

2.2. Natural history of HIV infection

HIV is a single-stranded positive-sense RNA virus that belongs to the family *Retroviridae* and genus *Lentivirus* (21). HIV infects the CD4⁺ cells and macrophages, which are part of the human immune system. HIV causes depletion of the CD4⁺ cells, which makes the host immunocompromised and thus vulnerable to opportunistic infections. Additionally, viral enzymes including reverse transcriptase and integrase help the HIV genome to integrate into the human DNA leading to the characteristic persistent infection (22). Figure 2.1 shows the natural history of HIV infection, as depicted by Melhuish et. al. (21). The

first stage of HIV infection is the primary infection, in which an acute illness occurs in 50% of those exposed to HIV within 2-5 weeks (15). The clinical symptoms are flu-like with fever, sore throat, malaise, nausea, diarrhoea, and rash. As seen in figure 2.1, during the primary infection the viral load increases and peaks and there is lowering of the CD4⁺ cell count. In between 4-6 weeks after exposure, seroconversion occurs and antibodies against HIV are detectable (21). During the next asymptomatic stage of the disease after seroconversion, viral load declines and remains stable for as long as 6-8 years without treatment. During this time the CD4⁺ cell count gradually declines. Without ART, the CD4+ cells continue to decline, which leads to opportunistic infections and other AIDS-defining conditions including tuberculosis, pneumocystis jirovecii pneumonia, cryptococcal meningitis, Kaposi's sarcoma, non-Hodgkin's lymphoma, and Burkitt's lymphoma, among others (21).



Change in HIV viral load and CD4 count after infection

Ab, antibody; Ag, antigen.

Figure 2.1: Natural history of HIV infection

Reproduced from Medicine, 46 (6), Melhuish A. and Lewthwaite P., Natural history of HIV and AIDS, 356-361., Copyright (2018), with permission from Elsevier

However, with the availability of ART, the natural history of HIV infection has changed dramatically. As the drugs act on inhibiting viral replication, the plasma viral load is reduced to undetectable levels, which helps in replenishing the host CD4⁺ cells. A recent study suggested that ART initiation at high CD4⁺ cell counts was associated with a longer life span and more comorbidity-free years for individuals with HIV infection (23). However, as PLWH on ART are aging, they now have to deal with several co-morbid conditions like cardiovascular disease, chronic kidney disease, liver disease, osteoporosis, cancer and neurocognitive disorders including depression and anxiety (24). Thus, study of these co-morbid conditions is necessary to further improve the quality of life of PLWH.

2.3. Epidemiology of HCV infection

Approximately 71.1 million people are living with HCV globally according to WHO estimates and there were 1.75 million new HCV infections in 2015, representing an incidence rate of 23.7 per 100,000 (2). The five countries, Pakistan, Nigeria, Egypt, India, and Russia, account for more than 50% of the infections worldwide (25). However, in 2015, HCV incidence was highest in the European and Eastern Mediterranean WHO regions (26). In Canada, approximately 250,000 people are estimated to be living with a chronic HCV infection, with the largest numbers in Ontario, British Columbia, and Quebec (27). HCV is known to have high genetic diversity and has eight known genotypes distributed globally; majority of infections in North America including Canada are of genotype 1 followed by 3 (26).

HCV is a bloodborne virus and is transmitted via small quantities of blood (2). The main modes of transmissions differ between low/middle and high income countries (28). Injection drug use is the major driver of HCV transmission in high-income countries (29). Transmission occurs via sharing of contaminated needles as well as other drug use paraphernalia and thus harm-reduction services like needle and syringe programmes and opioid substitution therapy are key for HCV prevention (30). In low- and middle-income countries, transmission occurs via unsafe medical practices and iatrogenic transmission through unscreened blood and blood products in addition to injection drug use (31). Other less common sources of infection include sexual and mother-to-child transmission (32).

The Canadian HCV epidemic is called the "twin epidemics" - one among young people, especially those who inject drugs, and the other among baby boomers (those born between 1945-75) (29). Studies have shown that baby boomers are five times more likely to be affected by detrimental effects of chronic HCV than the general population and many in this population were infected possibly via nosocomial or iatrogenic transmission by e.g., reuse of needles in health care settings and via donated blood and organs (33, 34). HCV affects certain key populations disproportionately and the "Canadian blueprint for HCV elimination (2019)" identified priority populations that need to be targeted for HCV elimination in Canada: PWID, Indigenous peoples, people with experience in prisons, immigrants and newcomers from countries with high rates of HCV, and gbMSM (29). Estimates suggest that up to 85% of new infections in Canada occur among PWID and prevalence of HCV is almost five times higher in Indigenous people compared to non-Indigenous people (29, 35, 36).

2.4. Natural history of HCV infection

HCV is a positive-sense RNA virus belonging to the family Flaviviridae and genus Hepacivirus (37). HCV causes both acute (short-term) and chronic (long-term) infection. Only about 25% of those infected spontaneously clear the virus after the acute phase (~6 months); the remaining 75% progress to chronic infection (38). In the natural history of this disease, approximately 10-15% of the people exposed to HCV have a symptomatic infection and 85-90% remain asymptomatic (39). Spontaneous clearance is higher among those with symptomatic infection compared those with asymptomatic infection (39). Other factors associated with spontaneous clearance include female sex, and younger age at infection among others. In terms of clinical manifestations, among those with acute HCV infections, the symptoms are non-descript and include malaise, right upper quadrant pain, and nausea (40). Among those with chronic HCV infection, data suggest that, over 20-30 years, approximately 10-20% develop advanced fibrosis, 10% develop cirrhosis, and 1-5% develop hepatocellular carcinoma (39, 41). HCV infection is also known to be associated with a number of extrahepatic manifestations, including type-2 diabetes, chronic kidney disease, B-cell lymphoma, and psychiatric manifestations like depression (42).

HCV RNA is detectable approximately 1 to 3 weeks after exposure to HCV. Anti-HCV antibodies are detectable from between 6-12 weeks from exposure and then they remain positive for life. The viral loads fluctuate initially during the first months, and, in the case
of spontaneous clearance, HCV RNA becomes undetectable approximately 6 months from infection. HCV virus clearance (either spontaneously or through treatment) does not provide immunity against future infection and no vaccine is currently available. Thus, the risk of re-infection remains. This genetic diversity of HCV has had an impact on both treatment and vaccine development (43).

2.5. HIV-HCV Co-infection - Epidemiology and impact on natural history

Due to the common modes of transmission, approximately 2.3 million people worldwide are co-infected with HCV and HIV (2). Of these, it is estimated that more than half (about 1.3 million) are PWID (44). Followed by PWID, the highest prevalence of HCV co-infection is among gbMSM. The odds of HCV infection was found to be six times higher in PLWH vs. those without HIV infection (44). In Canada, an estimated 15-25% of the 70,000 HIVinfected persons are co-infected with HCV (45). In addition to PWID, HIV-HCV coinfection is disproportionately seen in the Indigenous population and people incarcerated in the Canadian provincial and federal prison systems (46).

The natural history of both HIV and HCV are impacted due to co-infection. Given CD4+ T cells are key in raising the antiviral immune response against HCV, co-infection with HIV compromises these immune responses (47). There is evidence of impaired immune response among people who have cleared HCV previously, indicative of overall impaired immunity against HCV (48), even among people who have cleared an HCV infection. HIV infection can reduce the chance of spontaneous clearance of HCV as well (47, 49). In

general, chronic HCV infection with HIV leads to accelerated progression to serious hepatic conditions including liver cirrhosis and hepatocellular carcinoma (2).

Similarly, when co-infected with HCV, the natural history of HIV is affected. In the pre-ART era, HCV was shown to lead to increased immune activation which was associated with immune responses being impaired and also a rapid progression to AIDS (47). There is evidence of reduced CD4+ T cells restoration among PLWH on ART due to co-infection with HCV, as the liver is involved in T cell homeostasis (47).

2.6. HCV Treatment evolution

HCV treatment has evolved over time, from IFN-based regimens with or without ribavirin to IFN-free DAA regimens (50, 51). Treatment is deemed successful if a patient achieves SVR, that is, HCV viral load is undetectable 12 weeks after the end of treatment. SVR equates to a cure. SVR rates have improved over time, with each new treatment regimen, as seen in figure 2.2 by Webster et. al. 2015 (50, 51). In the early 1990s, the IFN-based



Reproduced from The Lancet, 385 (9973), Webster, D. P., Klenerman, P., Dusheiko, G. M., Hepatitis C, 1124-35, Copyright (2015), with permission from Elsevier.

treatment regimens became available for HCV. These regimens had many issues including longer regimens (24-48 week) with weekly injections and poor outcomes (<10% SVR rates) (50). The pegylated-IFN regimens with the addition of ribavirin available by the late 1990s improved the response rates (30-40%), however the regimens had many side effects including psychiatric illness.

By 2011, the first-generation DAAs were developed. These included boceprevir and telaprevir, which improved SVR rates to up to 75% with triple therapy (i.e., when combined with peg-IFN); however, these were limited to genotype 1 and still had many side effects (50). Thus, efforts to find better alternatives led to the development of the second-generation DAAs which include drugs from various classes including, NS3/4A protease inhibitors (e.g., Simeprevir), NS5A inhibitors (e.g., daclatasvir), non-nucleoside NS5B polymerase inhibitors (e.g., sofosbuvir). These IFN-free DAA-based regimens, when used in combination, e.g., sofosbuvir/ledipasvir (8-12 weeks), sofosbuvir/velpatasvir (12 weeks) and glecaprevir/pibrentasvir (8 weeks), achieve SVR rates of >95% in real world settings (50). These regimens are shorter, between 8-12 weeks, have fewer side effects and the latter two are pan-genotypic, vastly simplifying HCV treatment (2).

Among PLWH, SVR rates were even lower when treated for HCV with IFN and ribavirin regimens (<30%); thus, the co-infected population was considered a special treatment group (52). However, with the advent of the IFN-free second-generation DAAs regimens, SVR rates among those co-infected with HIV-HCV improved to over 95%, as seen in the

HCV mono-infected population (53). Guidelines, therefore, now recommend all HIV coinfected patients be treated similarly to HCV mono-infected, with careful consideration of drug interactions with ART regimens (46).

These advances in treatment have consequently opened the doors to elimination of HCV. In 2016, the World Health Assembly, the decision-making body of WHO, adopted the global health sector strategy on viral hepatitis 2016-20, with a goal to eliminate HCV as a public health threat by 2030 (2).

2.7. Depression - Epidemiology, Screening, Diagnosis and Treatment

Major depressive disorder or depression is described in the fifth edition of the Diagnostic and Statistical Manual of Mental Disorders (DSM-5) as presence of at least five of the nine listed symptoms, and at least one of these five has to be either a depressed mood or loss of all interest and pleasure for the same two-week period. The remaining symptoms are appetite or weight disturbance, sleep disturbance, psychomotor agitation or slowing, fatigue or loss of energy, abnormal inappropriate guilt, poor concentration, and thoughts of death or suicide (54, 55).

Under these DSM-5 criteria, a study suggested that approximately 227 combinations of symptoms could lead to a depression diagnosis (56). Due to these many possible symptom combinations, the clinical courses of depression are very varied, especially with symptoms with opposing behaviours for example, increased vs decreased appetite. Thus,

depression is described as a heterogeneous mental disorder (57, 58). Studies also show that the majority of people with depression have a chronic-recurrent course of illness; the lifetime prevalence estimates vary widely in literature across countries (59-61). The biological mechanism of depression includes several neurotransmitter systems and previous research has shown that the serotonergic (serotonin (5-HT)) and nor-epinephrine/noradrenergic systems are conclusively involved (57, 62, 63). The age of onset is typically in the early to mid 20s, however onset has a large age range (59). There are various other sociodemographic characteristics associated with occurrence of depression including sex, marital status, education, income, and employment (59, 60). Additional predictors include life events, social support, psychosocial impairment, and a number of clinical factors such as previous episodes, severity of episodes, and co-morbid conditions (60).

According to the Global Burden of Diseases, Injuries, and Risk Factors Study (GBD) 2019, depressive disorders were one of the leading causes of burden worldwide (ranked as the 13th leading cause of disability adjusted life years (DALY)), with prevalence estimates comparatively higher than many other diseases (64). A recent study estimated that the COVID-19 pandemic led to an additional 53 million cases of major depressive disorder globally (27.6% increase), increasing the total prevalence to 3,153 cases per 100,000 population (65). In Canada, a study in working age Canadian adults found the average annual prevalence of major depressive episodes to be 5.4% among those employed and as high as 11.7% among unemployed participants (66). The GBD 2019

estimates that there are approximately 804,700 prevalent cases of major depressive disorder in Canada (64).

Many tools have been developed to screen for and make a diagnosis of major depression. The diagnostic tools include structured or semi-structured psychiatric assessments and structured diagnostic interviews (54). The gold standard for diagnosis is generally considered to the structured clinical interview for DSM-5 (SCID-5), and different versions are available for clinical and research use.

In addition, a number of tools for identifying potential cases by assessing presence and severity of depressive symptoms are available. Some commonly used depression screening tools are self-rated, such as the 2-item and 9-item Patient Health Questionnaires (PHQs), the 20- or 10-item Center for Epidemiologic studies Depression scale (CES-D) and Beck Depression Inventory (BDI, while others are clinician-rated, such as the Hamilton Depression rating scale, among others (54). These tools, which are widely used in both clinical practice and research, were developed for a variety of purposes: for example, the CES-D-20 scale was developed to screen for depression in the general population, while the BDI was originally developed to assess severity of depressive symptoms in individuals with diagnosed depression (67).

A large number of studies in psychology literature have focused on understanding the constructs that screening tools measure. The Encyclopedia Britannica defines a psychological construct as a "label for a cluster or domain of covarying behaviours" (68).

The underlying constructs of depression cannot be measured directly and thus it is important to consider the validity, reliability, and other psychometric properties of the different tools. A study by Skorikov et al. in 2003 compared the factor structure for two scales, the CES-D-20 and BDI, in order to assess if they measure the same underlying latent construct; they concluded that "the BDI and CES-D represent related, but theoretically distinct and empirically different, aspects of the same higher order latent variable, corresponding to the general construct of depression", as represented in figure 2.3 (67). The figure shows the different components of depression measured by the CES-D versus the BDI scales.



Figure 2.3: Model for BDI and CES-D factor structure

Reproduced from Educational and Psychological Measurement Skorka VB, VanderVoort DJ. Relationships Between The Underlying Constructs Of The Beck Depression Inventory And The Center For Epidemiological Studies Depression Scale.63(2):319-335 (2003)

It is very important to consider when using screening tools that one may not measure the same underlying construct as another tool developed to measure the severity of depressive symptoms in someone with a clinical diagnosis of depression, and their use in place of a formal diagnosis may lead to different estimates than the ones reached following a diagnostic process. It is essential to recognize that screening tools are not used for clinical diagnosis of depression, and presence of depressive symptoms based on pre-established cut-offs on screening tools is not to equivalent to a clinical diagnosis of depression (69, 70). Additionally, these tools should not also be used as the basis for prescribing antidepressants, as this could lead to possible overtreatment (71).

The screening tools are, however, useful for case-finding, and can help guide detection and assessment of severity (54, 69). These tools are frequently used by clinicians and researchers to identify those at risk of depression, as these symptoms are along the spectrum for major depression. The presence of these symptoms will thus have an impact on different serious outcomes in patients, even before a diagnosis of depression is reached. This assessment can help clinicians recognise segments of the population at risk and intervene with tailored policies to improve outcomes. It is these symptoms that this thesis will focus on.

Both psychotherapy and medications are part of the available treatments for depression. Medications for depression belong to several classes which include selective serotonin reuptake inhibitors (SSRIs), serotonin-norepinephrine reuptake inhibitors (SNRIs), atypical antidepressants, and serotonin modulators. The older treatments include tricyclic antidepressants, and monoamine oxidase inhibitors (72). The use of a combination of medications would depend on patient tolerance due to several possible side effects; usually the use of both pharmacotherapy and psychotherapy is suggested (73).

2.8. Depression and HIV-HCV infection

There is evidence suggesting that HIV and HCV infections are associated with neuropsychiatric manifestations, mainly depression (3, 74). Studies have shown depression prevalence to be close to 20-30% among PLWH and up to 24% among those with chronic HCV (3, 4, 75). However, these estimates for prevalence vary widely, due to studies conducted in different sub-populations, with different diagnostic/screening methods, study sample sizes and clinical factors such as HIV stage (3, 4). Nevertheless, most studies have consistently shown higher occurrence (2-4-fold) of depression among PLWH compared to the general population. A meta-analysis suggests the prevalence of depression may be higher among HIV-HCV co-infected patients than both HIV and HCV mono-infected individuals respectively (5).



Figure 2.4: Pathways of inflammation-induced neuropsychiatric symptoms

Reproduced by permission from Springer Nature: Springer, Inflammation-associated depression: evidence, mechanisms, and implications, by Dantzer R. and Capuron L., Copyright (2017)

Depression mechanisms related to HIV and HCV are both biological and psychosocial. It

is possible that co-existence of risk factors for depression related to these infections may

lead to this increased vulnerability of the co-infected population to depressive symptoms. Among co-infected persons, ongoing substance use is a common additional risk factor, and may also be affected by presence of depressive symptoms (76). HIV and HCV both affect the central nervous system directly, where they induce an activation of the immune system (3, 77); the release of pro-inflammatory cytokines such as tumour necrosis Factor (TNF)- α and interleukin (IL)-1 alter the levels of neurotransmitters involved in the maintenance of mood, such as serotonin and dopamine, resulting in the development of depression (3, 77, 78). Some of the pathways of depressive symptoms induction related to inflammation are shown in figure 2.3 illustrated by Capuron and Castanon (2017) (79). There are also many known psychosocial pathways to depression including social stresses caused by stigma, discrimination, and lack of social and financial support due to HIV and HCV infections (3, 74).

2.9. Depressive symptoms and HCV treatment

Depression is a well described major side effect of the earlier IFN-ribavirin based HCV antiviral treatments, with some studies showing that more than 20% of those treated developed depression (6). In terms of stages of depression symptoms after IFN treatment, studies have shown that the first set of symptoms that occur in the first week of treatment include fatigue, lack of energy (anergia), lassitude, decreased motivation, motor slowing, reduced appetite, and altered sleep. While these somatic symptoms are among the diagnostic criteria for depression, they are not specific to depression and when they develop in response to the initiation of IFN- α , they do not respond to antidepressant

treatment (79). The second set of symptoms include sadness, decreased mood, reduced ability to experience pleasure (anhedonia), and impaired cognitive function. These symptoms are more specific to a diagnosis of depression and are more responsive to treatment than the somatic symptoms (79). There was also evidence for an increased risk of suicidal ideation associated with IFN treatment among those infected with HIV (80). Due to this, people with current or a history of psychiatric illness often were not prescribed IFN therapy (81). However, with the second-generation IFN-free DAA regimens, there is no evidence of significant psychiatric side effects during and immediately after DAA treatment (7, 82, 83). HCV treatment guidelines have thus been updated and current or past psychiatric illness is no longer a contraindication for treatment (84).

There has been a marked increase in treatment initiation after the second-generation DAAs were licensed in Canada in 2013. Treatment initiation rates increased almost threefold from 8 to 28 per 100 person-years in the CCC (9). However, the study showed that there were major disparities in the initiation rates among people who actively inject drugs, women, and Indigenous populations, with rates as low as 5-12% (9). Thus, in spite of the advantages of DAAs, treatment initiation rates are still not optimal. Patient- and system-level barriers may explain incomplete treatment uptake in these vulnerable populations; depression could be one such barrier. Additionally, there are a few studies that have assessed depressive symptoms before treatment and at SVR ascertainment (83, 85-88). Assessments of long-term impacts of SVR on depression beyond SVR are also scarce in the literature. Thus, further studies are important to understand the possible trends of depression before and after HCV cure. There could be a possible decline over

time via the biological mechanism, or there may not be a change considering the remaining psychosocial risk factors of depression in the HIV and HCV co-infected populations.

3. CHAPTER 3: DATA SOURCES

3.1. Canadian Co-infection Cohort

3.1.1. Participant recruitment

I used data from the Canadian HIV-Hepatitis C Co-Infection Cohort (CCC), which is a publicly funded open multicentre prospective cohort study, ongoing since 2003. The cohort is described in detail elsewhere (89). Briefly, the initial objective of the cohort study was to determine the effect of ARTs for HIV and HCV therapies





on liver disease progression in HIV-HCV co-infected populations and has since evolved to evaluate the impacts of DAA. The cohort serves as a significant resource for secondary analyses to better understand co-infection and related health outcomes. The participants are recruited from 18 urban and semi-urban HIV clinics from across six Canadian provinces, to reflect the Canadian epidemic. The cohort sites are the Montreal Chest Institute Immunodeficiency Clinic, Hôpital Notre-Dame, Montréal General Hôpital, and la clinique médical du Quartier Latin in Quebec; the British Columbia Centre for Excellence in HIV/AIDS, Oak Tree Clinic, Pender Community Health Centre, and Native Health Centre in British Columbia; the Southern Alberta HIV Clinic in Alberta; the Sunnybrook Hospital, Windsor Regional Hospital, McMaster University Medical Centre, Ottawa General Hospital, Sudbury Regional Hospital, and Toronto General Hospital in Ontario; the Queen Elizabeth Halifax Health Centre in Nova Scotia; the Regina General Hospital

and the University of Saskatchewan Hospital in Saskatchewan. The map in figure 3.1 shows distribution of participants enrolled across Canada from 2003 to 2022 (90).

The CCC inclusion criteria are: \geq 16 years of age, documented HIV infection and evidence of HCV infection (HCV RNA positive and/or HCV seropositive) (89). The serologic evidence of HCV exposure is defined by a positive HCV antibody serology with or without detectable HCV viremia measured via molecular reverse transcription-polymerase chain reaction (RT-PCR) assay. As of February 2022, the CCC has enrolled 2057 participants. The participants are predominately white (70%), male (70%), single (68%), aged 41-50 (42%), with no post-secondary education (50%) and high levels of poverty (monthly income of \$501-1000 (39%)). The cohort has a strong representation of Indigenous peoples (21%) (90). Of all recruited participants over time, 644 participants are currently active in the cohort and 437 participants have died. Very recently, overdose has surpassed end stage liver disease as a cause of death in the cohort, indicative of the scale and impact of the overdose epidemic in Canada (90, 91).

3.1.2. Data collection

After obtaining informed consent, sociodemographic and behavioural data are collected from participants by a standardized self-administered baseline questionnaire. Clinical data regarding HIV/HCV treatment histories and co-morbidities are collected during the baseline visit through medical chart reviews by the study coordinators. Blood samples are provided by participants for HIV/HCV testing - viral load, antibody and CD4 testing, and for basic laboratory tests including serum biochemistry, liver profiles, and blood counts.

The participants are then followed longitudinally, with visits every six months. The followup questionnaire collects updated demographic and behavioral information. Clinical data for the last six months are collected through medical chart review, including psychiatric diagnosis, medications and HCV treatment initiation and response data with exact start and stop dates for each regimen prescribed, and biological specimens are collected. Health Related-Quality of Life (HR-QoL) is a patient-reported measure collected at each visit using a standardized instrument, EQ-5D-3L (92, 93). It consists of a descriptive system (mobility, self-care, usual activities, pain/discomfort, anxiety/depression) and a current health visual analog scale. HSU is determined in the cohort by the number of selfreported inpatient, outpatient, specialist, emergency, and walk-in visits in the past 6 months.

3.2. Food Security and HIV-HCV co-infection study

3.2.1. Participant recruitment

I used data from a mixed methods study conducted within the CCC, the Food Security and HIV-HCV co-infection study (FS sub-study), conducted between 2012-2015 (94). The primary objectives were to understand the relationship between food security and behavioural and clinical factors related to HIV-HCV co-infection, HR-QoL, health and treatment outcomes among co-infected patients participating in HIV care in Canada (95). All CCC participants were invited to participate, and the study visits were integrated into the biannual CCC study visits. The FS sub-study recruited 725 CCC participants who were followed for up to 5 visits. The participants had a baseline median age of 49 years, 73% were male, 75% white, and 57% experienced moderate or severe food insecurity at baseline; the percentage of participants with food insecurity remained around 50% throughout follow-up (96).

3.2.2. Data collection

The FS sub-study used self-administered questionnaires, with an option of interviewer support for the participants. The questions were regarding food insecurity, general and mental health of the participants, ART adherence, health care and social service utilization, housing, and income (94). Food security in the past 6 months was measured using a 10-item adult scale of the Household Food Security Survey Module (HFSSM) (97). A categorical variable was created, with participants with 0-1, 2-5, or \geq 6 affirmative responses classified respectively as being food secure, moderately food insecure, or severely food insecure, as per the health Canada criteria. Depression screening was performed in the sub-study using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) (98), a 10-item Likert scale that assesses presence and severity of depressive symptoms in the past one week. Each item is measured on a 4-point scale, with reverse scoring for the 2 positive items and a total score range of 0-30. We dichotomized the score at 10 to create CES-D-10 classes (1/0), as a score \geq 10 is considered depressive symptoms indicative of high risk for major depression (99).

3.3. Ethics approval

The CCC and the FS Sub-Study were approved by the Research Ethics Board of the McGill University Health Centre (2006-1875, BMB-06-006t, 2013-994) and the research

ethics boards of participating institutions. The studies were conducted according to the Declaration of Helsinki. Informed consent was obtained from all individual participants included in the studies. The secondary analysis done for this thesis was approved by the Research Ethics Board of the McGill University Health Centre (2021-6985).

4. CHAPTER 4: PREDICTION OF DEPRESSIVE SYMPTOMS PRESENCE IN THE CANADIAN CO-INFECTION COHORT

4.1. Preface to Manuscript 1

4.1.1. Rationale

As described in the previous chapter, in the parent study, CCC, a major depression diagnosis was available for participants only if it was documented in medical charts reviewed at each visit every 6 months. Through the chart reviews, the CCC did have information about medications prescribed in every 6-month period, which included antidepressant medications. However, antidepressants are prescribed for a number of possible indications other than major depression and thus this alone could not be used as an indicator of major depression (100). Additionally, screening for depression was not part of the standardized CCC questionnaire. However, in the smaller sub-set of the participants (n=725) who were part of the FS sub-study that was carried out between 2012 and 2015, depression screening was performed using the CES-D-10 screening tool. Approximately 1200 participants were recruited into the CCC by end of 2012, and thus more than half of the participants were part of this sub-study. The population was found to be generalizable to the CCC, as shown in table 4.7 in the appendix for Manuscript 1.

There are multiple demographic, clinical, and behavioural characteristics that have been documented as risk factors for depression (101). The CCC collects extensive data regarding these characteristics in people co-infected with HIV and HCV at baseline as well as at follow-up visits. The study also collects participants' HR-QoL data and

medication history. These data can thus be used as predictors of presence of clinically relevant depressive symptoms.

There are several supervised and unsupervised machine learning techniques that enable accurate outcome predictions. These have been used to predict current or future onset, disease course, etc. for psychiatric disorders including depression, anxiety, and schizophrenia (102-104). Demographic and clinical data have been used to create depression prediction models in elderly and people with diabetes (105, 106). However, similar models have not been developed in people living with HIV-HCV co-infection.

In this manuscript, I developed Random Forest classification algorithms to predict presence of depressive symptoms using the depression screening data from the FS substudy and predictive variables from the CCC.

This manuscript is submitted for publication and is under review at BMC Medical Research Methodology.

This analysis was presented as posters and oral talks at the following conferences/seminars:

 Work in Progress seminar, Infectious Diseases, and Immunity in Global Health program, Research Institute of the McGill University Health Centre, Montreal, Canada - November 2019 - Oral presentation

- International Workshop on HIV and Hepatitis Observational Databases (IWHOD)
 2020, Sitges, Spain (Cancelled due to the COVID-19 pandemic) March 2020 Poster
- 16th Annual EBOSS Research Day, McGill University, Montreal, Canada March
 2020 Oral presentation
- 29th Annual Canadian Conference on HIV/AIDS Research, Virtual May 2020 Poster

4.1.2. Application

The algorithms developed in this manuscript were used to predict the CES-D-10 class at each CCC visit to be then used in the next objectives.

In Manuscripts 2 and 3, I used the CES-D-10 classes predicted by the full algorithm, which used all 137 candidate predictors. These predicted classes were then used as exposure measure in Manuscript 2 and outcome measure in Manuscript 3. In Manuscript 4, I used the CES-D-10 classes predicted by the reduced algorithm, with 46 predictors, as exposure. I chose not to use the full algorithm for prediction of exposure in this analysis, as the full algorithm included HSU related predictors.

In all three objectives, I used record-level correction in order to correct for exposure/outcome misclassification. I decided to use the predictive value-based record-level correction, as positive and negative predictive values were estimated for both the full and reduced algorithm in Manuscript 1. This method includes simulation of corrected

exposure at each visit by repeated Bernoulli trials with probability equal to PPV for those classified as CES-D-10 class=1 and 1-NPV for those classified as CES-D-10 class=0 (107).

4.2. Manuscript 1: Predicting the presence of depressive symptoms in the HIV-HCV co-infected population in Canada using supervised machine learning

Gayatri Marathe^{1, 3}, Erica EM Moodie¹, Marie-Josée Brouillette^{2, 3}, Joseph Cox^{1, 3}, Curtis Cooper⁴, Charlotte Lanièce Delaunay^{1, 3}, Brian Conway⁵, Mark Hull⁶, Valérie Martel-Laferrière⁷, Marie-Louise Vachon⁸, Sharon Walmsley⁹, Alexander Wong¹⁰, Marina B. Klein^{1, 3, 11}; Canadian Co-Infection Cohort³ *

- 1. Department of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada
- 2. Department of Psychiatry, McGill University, Montreal, Quebec, Canada
- 3. McGill University Health Center-Research Institute, Centre for Outcomes Research and Evaluation, Montreal, Quebec, Canada
- 4. Department of Medicine, University of Ottawa, Ottawa, Ontario, Canada
- 5. Vancouver Infectious Diseases Centre, Vancouver, British Columbia, Canada
- 6. Centre for Excellence in HIV/AIDS, St. Paul's Hospital, Vancouver, British Columbia, Canada
- Department of microbiology, infectious diseases and immunology, Centre de Recherche du Centre Hospitalier de l'Université de Montréal, Montreal, Quebec, Canada
- 8. Centre Hospitalier de l'Université Laval, Quebec City, Quebec, Canada
- 9. Toronto General Hospital Research Institute, Toronto, Ontario, Canada
- 10. Department of Medicine, University of Saskatchewan, Regina, Saskatchewan, Canada
- 11. CIHR Canadian HIV Trials Network (CTN), Vancouver, British Columbia, Canada

* A list of the cohort co-investigators and their affiliations appears at the end of the paper

Corresponding author:

Marina B. Klein

Division of Infectious Diseases and Chronic Viral Illness Service

McGill University Health Centre, 1001 Decarie Boulevard, D02.4110

Montreal, Canada H4A 3J1

Phone. 1-514-843-2090; Fax: 1-514-843-2092

Email address: marina.klein@mcgill.ca

Abstract

Background

Depression is common in the human immunodeficiency virus (HIV)-hepatitis C virus (HCV) co-infected population. Demographic, behavioural, and clinical data collected in research settings may be of help in identifying those at risk for major depression. We aimed to predict the presence of depressive symptoms indicative of a risk of depression and identify important classification predictors using supervised machine learning.

Methods

We used data from the Canadian Co-infection Cohort, a multicentre prospective cohort, and its associated sub-study on Food Security (FS). The Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) was administered in the FS sub-study; participants were classified as being at risk for major depression if scores \geq 10. We developed two random forest algorithms using the training data (80%) and 10-fold cross validation to predict the CES-D-10 classes - 1. Full algorithm with all candidate predictors (137 predictors) and 2. Reduced algorithm using a subset of predictors based on expert opinion (46 predictors). We evaluated the algorithm performances in the testing data using area under the receiver operating characteristic curves (AUC) and generated predictor importance plots.

Results

We included 1,934 FS sub-study visits from 717 participants who were predominantly male (73%), white (76%), unemployed (73%), and high school educated (52%). At the

first visit, median age was 49 years (IQR:43-54) and 53% reported presence of depressive symptoms with CES-D-10 scores \geq 10. The full algorithm had an AUC of 0.82 (95% CI:0.78-0.86) and the reduced algorithm of 0.76 (95% CI:0.71-0.81). Employment, HIV clinical stage, revenue source, body mass index, and education were the five most important predictors.

Conclusion

We developed a prediction algorithm that could be instrumental in identifying individuals at risk for depression in the HIV-HCV co-infected population in research settings. Development of such machine learning algorithms using research data with rich predictor information can be useful for retrospective analyses of unanswered questions regarding impact of depressive symptoms on clinical and patient-centred outcomes among vulnerable populations.

Keywords: HIV-HCV co-infection; Depressive symptoms; Supervised machine learning; Random forests

Background

With shared modes of transmission, co-infection of human immunodeficiency virus (HIV) and hepatitis C virus (HCV) is common, with approximately 2.3 million co-infected individuals worldwide [1] [2]. Depression is the most common neuropsychiatric manifestation among people living with HIV and those with chronic HCV. The prevalence of diagnosed major depression is 2-4-fold higher among people living with HIV than the general population and reported to be as high as 24% among those with chronic HCV infection [3, 4]. Potential biological mechanisms include direct infection of the central nervous system and peripheral immune responses which have been shown to induce depression [3, 5]. Psychosocial risk factors including stigma, discrimination, lack of support, and substance use have also been shown to be contributory [3, 5]. Studies report an even higher depression prevalence in the co-infected population, which may be due to the co-existence of risk factors [6].

The presence of significant depressive symptoms may have an impact on outcomes in patients, even in the absence of a major depression diagnosis; for example, the presence of depressive symptoms is associated with non-adherence to antiretroviral therapy among people living with HIV, which may lead to increased viral load and suppressed immune function [7, 8]. Screening tools can be used to assess presence and severity of depressive symptoms and identify those at risk for major depression [9, 10], permitting early intervention. Co-infected individuals often live with multiple co-morbid conditions and thus spend a considerable amount of time in healthcare settings [11]. Depression screening among HIV and HCV infected patients is seldom routinely performed during clinical assessments or in longitudinal cohort studies, despite the known high prevalence

of depression [12-14]. Multiple demographic, clinical and behavioural characteristics have been documented as risk factors for depression [15] and these data are generally collected in clinical and research cohorts. Thus, such data could be used retrospectively to predict the presence of depressive symptoms severe enough to be associated with negative health outcomes or an increased risk of being diagnosed with major depression and to follow this risk over time. Such measures will be useful for exploring important questions regarding depressive symptoms, their evolution and response to therapies in the co-infected population.

Machine learning includes robust techniques that enable accurate outcome predictions in medical research. In mental health research, machine learning has been used to predict current or future onset, disease course and treatment outcomes for psychiatric disorders including depression, anxiety, and schizophrenia [1, 16, 17]. A wide range of data sources have been used for developing these prediction algorithms including electronic medical records, neuroimaging, and social media. Demographic and clinical data have been used to create depression prediction models in the elderly and people with diabetes [18, 19]. However, similar models have not yet been developed in people living with HIV-HCV co-infection.

We leveraged a non-parametric supervised machine learning technique using cohort data to develop classification algorithms to predict the presence of depressive symptoms indicative of a risk for major depression and characteristics important for prediction of depressive symptoms in HIV-HCV co-infected individuals in Canada.

Methods

Data Sources and study sample

We used data from the Canadian HIV-HCV Co-Infection Cohort (CCC), an open multicenter prospective cohort study, ongoing since 2003 and an associated sub-study, the Food Security and HIV-HCV co-infection study (FS sub-study) [20, 21]. The CCC recruits from 18 HIV centers, both urban and semi-urban across six Canadian provinces (Quebec, British Columbia, Alberta, Ontario, Nova Scotia, and Saskatchewan). [20] Eligibility criteria include ≥16 years of age, documented HIV infection, and evidence of HCV infection (HCV RNA positive and/or HCV seropositive). The study had recruited 2018 participants as of July 2020. Participants are followed longitudinally, with follow-up visits every six months. Sociodemographic, behavioural, and health-related quality of life (HR-QoL) data are collected from participants by a standardized self-administered questionnaire at each visit. HR-QoL is measured using EuroQol-5 Dimension-3 Level (EQ-5D-3L) [22]. Clinical data including HIV/HCV treatment, co-morbidities, psychiatric diagnoses, and other medications are collected via medical chart reviews. Laboratory testing at each visit include HIV and HCV related tests, hematology, biochemistry, and liver profiles.

The FS sub-study is a mixed methods study conducted within the CCC between 2012 and 2015. All CCC participants were invited to participate and study visits were integrated into the biannual CCC visits. The FS sub-study recruited 725 participants and they were followed up for a maximum of 5 visits. The study collected data on food insecurity, general and mental health (including depression screening), treatment adherence and health care utilization using a self-administered questionnaire [21].

Depression screening was performed only in the FS sub-study; thus, the analytic sample in this study only included FS sub-study participant visits. The FS sub-study visits were merged with corresponding CCC visit data. As the two study visits for CCC and FS substudy were, on occasion, not on the same day, information from visits within 3 months of each other were considered 'concurrent'. We used three exclusion criteria to create the final study sample - 1) participant visits were excluded if no depression screening measure (see below) was available at that visit; 2) participant visits were excluded if there was no corresponding CCC visit (within the 3-month window), as the predictors used in this analysis were derived from the CCC; and 3) all visits for a participant were excluded if no data was available for a predictor in all of their study visits.

Outcome

Depression screening was conducted in the FS sub-study using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10), which is a shortened version of the CES-D-20 scale [23]. The CES-D-10 is a 10-item Likert scale questionnaire that assesses presence and severity of depressive symptoms in the past one week. Each item is measured on a 4-point scale, with reverse scoring for the 2 positive items and a total score range of 0-30. We dichotomized the score at 10 to create the CES-D-10 classes (1/0), as a score \geq 10 is widely considered for the presence of depressive symptoms indicative of high risk for major depression, hereafter referred to as depressive symptoms for brevity [23]. Both the scale and the dichotomization at 10 have been validated in HIV populations in Canada [24].

Predictors

We selected candidate predictors (x) from the CCC data based on the literature and subject matter expertise. We included predictors from five major categories: questions related to mental health, HR-QoL, sociodemographic, behavioural, and clinical characteristics; see Table 4.1. We selected a total of 137 candidate predictors, of which 136 were categorical and 1 was continuous (EQ-5D-3L - health state). From this list of candidate predictors, we selected a subset of predictors (x=46) that may be more regularly available in most research settings based on expert opinion. See Table 4.5 in Appendix A for an exhaustive list of candidate predictors (x=137) and their corresponding categories.

Statistical Analysis

Primary analysis

We assessed proportion of missing data for each predictor. For predictors with <5% missing data, we carried the value from the last visit forward. For predictors with $\geq 5\%$ missing data and when data were missing for a predictor for all visits for a participant, we used an additional category of "no response" for categorical variables, which we hypothesized could be informative in the prediction algorithm (See Table 4.5 in appendix A) since the participant decision to respond (or not) is itself potentially clinically informative.

We used the supervised machine learning technique of Random Forests (RF), an ensemble learning approach which uses bootstrap aggregation of multiple decision trees, combining predictions from these many trees [25]; see more details about RF in Appendix

B. We used probability machines to estimate the CES-D-10 class probabilities at each visit and then determined CES-D-10 class at default probability threshold [26]. We developed two RF algorithms - 1. Full algorithm: Using all candidate predictors (x = 137) and 2. Reduced algorithm: Using a selected subset of more commonly available predictors (x=46) based on expert opinion, which could be more generalizable to research studies beyond the CCC; see appendix A. We split the analytical sample into training and testing data, using the recommended 80:20 split, such that we had data for performance evaluation (testing) that was completely independent of data used for model development (training) and thus ensure an unbiased evaluation [27]. We performed the 80/20 split using the "createdatapartition" function from the Caret package in R, such that both CES-D-10 classes were represented in each set [28]. The algorithm was then developed by 10-fold cross-validation using only the training data and RF hyperparameters (i.e., various RF settings like number of decision trees) were tuned to maximize accuracy [27]; see Appendix C for details.

Additional Analyses

We conducted several analyses to provide additional details about the classification characteristics from the main analysis and assess robustness of the results: A) Using one visit per individual (total 717 visits), to assess difference in performance compared to the use of multiple visits per individual; B) Algorithms using three different CES-D-10 thresholds - 8, 13, and 15 - based on suggested cut-offs in the literature [29, 30]; C) An algorithm that included food insecurity as a predictor, which was collected only in the FS sub-study. Food insecurity in the past 6 months was measured using a 10-item adult scale

of the Household Food Security Survey Module (HFSSM) [31]. A categorical variable was used, with participants with 0-1, 2-5, or \geq 6 affirmative responses classified respectively as being food secure, moderately food insecure, or severely food insecure, as per the Health Canada criteria ; and D. RF regression algorithms to predict continuous CES-D-10 score, evaluated using R-squared and root mean squared error (RMSE), which provides information regarding differences between the predicted scores and the actual scores [32].

Performance evaluation

The final tuned algorithms were implemented in the testing data, which was not used in the development stage. The tuning parameters are shown in Table 4.6 in Appendix C. The overall performance and calibration measures are described in detail in Appendix C. To assess the ability to distinguish between classes (discrimination), we plotted receiver operating characteristic (ROC) and estimated the area under the ROC curve (AUC) [32, 33]. We used the default probability threshold of 0.50 for classification and at this threshold, sensitivity, specificity, positive predictive value (PPV), negative predictive value (NPV), positive likelihood ratio (LR+) and negative likelihood ratio (LR-) measures were then estimated with the 95% confidence intervals (CI) [34]. Finally, the RF importance metrics were generated, and importance plots were generated to present the 25 most important predictors in classifying participants with depressive symptoms by the two algorithms. We used RStudio v.1.2 and Stata v.16.0 to develop and evaluate these algorithms [35, 36]. The development of the RF algorithms was done using R package

ranger and caret; for performance evaluation, we used the performance assessment function in R by Wong et. al. (2019) [28, 37-39].

Results

Study population

Of the 1973 FS sub-study visits in a total of 725 participants, 39 study visits were excluded based on the exclusion criteria described in the methods - 16 visits (2 participants) with no CES-D-10 score, 18 visits (5 participants) with no con-current CCC visit and all 5 visits from 1 participant with no predictor data (EQ-5D health state) in all visits. Thus, 717 participants with a total of 1934 visits contributed to the final study sample. The participant characteristics at the first visit included in the sample are described in Table 4.2. The median CES-D-10 score was 10 (IQR, 5, 15), with 53% of the participants reporting the presence of depressive symptoms with CES-D-10 scores ≥10; 45% were prescribed one or more psychotropic medications such as bupropion and citalopram at baseline, but only 10% had a diagnosis of depression documented in their medical chart. Participants were predominantly male (73%) and white (76%). The population was vulnerable in terms of socioeconomic status (SES) characteristics with 73% unemployed, 76% with monthly income <\$1500, 52% with high school being the highest level of education and 46% receiving welfare at baseline. Approximately 34% were current injection drug users, 62% current alcohol drinkers and 75% were current tobacco smokers. Only a small proportion had advanced liver disease (4%) or a current AIDS related illness (4%) and 35% were asymptomatic with a current CD4 cell count >500 cells/ μ l (CDC clinical staging - A1).

Performance evaluation

The training data consisted of 1548 visits and testing data consisted of 386 visits. The algorithms in the primary analysis showed acceptable calibration, as seen in Figure 4.3 in appendix C. With regard to discrimination, the ROC curve for the primary analyses is shown in Figure 4.1, with the curve close to the upper left-hand corner. The estimated AUCs wert 0.82 (95% CI: 0.78-0.86) and 0.76 (95% CI: 0.71-0.81) for the full and reduced algorithms respectively. The estimated sensitivity, specificity, PPV, NPV, LR+ and LR-are presented in Table 4.3. The importance plots with 25 most important predictors for both algorithms are shown in Figure 4.2. Employment, HIV clinical stage, revenue source, body mass index (BMI), and education were the 5 most important predictors.

Additional analyses

The results for the additional analyses A-D are shown in Table 4.4 and summarized here. A) When using information from a single visit per individual, the overall performance was much lower, with AUC of 0.74 vs 0.82 for the full algorithm and 0.60 vs 0.76 for the reduced algorithm. B) For algorithms with the additional CES-D-10 thresholds, the AUC point estimate was higher for the full algorithm for cut-off 15 compared to 10 (0.87 vs 0.82). The other cut-off estimates were similar for to the corresponding algorithms for cutoff 10, with overlapping confidence intervals. C.) The algorithm including the additional predictor of food insecurity had a similar AUC estimate to the full algorithm, with overlapping confidence intervals. D.) The full algorithm predicting continuous CES-D-10 scores had a R-squared of 0.5 indicating that the algorithm explained only 50% of the variability in the CES-D-10 scores and had a high RMSE of 4.8, while for the reduced

algorithm with a r-squared of 0.3, the algorithm explained only 30% of the variability in the scores and also had a high RMSE of 5.5.

Discussion

We developed a random forest algorithms using patient data from a cohort study that reliably predicted the presence of depressive symptoms indicative of a risk of major depression in a vulnerable HIV-HCV co-infected population. The algorithms used a set of selected candidate predictors (x = 137) from the cohort. The full algorithm using all candidate predictors performed better, with an AUC of 0.82, which indicates a 82% chance of distinguishing between CES-D-10 classes compared 0.76 for the reduced algorithm which used a smaller subset (x = 46) [33]. The prevalence of depressive symptoms was very high in our study, with more than 50% individuals found to be at risk for depression by CES-D-10 at their first visit. Despite this, only 10% had a documented depression diagnosis in their medical record suggesting there could be a substantial underestimation of the burden of depressive illness in this population without screening. We developed this tool to assess if patient data that is commonly collected in clinical charts and research studies could be useful in predicting presence of depressive symptoms, which is seldom directly measured routinely for all patients, nor measured repeatedly over time. This tool will be most useful for conducting longitudinal clinical and epidemiologic research rather than for clinical care. It may prove useful to help identify people at risk for depression, study how this risk changes over time and with various interventions.

Most studies using machine learning have predicted the future onset of depressive symptoms [19, 40, 41] while a few, like ours, have focused on current depression prediction [42, 43]. A variety of predictors including demographic and clinical data, past medical history and life events have been studied. A range of machine learning algorithms like artificial neural networks, support vector machines, naïve Bayes classifier, and random forest were used in the general population and for specific groups like geriatric population and people with diabetes. These algorithms yielded AUC measures similar to ours, ranging between 0.70-0.95.

CCC collects extensive demographic, behavioral and clinical data. Using the full range and diversity of available predictors did show excellent discrimination in the full algorithm. However, for greater applicability, we chose to use a subset of 46 predictors that may be more readily available in other research settings, and despite using one third the number of predictors and less granular data, the overall discrimination was still acceptable. The additional analysis using only one visit per participant had a comparatively lower AUC, which may have been due to the smaller sample size and thus lower variability in the available data.

The algorithm we developed was a purely prediction algorithm and hence estimation of the strength of the effect of individual predictors is not possible. Further analysis with different modeling strategies would be needed for this assessment. However, the algorithm does provide some insight into factors that may be important for classification. The five most important predictors are related two main themes - i. SES (education, revenue source and employment) and ii. overall health status (HIV clinical stage and BMI). SES is a known strong determinant of depression. Receiving welfare and being
from a low-income household has been associated with an elevated risk of food insecurity, and mental health issues [44, 45]. In Canada, almost 20% of people with major depression have been reported to be unemployed [46]. Another important health status related predictor was BMI. There have been studies with conflicting results regarding association between BMI and depression, and possible difference across race and gender [47, 48] and that BMI categories may not adequately capture people's health status and thus this predictor needs to be considered with caution [49]. Finally, in the full algorithm with all 137 predictors, the EQ-5D-3L anxiety/depression dimension was the most important predictor and all EQ-5D dimensions (mobility, self-care, usual activities, pain/discomfort, and health state) were among the 25 most important. This provides further evidence that participant's health status, and in the case of EQ-5D-3L, their perceived health status, are important in predicting depressive symptoms.

This study thus has many strengths. The CCC is generalizable to the HIV-HCV coinfected patients engaged in care in Canada, due to the recruitment from a variety of clinical settings (outreach, primary and tertiary care clinics in urban and semi-urban areas across the country). The sample used to develop these algorithms was generalizable to the parent CCC (see Appendix D; Table 4.7). The methodology used, RF, is nonparametric, highly accurate, and relatively robust to outliers, noise and does have safeguards from overfitting and thus improves chances of applicability beyond the data. Nevertheless, external validation is needed before application in other cohorts and research, to mitigate the risk of overfitting. In addition, the predictor importance plots provided some insight regarding predictors that play a major role in the accurate prediction of depressive symptoms.

The study however has limitations. The sample size is small as compared to big data applications of RF using electronic health records. Some predictors described in other studies such as childhood trauma, food insecurity among others, were not available for the full CCC. For example, in the additional analysis where we added the food security variable that was collected only in the food security sub-study was included, the AUC was slightly higher. Additionally, we categorized the CES-D-10 to create the binary classes, and thus may have lost some data by not predicting the individual CES-D-10 scores. We did develop a regression algorithm to predict the continuous CES-D-10 scores in additional analysis E, but it could only explain a small portion of the variability in the outcome. The gold standard depression diagnosis was not available in this study and thus the validity of the cut-off of 10 could not be assessed directly in this sample. In general, the overall AUCs were similar when using three other suggested CES-D- 10 cut-offs (8, 13, and 15) compared to a cut-off of 10. However, the full algorithm using a cut-off of 15 appears to have a higher AUC (0.87) than that of using a cut-off 10 (0.82). It will be important to assess in future studies whether this higher threshold may be more applicable to the co-infected population. However, since the CES-D-10 cut-off of 10 has been validated in HIV populations in Canada [24], we decided to use this threshold for comparability with available literature and future studies which may use this common threshold.

With a high proportion of participants with depressive symptoms in this population, it is important not to miss possible cases. Even if the algorithms we developed are considered to have acceptable discrimination (\geq 0.7) based on arbitrary thresholds, we would still

misclassify a fair proportion of cases and thus this possible misclassification needs to be considered. Finally, this algorithm is applicable when the majority of the predictors are collected. However, in settings where such data is not available, especially completely clinical non-research setting implementing routine screening tools like the CES-D-10 should be considered, especially given the high prevalence of depressive symptoms we observed in this co-infected population.

Conclusions

Depressive symptoms indicative of a risk for depression were common in our population of people living with HIV-HCV co-infection. The random forest algorithms we developed shows promise in accurately predicting an elevated risk of major depression using data on patient characteristics collected in research settings. The algorithms identified important characteristics for depressive symptoms classification including employment, HIV clinical stage, revenue source, BMI, and education. Such machine learning algorithms can be used in research settings especially cohort studies where such data may be available to predict presence of depressive symptoms and use this information to understand the impact of depressive symptoms on clinical, health service and patientreported outcomes in vulnerable populations.

Table 4.1: Can	didate predictors	s used in the	random fores	st algorithms

Category	All Candidate Predictors
Questions related to mental health	Psychiatric institution or psychiatric hospital stay; Psychiatric diagnoses in chart reviews; Use of psychotropic medications (e.g., antidepressants, sedative hypnotics, atypical and typical anti-psychotics)
HR-QoL	EQ-5D-3L - standardized instrument - Descriptive system (mobility, self-care, usual activities, pain/discomfort, and anxiety/depression) - Current health state with a visual analog scale – scores range between 0-100
Sociodemographic characteristics	Age; Gender; Race/ethnicity; Immigration status; Living situation; Shared accommodation details; Education; Employment; Monthly income; Source of income
Behavioral characteristics	Injection drug use (ever/P6M); Non-injection drug use (ever/P6M); Needle/equipment sharing behaviors (ever/P6M); Snort (ever/P6M); Sharing behaviors for snorting apparatus (ever/P6M); Marijuana use; Therapy for drug addiction; Alcohol use (ever/P6M); Alcohol abuse; Smoking (ever/P6M); Sexual orientation; number of sexual partners; Sex work; Incarceration (ever/P6M); Tattoos; Body piercing
Clinical characteristics	BMI category; HIV viral load; CD4 count; HCV RNA; HIV disease stage; AIDS defining illness; health services used in the past 6 months (walk-in clinic, emergency room, inpatient, general practitioner, HIV clinic and specialist); Previous interferon-based HCV treatment; current antiretroviral therapy; Hepatitis B diagnosis; sexually transmitted disease diagnosis; end-stage liver disease (cirrhosis, ascites, varices, portal hypertension, encephalopathy, hepatocellular carcinoma); APRI, a measure of liver fibrosis; cardiovascular disease; autoimmune disease; hypertension; thyroid disease; psoriasis; lipodystrophy; hypercholesterolemia

Abbreviations: HR-QoL: Health related quality of life; EQ-5D-3L: EuroQoL-5Dimension-3Level; P6M: in the past 6 months; BMI: Body Mass Index; HIV: Human Immunodeficiency Virus; CD4: Cluster of differentiation 4 receptor; HCV: Hepatitis C virus; RNA: Ribonucleic acid; AIDS: Acquired Immunodeficiency Syndrome; APRI: Aspartate Aminotransferase (AST) to platelet ratio

Characteristics	Participants (n = 717)
	n (%) or median (IQR)
Age	49 (43, 54)
Gender – Male	522 (73)
Race/Ethnicity	
Asian	11 (2)
Black	28 (4)
White	541 (76)
Metis	32 (5)
First nation	102 (14)
Hispanic/Latino	7 (1)
Born outside Canada	64 (9)
Education - High school educated	376 (52)
Employment - Unemployed	525 (73)
Monthly income - < \$1500	543 (76)
Revenue Source - Welfare	332 (46)
Current injection drug use	244 (34)
Current alcohol use	444 (62)
Current smoking	534 (75)
BMI category	
Underweight (< 18.5 kg/m ²)	40 (6)
Normal weight (18.5-25 kg/m ²)	312 (44)
Overweight (25-29.9 kg/m ²)	189 (26)
Obese (30.0 kg/m ²)	87 (12)
End-stage Liver disease	27 (4)
HIV clinical stage - A1 (Asymptomatic and CD4 >500 cells/µl)	248 (35)
Past AIDS related illness	28 (4)
CES-D-10 score	10 (5, 15)
CES-D-10 category - ≥10	382 (53)
Depression diagnosis	68 (10)
Prescribed antidepressant medications	320 (45)
HR-QoL using EQ-5D-3L instrument	
Anxiety/depression	
Not anxious or depressed	352 (49)
Moderately anxious or depressed	302 (42)
Extremely anxious or depressed	60 (8)
Current health state (visual analog scale)	70 (56, 80)

Table 4.2: Baseline characteristics of participants in the study sample (n=717)

Abbreviations: IQR: Interquartile range; CES-D-10: Center for Epidemiologic Studies Depression Scale-10; AIDS: Acquired Immunodeficiency Syndrome; HIV: Human Immunodeficiency Virus; BMI: Body Mass Index; HR-QoL: Health related quality of life; EQ-5D-3L: EuroQoL-5Dimension-3Level

Table 4.3: Performance evaluation in the primary analysis

Evaluation measure (95% CI)	Full algorithm (x = 137)	Reduced algorithm (x=46)
AUC	0.82 (0.78-0.86)	0.76 (0.71-0.81)
Sensitivity	0.77 (0.70-83)	0.70 (0.63-0.76)
Specificity	0.73 (0.66-0.79)	0.70 (0.63-0.76)
PPV	0.74 (0.68-0.80)	0.70 (0.63-0.76)
NPV	0.76 (0.69-0.82)	0.69 (0.62-0.76)
LR +	2.8 (2.2-3.6)	2.3 (1.8-2.9)
LR -	0.3 (0.2 – 0.4)	0.4 (0.3-0.6)

Abbreviations: CI: Confidence interval; AUC: Area under the Receiver Operating Characteristic curve; PPV: Positive

Predictive Value; NPV: Negative Predictive Value; LR +: Positive likelihood ratio; LR -: Negative likelihood ratio

Table 4.4: Comparison of performance evaluation measures for primary and additional

analyses

Sr.	Analysis	OOB	AUC	Sensitivity	Specificity		
No.	Analysis	error	(95% CI)	(95% CI)	(95% CI)		
Prim	Primary Analyses						
1	Full algorithm (x = 137)	0.16	0.82 (0.78-0.86)	0.77 (0.70-0.83)	0.73 (0.66-0.79)		
2	Reduced algorithm ($x = 46$)	0.20	0.76 (0.71-0.81)	0.70 (0.63-0.76)	0.70 (0.63-0.76)		
Addi	tional Analyses						
Α	One visit per individual						
1	Full algorithm (x = 137)	0.17	0.74 (0.66-0.82)	0.70 (0.58-0.80)	0.64 (0.52-0.76)		
2	Reduced algorithm ($x = 46$)	0.23	0.60 (0.50-0.69)	0.73 (0.61-0.82)	0.37 (0.26-0.50)		
В	Different CES-D-10 cut-offs						
I	Cut-off – 8						
1	Full algorithm (x = 137)	0.17	0.85 (0.81-0.89)	0.91 (0.87-0.95)	0.67 (0.59-0.74)		
2	Reduced algorithm ($x = 46$)	0.19	0.79 (0.74-0.83)	0.83 (0.77-0.87)	0.60 (0.52-0.67)		
II	Cut-off – 13						
1	Full algorithm (x = 137)	0.15	0.80 (0.76-0.85)	0.62 (0.54-0.70)	0.81 (0.76-0.86)		
2	Reduced algorithm ($x = 46$)	0.20	0.79 (0.74-0.83)	0.45 (0.37-054)	0.87 (0.82-0.91)		
III	Cut-off - 15						
1	Full algorithm (x = 137)	0.14	0.87 (0.83-0.91)	0.55 (0.45-0.65)	0.95 (0.91-0.97)		
2	Reduced algorithm ($x = 46$)	0.17	0.73 (0.67-0.79)	0.27 (0.19-0.36)	0.94 (0.91-0.97)		
С	With food insecurity variable (x = 138)	0.16	0.84 (0.80-0.87)	0.77 (0.70-0.83)	0.73 (0.67-80)		

Abbreviations: OOB: Out-of-Bag Samples; AUC: Area under the Receiver Operating Characteristic curve; CI:

Confidence interval; CES-D-10: Center for Epidemiologic Studies Depression Scale-10

Figure 4.1: Receiver Operating Characteristic (ROC) curve for the A. Full algorithm (x = 137) and B. Reduced algorithm (x=46)



54









Abbreviations: BMI: Body Mass Index; P6M: In the past 6 months; CD4: Cluster of differentiation 4 receptor; EQ-5D-3L: EuroQoL-5Dimension-3Level; RNA: Ribonucleic acid; Hep B: Hepatitis B virus

Declarations

Ethics approval and consent to participate

This study was approved by the Research Ethics Board of the McGill University Health Centre (2021-6985). The CCC and the FS Sub-Study were approved by the Research Ethics Board of the McGill University Health Centre (2006-1875, BMB-06-006t, 2013-994) and the research ethics boards of participating institutions. The study was conducted according to the Declaration of Helsinki. Informed consent was obtained from all individual participants included in the study.

Consent for publication

Not applicable

Availability of data and materials

The datasets generated and/or analysed during the current study are not publicly available. According to the stipulations of patient consent provided and our Institutional Ethics review boards, study records including confidential information collected during the study must be stored securely for 25 years after study completion, as required by Canadian clinical trial regulations. However, data stripped of personal identifiers, may be shared upon request to the corresponding author, or to: Mr. Sheldon Levy, Clinical Trials 2 Research Ethics Board (REB) Coordinator, MUHC Centre for Applied Ethics (sheldon.levy@muhc.mcgill.ca).

Competing interests

JC received grants and consulting fees from ViiV Healthcare, Merck, and Gilead and personal fees from Bristol-Myers Squibb. CC has received personal fees for being a member of the national advisory boards of Gilead, Merck, Janssen, and Bristol-Myers Squibb. BC is a board member, consultant, and has received grants and payment for lectures from AbbVie, Gilead, and Merck, and payment for educational presentations from AbbVie. MH has served as a consultant for Merck, Vertex Pharmaceuticals, Pfizer, Viiv Healthcare, and Ortho-Jansen. MH has also received grants from the National Institute on Drug Abuse, as well as payment for lectures from Merck and Ortho-Janssen. MLV reports personal fees from Abbvie, personal fees from Merck, personal fees from Gilead, outside the submitted work. SW received grants, consulting fees, lecture fees, nonfinancial support, and fees for the development of educational presentations from Merck, ViiV Healthcare, GlaxoSmithKline, Pfizer, Gilead, AbbVie, Bristol-Myers Squibb, and Janssen. MBK reports grants for investigator-initiated studies from ViiV Healthcare, AbbVie, Merck, and Gilead; and consulting fees from ViiV Healthcare, Merck, AbbVie, and Gilead. GM, EEMM, MJB, CLD, VML, and AW have no conflicts of interest to disclose.

Funding

This work was supported by Fonds de recherche du Québec-Santé; Réseau sida/maladies infectieuses, the Canadian Institute for Health Research (CIHR; FDN-143270); and the CIHR Canadian HIV Trials Network (CTN222 & CTN264). GM is supported by the PhD trainee fellowship from the Canadian Network on Hepatitis C. MBK

is supported by a Tier I Canada Research Chair. The funders had no role in the production of this manuscript. EEMM is supported by a chercheur de mérite award from the Fonds de recherche du Québec-Santé and a Canada Research Chair (Tier 1). VML is supported by Clinical Research Scholars–Junior 1 from the Fonds de recherche du Québec-Santé.

Author contributions

All authors contributed to this study, as required by the International Committee of Medical Journal Editors. MBK, EEMM and GM conceived of and designed the study. GM and CLD prepared the analytical dataset. GM performed all statistical analyses. GM, EEMM and MBK drafted the initial manuscript. All co-authors MJB, JC, CC, BC, MH, VML, MLV, SW, and AW revised the document critically and gave final approval prior to completion. All authors take responsibility for the accuracy and integrity of this work.

Acknowledgements

We would like to acknowledge the participants of the Canadian Co-Infection Cohort (CTN222), the study coordinators and nurses for their assistance with study coordination, participant recruitment, and care, and the Canadian Co-Infection Cohort (CTN222) co-investigators - Drs. Lisa Barrett, Jeff Cohen, Brian Conway, Curtis Cooper, Pierre Côté, Joseph Cox, M. John Gill, Shariq Haider, David Haase, Mark Hull, Valérie Martel-Laferrière, Julio Montaner, Erica E. M. Moodie, Neora Pick, Danielle Rouleau, Aida Sadr, Steve Sanche, Roger Sandre, Mark Tyndall, Marie-Louise Vachon, Sharon Walmsley and Alexander Wong.

Authors' information

Not applicable.

List of Abbreviations

- AIDS: Acquired Immunodeficiency Syndrome
- APRI: Aspartate Aminotransferase (AST) to platelet ratio
- AUC: Area under the Receiver Operating Characteristic Curve
- BMI: Body Mass Index
- CCC: Canadian Co-infection Cohort
- CD4: Cluster of differentiation 4 receptor
- CDC: Centre for Disease Control
- CES-D-10: Center for Epidemiologic Studies Depression Scale-10
- CI: Confidence Interval
- EQ-5D-3L: EuroQol-5 Dimension-3 Level
- FS: Food Security
- HCV: Hepatitis C virus
- Hep B: Hepatitis B virus
- HFSSM: Household Food Security Survey Module
- HIV: Human Immunodeficiency Virus
- HR-QoL: Health-related Quality of Life
- IQR: Interquartile Range
- LR-: Negative likelihood ratio
- LR+: Positive likelihood ratio

NPV: Negative Predictive Value (NPV),

OOB: Out-of-Bag Samples

P6M: in the past 6 months

PPV: Positive Predictive Value

RF: Random Forests

- RMSE: Root Mean Squared Error
- RNA: Ribonucleic acid
- **ROC: Receiver Operating Characteristic**
- SES: Socioeconomic Status

References

 Shatte ABR, Hutchinson DM, Teague SJ. Machine learning in mental health: a scoping review of methods and applications. Psychological Medicine. 2019;49(9):1426-48.

2. Guidelines for the care and treatment of persons diagnosed with chronic hepatitis C virus infection. Geneva: World Health Organization; 2018. Report No.: License: CC BY-NC-SA 3.0 IGO.

3. Nanni MG, Caruso R, Mitchell AJ, Meggiolaro E, Grassi L. Depression in HIV infected patients: a review. Current psychiatry reports. 2015;17(1):530.

4. Younossi Z, Park H, Henry L, Adeyemi A, Stepanova M. Extrahepatic Manifestations of Hepatitis C: A Meta-analysis of Prevalence, Quality of Life, and Economic Burden. Gastroenterology. 2016;150(7):1599-608.

5. Yeoh SW, Holmes ACN, Saling MM, Everall IP, Nicoll AJ. Depression, fatigue and neurocognitive deficits in chronic hepatitis C. Hepatology international. 2018.

6. Fialho R, Pereira M, Rusted J, Whale R. Depression in HIV and HCV co-infected patients: a systematic review and meta-analysis. Psychology, health & medicine. 2017;22(9):1089-104.

7. Belenky NM, Cole SR, Pence BW, Itemba D, Maro V, Whetten K. Depressive Symptoms, HIV Medication Adherence, and HIV Clinical Outcomes in Tanzania: A Prospective, Observational Study. PLOS ONE. 2014;9(5):e95469.

8. Aibibula W, Cox J, Hamelin A-M, Moodie EEM, Anema A, Klein Marina B, et al. Association between depressive symptoms, CD4 count and HIV viral suppression among HIV-HCV co-infected people. AIDS Care. 2018;30(5):643-9.

9. Malhi GS, Mann JJ. Depression. Lancet (London, England). 2018;392(10161):2299-312.

10. Eaton WW, Johns Hopkins Bloomberg School of Public Health. Department of Mental H. Public mental health. New York, NY: Oxford University Press; 2012.

11. Ma H, Villalobos CF, St-Jean M, Eyawo O, Lavergne MR, Ti L, et al. The impact of HCV co-infection status on healthcare-related utilization among people living with HIV in British Columbia, Canada: a retrospective cohort study. BMC Health Services Research. 2018;18(1):319.

12. Grebely J, Oser M, Taylor LE, Dore GJ. Breaking Down the Barriers to Hepatitis C Virus (HCV) Treatment Among Individuals With HCV/HIV Coinfection: Action Required at the System, Provider, and Patient Levels. The Journal of Infectious Diseases. 2013;207(suppl_1):S19-S25.

13. Bonner JE, Barritt ASt, Fried MW, Evon DM. Time to rethink antiviral treatment for hepatitis C in patients with coexisting mental health/substance abuse issues. Dig Dis Sci. 2012;57(6):1469-74.

14. Knott A, Dieperink E, Willenbring ML, Heit S, Durfee JM, Wingert M, et al. Integrated psychiatric/medical care in a chronic hepatitis C clinic: effect on antiviral treatment evaluation and outcomes. Am J Gastroenterol. 2006;101(10):2254-62.

15. Anagnostopoulos A, Ledergerber B, Jaccard R, Shaw SA, Stoeckle M, Bernasconi E, et al. Frequency of and Risk Factors for Depression among Participants in the Swiss HIV Cohort Study (SHCS). PLOS ONE. 2015;10(10):e0140943.

16. Dwyer DB, Falkai P, Koutsouleris N. Machine Learning Approaches for Clinical Psychology and Psychiatry. Annual Review of Clinical Psychology. 2018;14(1):91-118.

17. Graham S, Depp C, Lee EE, Nebeker C, Tu X, Kim HC, et al. Artificial Intelligence for Mental Health and Mental Illnesses: an Overview. Current psychiatry reports. 2019;21(11):116.

18. Sau A, Bhakta I. Artificial Neural Network (ANN) Model to Predict Depression among Geriatric Population at a Slum in Kolkata, India. J Clin Diagn Res. 2017;11(5):Vc01-vc4.

19. Jin H, Wu S, Di Capua P. Development of a Clinical Forecasting Model to Predict Comorbid Depression Among Diabetes Patients and an Application in Depression Screening Policy Making. Prev Chronic Dis. 2015;12:E142.

20. Klein MB, Saeed S, Yang H, Cohen J, Conway B, Cooper C, et al. Cohort profile: the Canadian HIV-hepatitis C co-infection cohort study. International journal of epidemiology. 2010;39(5):1162-9.

21. Cox J, Hamelin AM, McLinden T, Moodie EE, Anema A, Rollet-Kurhajec KC, et al. Food Insecurity in HIV-Hepatitis C Virus Co-infected Individuals in Canada: The Importance of Co-morbidities. AIDS and behavior. 2017;21(3):792-802.

22. EuroQol--a new facility for the measurement of health-related quality of life. Health policy (Amsterdam, Netherlands). 1990;16(3):199-208.

23. Andresen EM, Malmgren JA, Carter WB, Patrick DL. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). American journal of preventive medicine. 1994;10(2):77-84.

24. Zhang W, O'Brien N, Forrest JI, Salters KA, Patterson TL, Montaner JS, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. PLOS ONE. 2012;7(7):e40793.

25. Breiman L. Random Forests. Machine Learning. 2001;45(1):5-32.

26. Malley JD, Kruppa J, Dasgupta A, Malley KG, Ziegler A. Probability machines: consistent probability estimation using nonparametric learning machines. Methods Inf Med. 2012;51(1):74-81.

27. Hastie T, Tibshirani R, Friedman JH. The elements of statistical learning : data mining, inference, and prediction. Second edition. ed. New York: Springer; 2009.

Max Kuhn Contributions from Jed Wing SW, Andre Williams, Chris Keefer, Allan Engelhardt, Tony Cooper, Zachary Mayer and Brenton Kenkel, R Core Team, Michael Benesty, Reynald Lescarbeau, Andrew Ziem, Luca Scrucca, Yuan Tang, Can Candan, Tyler Hunt. caret: Classification and Regression Training. R package version 6.0-842019.
Baron EC, Davies T, Lund C. Validation of the 10-item Centre for Epidemiological Studies Depression Scale (CES-D-10) in Zulu, Xhosa and Afrikaans populations in South Africa. BMC Psychiatry. 2017;17(1):6.

30. Bjorgvinsson T, Kertz SJ, Bigda-Peyton JS, McCoy KL, Aderka IM. Psychometric properties of the CES-D-10 in a psychiatric sample. Assessment. 2013;20(4):429-36.

31. Canadian Community Health Survey, Cycle 2.2, Nutrition (2004): Income-Related Household Food Security in Canada. Health Canada, Ottawa, Canada; 2007.

32. Steyerberg EW. Clinical prediction models : a practical approach to development, validation, and updating. Cham, Switzerland: Springer; 2019.

Hosmer DW, Lemeshow S, Sturdivant RX. Applied Logistic Regression. Chicester:
Wiley; 2013.

34. Youden WJ. Index for rating diagnostic tests. Cancer. 1950;3(1):32-5.

35. StataCorp. Stata Statistical Software: Release 16. College Station, TX: StataCorp LLC; 2019.

36. R Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria2013.

37. Wright MN, Ziegler A. ranger: A Fast Implementation of Random Forests for High Dimensional Data in C++ and R. 2017. 2017;77(1):17.

38. Wong J, Manderson T, Abrahamowicz M, Buckeridge DL, Tamblyn R. Can Hyperparameter Tuning Improve the Performance of a Super Learner?: A Case Study. Epidemiology. 2019;30(4):521-31.

39. Joie E, Kym IES, Emma CM. PMCALPLOT: Stata module to produce calibration plot of prediction model performance. S458486 ed: Boston College Department of Economics; 2018.

40. Sau A, Bhakta I. Screening of anxiety and depression among the seafarers using machine learning technology. Informatics in Medicine Unlocked. 2018:100149.

41. Rosellini AJ, Liu S, Anderson GN, Sbi S, Tung ES, Knyazhanskaya E. Developing algorithms to predict adult onset internalizing disorders: An ensemble learning approach. J Psychiatr Res. 2019;121:189-96.

42. Wang J, Sareen J, Patten S, Bolton J, Schmitz N, Birney A. A prediction algorithm for first onset of major depression in the general population: development and validation. J Epidemiol Community Health. 2014;68(5):418-24.

43. King M, Bottomley C, Bellon-Saameno J, Torres-Gonzalez F, Svab I, Rotar D, et al. Predicting onset of major depression in general practice attendees in Europe:

extending the application of the predictD risk algorithm from 12 to 24 months. Psychol Med. 2013;43(9):1929-39.

44. Coiro MJ. Depressive symptoms among women receiving welfare. Women Health. 2001;32(1-2):1-23.

45. Wu S, Fraser MW, Chapman MV, Gao Q, Huang J, Chowa GA. Exploring the relationship between welfare participation in childhood and depression in adulthood in the United States. Soc Sci Res. 2018;76:12-22.

46. Rizvi SJ, Cyriac A, Grima E, Tan M, Lin P, Gallaugher LA, et al. Depression and employment status in primary and tertiary care settings. Can J Psychiatry. 2015;60(1):14-22.

47. Monda V, La Marra M, Perrella R, Caviglia G, Iavarone A, Chieffi S, et al. Obesity and brain illness: from cognitive and psychological evidences to obesity paradox. Diabetes Metab Syndr Obes. 2017;10:473-9.

48. Banack HR, Kaufman JS. From bad to worse: collider stratification amplifies confounding bias in the "obesity paradox". Eur J Epidemiol. 2015;30(10):1111-4.

49. Nuttall FQ. Body Mass Index: Obesity, BMI, and Health: A Critical Review. Nutr Today. 2015;50(3):117-28.

Consortium name: Canadian Co-infection Cohort

Principal investigator: Marina B. Klein^{1, 3, 11}

Co-investigators: Lisa Barrett¹², Jeff Cohen¹³, Brian Conway⁵, Curtis Cooper⁴, Pierre Côté¹⁴, Joseph Cox^{1, 3}, John Gill¹⁵, Shariq Haider¹⁶, Mark Hull⁶, Valérie Martel-Laferrière⁷, Erica E. M. Moodie¹, Neora Pick¹⁷, Danielle Rouleau¹⁸, Steve Sanche¹⁹, Roger Sandre²⁰, Marie-Louise Vachon⁸, Sharon Walmsley⁹, Alexander Wong¹⁰

- 12. Dalhousie University, Halifax, Nova Scotia, Canada
- 13. Windsor Regional Hospital Metropolitan Campus, Windsor, Ontario, Canada
- 14. Clinique Médicale du Quartier Latin, Montreal, Quebec, Canada
- 15. Southern Alberta HIV Clinic, Calgary, Alberta, Canada
- 16. McMaster University, Hamilton, Ontario, Canada
- 17. Oak Tree Clinic, Vancouver, British Columbia, Canada
- 18. Université de Montréal, Montreal, Quebec, Canada
- 19. University of Saskatchewan, Saskatoon, Saskatchewan, Canada
- 20. Sudbury Regional Hospital, Sudbury, Ontario, Canada

4.3. Manuscript 1: Appendix

- Appendix A: Candidate predictors
- Appendix B: Introduction to Random Forests
- Appendix C: Supplementary methods and results
- **Appendix D:** Sample generalizability

Appendix A: Candidate Predictors

Table 4.5 below provides an exhaustive list of predictors and their corresponding categories used in this analysis.

Table 4.5: Supplementary table - List of candidate predictors

			Participant visits (n=1934)	Include Reduced	ded in
Sr. no.	Predictors	Categories	n (%) or Median (IQR)	Reduced Algorithm (x = 46)	Full Algorithm (x = 137)
1	Have you ever been in a psychiatric institution or psychiatric hospital? (Ever or since the last interview)	Yes No No response	106 (5.5) 1825 (94.4) 3 (0.2)	Yes	Yes
2	Use of psychotropic medications	Yes No	884 (45.7) 1050 (54.3)	Yes	Yes
	Has the patient ever had any of the following psychiatric diagnoses? (Ever at baseline)				
3	Depression	Yes No No response	102 (5.3) 1829 (94.6) 3 (0.2)	Yes	Yes
4	Bipolar disorder	Yes No No response	12 (0.6) 1918 (99.2) 4 (0.2)	No	Yes
5	Schizophrenia	Yes No No response	15 (0.8) 1915 (99.0) 4 (0.2)	No	Yes
6	Personality disorder	Yes No No response	18 (0.9) 1912 (98.9) 4 (0.2)	No	Yes
7	Other psychiatric disorder	Yes	40 (2.1) 1894 (97.9)	No	Yes
8	Any psychiatric diagnoses (other than depression)	Yes No No response	69 (3.6) 1862 (96.3) 3 (0.2)	Yes	No
	Use of specific substances				
9	Cocaine	Yes No	553 (28.6) 1381 (71.4)	No	Yes
10	Crack	Yes No	373 (19.3) 1561 (80.7)	No	Yes
11	Heroin	Yes No No response	216 (11.2) 1717 (88.8) 1 (0.1)	No	Yes

12	Speedball (Cocaine + Heroine)	Yes No No response	34 (1.2) 1899 (98.2) 1 (0.1)	No	Yes
13	PCP/Mescaline	Yes No No response	6 (0.3) 1927 (99.6) 1 (0.1)	No	Yes
14	Methadone	Yes No	125 (6.5) 1809 (93.5)	No	Yes
15	Morphine	Yes No No response	125 (6.5) 1808 (93.4) 1 (0.1)	No	Yes
16	LSD	Yes No No response	9 (0.5) 1924 (99.4) 1 (0.1)	No	Yes
17	Amphetamines	Yes No No response	91 (4.7) 1842 (95.2) 1 (0.1)	No	Yes
18	Methamphetamine	Yes No No response	152 (7.8) 1781 (92.1) 1 (0.1)	No	Yes
19	Talwin/Ritalin	Yes No No response	8 (0.4) 1925 (99.5) 1 (0.1)	No	Yes
20	Ritalin alone	Yes No No response	43 (2.2) 1890 (97.7) 1 (0.1)	No	Yes
21	Benzodiazepines	Yes No	86 (4.5) 1848 (95.5)	No	Yes
22	Barbiturates	Yes No No response	16 (0.8) 1917 (99.1) 1 (0.1)	No	Yes
23	Dilaudid	Yes No	130 (6.7) 1804 (92.3)	No	Yes
24	Percocet	Yes No No response	39 (2.0) 1894 (97.9) 1 (0.1)	No	Yes
25	Oxycodone	Yes No No response	61 (3.2) 1872 (96.7) 1 (0.1)	No	Yes
26	Alcohol	Yes No No response	13 (0.7) 1920 (99.2) 1 (0.1)	No	Yes

27	Freebase	Yes No	71 (3.7) 1863 (96.3)	No	Yes
28	Demerol	Yes No	3 (0.2) 1931 (99.8)	No	Yes
29	MDA	Yes No	17 (0.9) 1917 (99.1)	No	Yes
30	Mushrooms	Yes No	15 (0.8) 1919 (99.2)	No	Yes
31	Solvent - drink (Aqua Velva)	Yes No	4 (0.2) 1930 (99.8)	No	Yes
32	Solvent - sniff (gas, glue, Lysol, Pam)	Yes No	3 (0.2) 1931 (99.8)	No	Yes
33	Tylenol with codeine	Yes No	69 (3.6) 1865 (96.4)	No	Yes
	EQ-5D- 3L instrument				
34	Mobility	No problem in walking Some problem in walking Confined to bed No response	1319 (68.2) 605 (31.3) 8 (0.4) 2 (0.1)	No	Yes
35	Self-care	No problem with self-care Some problem with self-care Unable to wash or dress myself No response No problem	1760 (91.0) 158 (8.2) 15 (0.8) <u>1 (0.1)</u> 1328 (68.7)	No	Yes
36	Usual activities	in performing my usual activities	575 (29.8)	No	Yes

		Some problem in performing my usual activities			
		Unable to perform my usual activities	28 (1.5)		
		No response	3 (0.2)		
		No pain or discomfort	759 (39.3)		
37	Pain and discomfort	Moderate pain or discomfort	929 (48.0)	No	Yes
		Extreme pain or discomfort	245 (12.7)		
		No response	1 (0.1)		
		Not anxious or depressed	1002 (51.8)		
38	Anxiety and Depression	Moderately anxious or depressed	772 (39.9)	No	Yes
		Extremely anxious or depressed	157 (8.1)		
		No response	3 (0.2)		
39	Health State	Continuous; Range: 0 – 100; 0 = worst, 100 = best	70 (60, 80)	No	Yes
40	Age	15-25 26-35 36-45 46-55 56-65 66-75 >=75	3 (0.12) 147 (7.6) 387 (20.0) 944 (48.8) 414 (21.4) 36 (1.9) 3 (0.20	Yes	Yes
41	Gender	Male Female Transgender No response	1421 (73.5) 489 (25.3) 21 (1.1) 3 (0.2)	Yes	Yes
42	Marital status	Single Married or common-law Widow(er) Divorced	1359 (70.3) 324 (16.8) 50 (2.6) 163 (8.4)	Yes	Yes

		No response	38 (1.9)		
	Race/ethnicity				
	A	Yes	26 (1.3)		
43	Asian	No	1899 (98.2)	Yes	Yes
		No response	9 (0.5)		
	Diad	Yes	75 (3.9)		
44	Black	No	1843 (95.3)	Yes	Yes
		No response	16 (0.8)		
	NA/1 1/	Yes	1483 (76.7)		
45	vvnite	No	437 (22.6)	Yes	Yes
43		No response	14 (0.7)		
	Matia	Yes	80 (4.1)		
46	mens	No	1841 (95.2)	Yes	Yes
		No response	13 (0.7)		
	First Nation	Yes	260 (13.4)		
47	FIISUNATION	No	1663 (86.0)	Yes	Yes
		No response	11 (0.6)		
		Yes	18 (0.9)		
48	Hispanic/Latino	No	1907 (98.6)	Yes	Yes
		No response	9 (0.5)		
		Yes	1470 (76.0)		
49	Country of origin – Born in Canada	No	175 (9.1)	Yes	Yes
		No response	289 (14.9)		
		Fixed	971 (50.2)		
		address			
		Share			
		accommodati	786 (40.6)		
		ons			
50	Living situation			Yes	Yes
	5	Live in	407 (0.0)		
		sneiter or	127 (6.6)		
		residence			
		Homologo	10 (D E)		
		TIOMEIESS	40 (2.5)		
		No response	2 (0 1)		
		Less than	16 (0.8)		
		elementary	10 (0.0)		
		ciementary			
		Elementary	348 (18.0)		
		school	010(10.0)		
51	Education	High school	1046 (54.1)	Yes	Yes
		5			
		College	303 (15.7)		
		J J	· · · · ·		
		University	190 (9.8)		
			· · /		
		No response	31 (1.6)		
		Not working	1234 (63.8)		
		for bealth			
52	Employment	reasons		Yes	Yes
		10030113			
			95 (4.9)		

			[
		Not working			
		for lifestyle			
		reasons			
			84 (4.3)		
		Not working			
		but able to			
		work			
		WOIK	25 (1 2)		
		Otherskiller at	25 (1.5)		
		Studying			
			187 (9.7)		
		Part-time			
		work			
			181 (9.4)		
		Full-time			
		work			
		Work	10 (0 5)		
		Homoworkor	10 (0.5)		
		Homeworker			
			66 (3.4)		
		Retired			
			46 (2.4)		
		Other			
			6 (0.3)		
		No response			
		<= \$1500	1504 (77.8)		
53	Monthly income	<pre>\$1500</pre>	422 (21.9)	Ves	Ves
00			9(0.1)	103	103
		None	0 (0.4)		
		None	25 (1.3)		
		Welfare	915 (47.3)		
		Employment			
		insurance	22 (1.1)		
		Disability			
		insurance	574 (29 7)		
		incurance	011 (2011)		
54	Revenue source	Donaion		Yes	Yes
		FEIISION	02 (4.0)		
		0.11	92 (4.8)		
		Self-			
		employment	39 (2.0)		
		Employment	190 (9.8)		
		Other	73 (3.8)		
			- ()		
		No response	4 (0 2)		
		0	362 (19 7)		
		1 5	302(10.7)		
	Shared accommodation – number of	C-1		N.I	Ver
55	adults	0-10	29 (1.5)	NO	Yes
		>10	23 (1.2)		
		No response	769 (39.8)		
		0	865 (44.7)		
50	Shared accommodation – number of	1-5	174 (9.0)	N.	
56	children	6-10	1 (0.1)	NO	Yes
		No response	894 (46 2)		
57	Sovual orientation	Hotoropovus	1216 (60 1)	Vee	Voo
57	Sexual Ullerilation	THELETUSEXUAL	1310 (00.1)	i es	res

		Homosexual	396 (20.5) 198 (10.2)		
		No response	24 (1.2)		
58	Current injection drug use	Yes	627 (32.4) 1307 (67.6)	Yes	Yes
59	Current non-injection drug use	Yes	633 (32.7) 1301 (67.3)	Yes	Yes
60	Current alcohol use	Yes	1163 (60.1)	Yes	Yes
		NO Ves	313 (16 2)		
		No	874 (45.2)		
61	Alcohol abuse	Not	739 (38.2)	Yes	Yes
		applicable	· · · · ·		
		No response	8 (0.4)		
		Yes	1443 (74.6)		
62	Current smoking	No	488 (25.2)	Yes	Yes
		No response	3 (0.2)		
		Yes	1203 (62.2)		
63	Been in jail	No	726 (37.5)	Yes	Yes
		No response	5 (0.3)		
	Therapy or in a program for drug	Yes	1233 (63.8)		
64	addiction	No	649 (33.6)	No	Yes
		No response	52 (2.6)		
	Therapy or in a program for drug	Yes	385 (19.9)		
65	addiction (last 6 months)	No	1497 (77.4)	No	Yes
-		No response	52 (2.7)		
		Yes	544 (28.1)		
	Therapy or in a program for alcohol	No	1220 (63.1)		
66	addiction	Not	2 (0.1)	No	Yes
		applicable	400 (0 7)		
		No response	168 (8.7)		
07	Mariiyana	Yes	1039 (53.7)	Vaa	Vaa
67	manjuana	No rosponso	092 (40.1)	res	res
		No response	3 (0.2) 802 (46 1)		
		Not	092 (40.1)		
		applicable (if			
		don't use pot)			
			296 (15 3)		
		Occasionally	200 (10.0)		
		- not every			
		week			
			142 (7.3)		
		Regularly - 1-	()		
68	Frequency of Marijuana use	2 days per		No	Yes
		week			
		Demile	149 (7.7)		
		Regularly - 3-	~ /		
		6 days per			
		week			
		Evendov			
		Everyday	452 (23.4)		
		No response			
		1010300136	3 (0.2)		

		Net	002 (40.2)		
		INUT	893 (46.2)		
		applicable			
		Delieve	270 (11 0)		
	Why do you use Marijuana?	Relieve	270 (14.0)		
		symptoms	220 (11 0)		
		Increase	229 (11.0)		
		Increase			
		appente	222 (17 2)		
		Euro	333 (17.2)		
		Fun	00(47)		
60		Sumptomo	90 (4.7)	No	Vee
69		Symptoms		INO	162
		and appente	26 (1 2)		
		Symptome	20 (1.3)		
		and fun			
			17 (0 0)		
		Appetite and	17 (0.9)		
		fun			
		Turr			
		All three	37 (1 0)		
			57 (1.5)		
		No response	39 (2 0)		
		Ves	88 (4 6)		
70	Engaged in sex work in the past 6	No	1832 (94 7)	No	Yes
10	months	No response	14 (0 7)		100
		Yes	61 (3.2)		
71	Used services of sex workers in the	No	1860 (96.2)	No	Yes
<i>'</i> '	past 6 months	No response	13 (0.7)	110	100
		Yes	122 (6.3)		
72	Body piercing	No	1808 (93.5)	No	Yes
		No response	4 (0.2)		
		Yes	189 (9.8)		
73	Tattoo	No	1735 (89.7)	No	Yes
_		No response	10 (0.5)	-	
		Yes	1616 (83.6)		
74	IDU ever	No	308 (15.9)	Yes	Yes
		No response	10 (0.5)		
		Yes	1191 (61.6)		
	Shared needles ever if IDU=1	No	421 (21.8)		
75		Not		No	Yes
		applicable	280 (14.5)		
		No response	42 (2.2)		
		Yes	1151 (59.5)		
		No	470 (24.3)		
76	Equipment share ever if IDU=1	Not	282 (14.6)	No	Yes
		applicable	31 (1.6)		
		No response			
		Yes	269 (13.9)		
		No	1023 (52.9)		
		Never been	317 (16.4)		
77	Injected Drugs in jail ever if IDU=1	in jail		No	Yes
		Not	289 (14.9)		
		applicable			
		No response	36 (1.9)		

78	Sport ever	Yes No	1609 (83.2) 282 (14 6)	Yes	Yes
10		No response	43 (2.2)	100	100
		Yes	1177 (60.9)		
		No	552 (28.5)		
79	Snort share ever	Not	144 (7.5)	No	Yes
		applicable	61 (3.2)		
		No response			
		Yes	38 (2.0)		
	Shared needles in the past 6 months	No	602 (31.3)		
80	if IDLL 6m-1	Not		No	Yes
		applicable	1145 (59.2)		
		No response	149 (7.7)		
		Yes	62 (3.2)		
	Equipment share in the past 6 months	No	705 (36.5)		
81	if IDU 6m=1	Not		No	Yes
		applicable	1119 (57.9)		
		No response	48 (2.5)		
		Yes	9 (0.5)		
		No	225 (11.6)		
		Not been in	921 (47.6)		
00	Injected Drugs in jail in the past 6	jail in the past		Nie	Vaa
82	months if IDU_6m=1	6 months		INO	res
		Not	772 (20.0)		
		applicable	113 (39.9)		
		No response	6 (0 2)		
		Voc	425 (21 Q)		
83	Sport in the past 6 months	No	1502 (27.7)	No	Ves
00	Short in the past o months	No response	7 (0 4)	NO	105
		Yes	103 (5.3)		
		No	1224 (63.3)		
84	Snort share in the past 6 months	Not	1221 (00.0)	No	Yes
		applicable	560 (28,9)		
		No response	47 (2.4)		
		Yes	79 (4.1)		
	Did you get youd equipment in the	No	668 (34.5)		
85	Dia you get usea equipment in the	Not		No	Yes
	past 6 months?	applicable	1139 (58.9)		
		No response	48 (2.5)		
		Yes	55 (2.8)		
	Did you give used equipment in the	No	831 (42.9)		
86	past 6 months?	Not		No	Yes
		applicable	989 (51.1)		
		No response	59 (3.1)		
-		Yes	1738 (89.9)		
87	Ever smoked	No	175 (9.1)	Yes	Yes
		No response	21 (1.1)		
		Yes	767 (39.7)		
		NO	166 (8.6)	N N	
88	Previously consumed alcohol	NOT	000 (54.0)	Yes	Yes
		applicable	992 (51.3)		
		ino response	9 (0.5)		
89	in life	0	769 (39.8)	No	Yes
		1	41 (Z.1)		

$\begin{array}{c c c c c c c c c c c c c c c c c c c $			0 5			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	1		2-5	162 (8.4)		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			6-10	164 (8.5)		
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$			11-50	219 (11.3)		
90 No. of female sexual partners ever had in life >100 No response 401 (20.7) 38 (2.0) 90 0 507 (26.2) 1 1 115 (6.0) 2-5 359 (18.6) 6-10 255 (13.2) 11-50 No 91 11-50 409 (21.2) 51-100 No 92 No response 35 (1.8) 116 (6.0) No response 1132 (68.9)			51-100	140 (7.2)		
No No response 38 (2.0) 90 No. of female sexual partners ever had in life 0 507 (26.2) 1 115 (6.0) 2-5 359 (18.6) 6-10 255 (13.2) No Yes 11-50 409 (21.2) 51-100 138 (7.1) >100 116 (6.0) No response 35 (1.8) 0 1332 (68.9) 0 1332 (68.9)			>100	401 (20.7)		
90 No. of female sexual partners ever had in life 0 507 (26.2) 1 115 (6.0) 2-5 359 (18.6) 90 No. of female sexual partners ever had in life 6-10 255 (13.2) No Yes 90 No. of female sexual partners ever had in life 6-10 255 (13.2) No Yes 90 No response 31-100 138 (7.1) >100 116 (6.0) Yes 90 No response 35 (1.8) 0 1332 (68.9) 1332 (68.9) 1332 (68.9)			No response	38 (2.0)		
90 No. of female sexual partners ever had in life 1 115 (6.0) 2-5 359 (18.6) 6-10 255 (13.2) No Yes 90 No. of female sexual partners ever had in life 11-50 409 (21.2) No Yes 51-100 138 (7.1) >100 116 (6.0) No Yes 0 1332 (68.9) 0 1332 (68.9) 1332 (68.9) 1332 (68.9)			0	507 (26.2)		
90 No. of female sexual partners ever had in life 2-5 359 (18.6) No Yes 90 No. of female sexual partners ever had in life 6-10 255 (13.2) No Yes 51-100 138 (7.1) 51-100 138 (7.1) No Yes 0 132 (68.9) 0 1332 (68.9) 1332 (68.9) 1332 (68.9)			1	115 (6.0)		
90 No. of female sexual partners ever had in life 6-10 255 (13.2) No Yes 90 had in life 11-50 409 (21.2) No Yes 51-100 138 (7.1) >100 116 (6.0) Ves No No Yes No Yes			2-5	359 (18.6)		
90 had in life 11-50 409 (21.2) NO Yes 51-100 138 (7.1) >100 116 (6.0) 116 (6.0) 1132 (68.9) 11332 (68.9)	00	No. of female sexual partners ever	6-10	255 (13.2)	No	Vaa
51-100 138 (7.1) >100 116 (6.0) No response 35 (1.8) 0 1332 (68.9)	90	had in life	11-50	409 (21.2)	INO	res
>100 116 (6.0) No response 35 (1.8) 0 1332 (68.9)			51-100	138 (7.1)		
No response 35 (1.8) 0 1332 (68.9)			>100	116 (6.0)		
0 1332 (68.9)			No response	35 (1.8)		
			0	1332 (68.9)		
			1	332 (17 2)		
2-5 169 (8 7)			2-5	169 (8 7)		
No of male sexual partners in the past 6-10 47 (2.4)		No of male sexual partners in the past	6-10	47 (2 4)		
91 6 months No Yes	91	6 months	11-50	23 (1 2)	No	Yes
51-100 6 (0.3)			51-100	6 (0 3)		
			>100	12 (0.6)		
			No response	12 (0.0)		
			0	1457 (75.2)		
			1	229 (17 5)		
			2.5	02 (17.5)		
$2-3 \qquad \qquad$		No of formale actual northears in the	2-0	03 (4.3)		
92 not 6 menthe Sexual partners in the 6-10 15 (0.8) No Yes	92	No. of remaie sexual partners in the	0-10	15 (0.8)	No	Yes
past o montris 11-50 10 (0.0)		pasi o montris	TT-30 54 400			
			51-100	7 (0.4)		
			>100	5 (0.3)		
			No response	13 (0.7)		
Yes 113 (5.8)			Yes	113 (5.8)		
Shot up with steady sex partner in the No. 370 (19.1)	00	Shot up with steady sex partner in the	NO Not	370 (19.1)	NL	Maa
93 past 6 months Not Not No Yes	93	past 6 months	Not	400.4 (00.0)	NO	Yes
applicable 1294 (66.9)			applicable	1294 (66.9)		
No response 157 (8.1)			No response	157 (8.1)		
Yes 135 (6.9)			Yes	135 (6.9)		
Shot up with close friend or family No 351 (18.2)		Shot up with close friend or family	NO	351 (18.2)		
94 member in the past 6 months Not Not No Yes	94	member in the past 6 months	Not		No	Yes
applicable 1296 (67.0)		member in the past o months	applicable	1296 (67.0)		
No response 152 (7.9)			No response	152 (7.9)		
Yes 63 (3.3)			Yes	63 (3.3)		
Shot up with people the patient No 407 (21.0)		Shot up with people the patient	No	407 (21.0)		
95 doesn't know very well in the past 6 Not No Yes	95	doesn't know very well in the past 6	Not		No	Yes
months applicable 1296 (67.0)		months	applicable	1296 (67.0)		
No response 168 (8.7)	L		No response	168 (8.7)		
Yes 26 (1.3)			Yes	26 (1.3)		
Shot up with people the patient No 434 (22.4)		Shot up with people the patient	No	434 (22.4)		
96 doesn't know at all in the past 6 Not No Yes	96	doesn't know at all in the past 6	Not		No	Yes
months applicable 1296 (67.0)		months	applicable	1296 (67.0)		
No response 178 (9.2)			No response	178 (9.2)		
Yes 353 (18.3)			Yes	353 (18.3)		
Shot up with pohody (clope) in the No 262 (13.6)		Chat up with nahady (alama) in the	No	262 (13.6)		
97 Shot up with hobody (alone) in the Not No Yes	97	shot up with hobody (alone) in the	Not		No	Yes
applicable 1311 (67.8)		past o months	applicable	1311 (67.8)		
No response 8 (0.4)			No response	8 (0.4)		

98	Got needles or injecting equipment from steady sex partner in the past 6 months	Yes No Not applicable No response	24 (1.2) 9 (0.5) 1799 (93.0) 102 (5.3)	No	Yes
99	Got needles or injecting equipment from close friend or family member in the past 6 months	Yes No Not applicable No response	24 (1.2) 11 (0.6) 1799 (93.0) 100 (5.2)	No	Yes
100	Got needles or injecting equipment from people the patient doesn't know very well in the past 6 months	Yes No Not applicable No response	22 (1.1) 10 (0.5) 1800 (93.1) 102 (5.3)	No	Yes
101	Got needles or injecting equipment from people the patient doesn't know at all in the past 6 months	Yes No Not applicable No response	6 (0.3) 15 (0.80) 1800 (93.1) 113 (5.8)	No	Yes
102	Got needles or injecting equipment from nobody in the past 6 months	No Not applicable No response	15 (0.8) 1800 (93.1) 99 (6.2)	No	Yes
103	Give needles or injecting equipment to steady sex partner in the past 6 months	Yes No Not applicable No response	23 (1.2) 3 (0.2) 1833 (94.8) 75 (3.9)	No	Yes
104	Give needles or injecting equipment to close friend or family member in the past 6 months	Yes No Not applicable No response	16 (0.8) 14 (0.7) 1829 (94.6) 75 (3.9)	No	Yes
105	Give needles or injecting equipment to people the patient doesn't know very well in the past 6 months	Yes No Not applicable No response	9 (0.5) 7 (0.4) 1815 (93.9) 103 (5.3)	No	Yes
106	Give needles or injecting equipment to people the patient doesn't know at all in the past 6 months	Yes No Not applicable No response	8 (0.4) 8 (0.4) 1815 (93.9) 103 (5.3)	No	Yes
107	Give needles or injecting equipment to nobody (alone) in the past 6 months	No Not applicable No response	8 (0.4) 1816 (93.9) 110 (5.7)	No	Yes
108	BMI	Underweight Normal weight Overweight Obese No response	117 (6.1) 860 (44.5) 479 (24.8) 238 (12.3) 240 (12.4)	Yes	Yes

		<= 50	1591 (82.3)		
		(undetectabl	1001 (0210)		
109	HIV Viral load	(dildotootdol e)	316 (16 3)	Yes	Yes
105		50	010 (10.0)	103	105
		> JU No response	27 (1 4)		
-			27 (1.4)		
110		<= 250 (IOW)	323 (16.7)	Maria	Mara
110	CD4 count	> 250	1599 (82.7)	Yes	Yes
		No response	12 (0.6)		
		Detectable	815 (42.1)		
111	HCV RNA status	Not	383 (19.8)	Ves	Ves
		detectable		163	165
		Not done	736 (38.1)		
		A1	719 (37.2)		
		A2	566 (29.3)		
		A3	168 (8.7)		
		B1	63 (3 3)		
		B2	45 (2 3)		
112	HIV disease stage	B3	11 (0.6)	Yes	Yes
		C1	28 (1.5)		
			20 (1.3)		
			30 (1.0)		
		03	38 (2.0)		
		No response	261 (13.5)		
	Sexually transmitted disease (STD) in	Yes	73 (3.8)		
113	the past 6 months	No	1843 (95.3)	Yes	Yes
		No response	18 (0.9)		
		Yes	364 (18.8)		
	Even die en eend with her stitle D	No	1184 (61.2)	Nia	Maa
114	Ever diagnosed with hepatitis B	Unknown	336 (17.4)	INO	res
		No response	50 (2.6)		
		Yes	46 (2 4)		
115	Cirrhosis	No	1888 (97.6)	Yes	Yes
		Ves	5 (0 3)		
116	Ascites	No	1020 (00.7)	Yes	Yes
		No	7 (0,4)		
117	Varices	res	7 (0.4)	Yes	Yes
		NO	1927 (99.6)		
118	Portal hypertension	Yes	6 (0.3)	Yes	Yes
		No	1928 (99.7)		
110	Encenhalonathy	Yes	1 (0.1)	Ves	Vec
113	Encephalopathy	No	1933 (99.9)	163	165
100		Yes	2 (0.1)	Vaa	Vee
120	Hepatocellular carcinoma	No	1932 (99.9)	res	res
		Yes	38 (2.0)		
121	AIDS defining illness	No	1885 (97 5)	Yes	Yes
		No response	11 (0.6)	100	100
-		Voc	27 (1 4)		
100	Cardiovessular diasass	Ne	27 (1.4) 1006 (09 E)	No	Vaa
122	Cardiovascular disease		1906 (96.5)	INO	res
		No response			
		Yes	22 (1.1)	<u>.</u>	
123	Hypercholesterolemia	No	1911 (98.8)	No	Yes
		No response	1 (0.1)		
		Yes	4 (0.2)		
124	Autoimmune disease	No	1928 (99.7)	No	Yes
		No response	2 (0.1)		
46-		Yes	47 (2.4)	. .	
125	Hypertension	No	1885 (97.5)	NO	Yes
L	1				

		No response	2 (0.10		
		Ves	9 (0 5)		
126	Thyroid disease	No	1022 (00 1)	No	Vec
120		No rosponso	2(0.2)	TNO T	163
		Nor			
407	L'estation de	Yes	12 (0.6)	NI.	Mara
127	Lipodystropny	NO	1916 (99.1)	NO	Yes
		No response	6 (0.3)		
		Yes	15 (0.8)		
128	Psoriasis	No	1913 (98.9)	No	Yes
		No response	6 (0.3)		
		Yes	15 (0.8)		
129	Diabetes	No	1914 (98.9)	No	Yes
		No response	5 (0.3)		
		Yes	249 (12.9)		
130	Previous Interferon-based HCV	No	1671 (86.4)	Yes	Yes
100	treatment	No response	14 (0 7)	100	100
	In the past 6 months, how many times		14 (0.1)		
	did you visit the following health				
	Services?	0	4500 (00.4)		
		1-10	325 (16.8)		
131	Walk-in clinic	11-20	11 (0.6)	No	Yes
		21-30	2 (0.1)		
		>30	2 (0.1)		
		No response	6 (0.3)		
		0	1377 (71.2)		
		1-10	537 (27.8)		
132	Emergency room	11-20	9 (0.5)	No	Yes
	0,	21-30	1 (0.1)		
		No response	10 (0.5)		
		0	1616 (83.6)		
		1-10	260 (13 44)		
		11-20	23 (1 2)		
133	Hospital inpatient – overnight	21-30	14(0.7)	No	Yes
		>20	14 (0.7)		
		No rosponso	10 (0.5)		
		0	1176 (60.8)		
		1-10	666 (34.4)		
134	General practitioner	11-20	70 (3.6)	No	Yes
		21-30	9 (0.5)		
		>30	5 (0.3)		
		No response	8 (0.4)		
		0	508 (26.3)		
		1-10	1328 (68.7)		
125	HIV clinic	11-20	78 (4.0)	No	Vee
135		21-30	12 (0.6)	INO	res
		>30	3 (0.2)		
		No response	5 (0.3)		
		0	1231 (63.7)		
		1-10	666 (34 4)		
		11-20	14 (0 70		
136	A specialist	21-30	7 (0.70	No	Yes
		~30	2(0.1)		
		No rooponoo	2(0.1)		
407			14(0.7)	N.I	V
137	APRI for significant liver fibrosis	APKI <1.5	304 (15.7)	NO	Yes

		APRI > 1.5 No response	1509 (78.0) 121 (6.3)		
138	Currently on ARV	Yes No	1809 (93.5) 125 (6.5)	Yes	Yes
139	Food insecurity using the Household Food Security Survey Module (HFSSM)*	Food secure Moderately food insecure Severely food insecure	906 (46.9) 424 (21.9) 604 (31.2)	No	No
		Total number of predictors		46	137

* Used only in the additional analysis.

Abbreviations: IQR: Interquartile range; PCP: Phenylcyclohexyl piperidine; LSD: Lysergic acid diethylamide; MDA: Methylenedioxyamphetamine; EQ-5D-3L: EuroQoL-5Dimension-3Level; IDU: Injection drug use; BMI: Body Mass Index; HIV: Human Immunodeficiency Virus; CD4: Cluster of differentiation 4 receptor; HCV: Hepatitis C virus; RNA: Ribonucleic acid; AIDS: Acquired Immunodeficiency Syndrome; APRI: Aspartate Aminotransferase (AST) to platelet ratio

Appendix B: Introduction to Random Forests

We used the supervised machine learning technique of random forests (RF), an ensemble learning technique developed by Leo Breiman (1, 2). Ensemble learning is based on the idea of combining the strengths of many simpler "base" models. The base models in the case of RF are decision trees, specifically classification and regression trees (CART) for binary and continuous outcomes, respectively (3). RF classification uses bootstrap aggregation of multiple decision trees, combining the predictions from these many trees. A decision tree is made up of nodes (root node, decision node and terminal nodes) and branches. Each node represents a predictor variable, which is chosen from a random subset of all predictor variables, which is a characteristic of the RF algorithm (3). Multiple decision trees are generated from bootstrapped samples of the training data, which is usually about a 2/3 subset of the data. The test set, which is the remaining data, is then run through these trees and the response estimate is the average over all the individual predictions in the forest (1-5).

The main characteristics of RF are:

1) Ensemble technique with multiple decision trees help reduce overfitting (3).

2) Selection of a random subset of predictors from all candidates at each node of which one predictor that most improves accuracy is then selected; the random selection makes each tree more independent of the other and reducing correlation (3, 4)

3) Out-of-Bag (OOB) samples are the observations that are not selected in a given bootstrap resample and using these samples, OOB error is calculated as the average
discrepancy between actual outcome and the outcome predicted by the RF, which provides an additional internal validation (3, 4).

When developing the RF, the parameters of the algorithm are tuned, i.e., a set of parameters are selected which minimize the OOB error and thus maximizing accuracy. These tuning parameters include the number of trees (B), the number of predictors chosen from all candidate predictors at each node (m), and three characteristics of tree depth: node size (s), which is maximum number of observations at each node, maximum terminal nodes (u), and tree level (k), which is maximum number of splits (6).

The algorithm also provides importance metrics to provide an idea which variables were the most influential in the classification algorithm. Overall, RFs are non-parametric, accurate, and relatively robust to outliers and noise. RF variable importance graphs provide insights regarding which variables play a major role in the accurate prediction. Regression RF algorithms can be used to predict for continuous outcomes and classification algorithms can be used to classify into binary or categorical outcomes. Additionally, probability machines for the classification can be used to estimate the probabilities for class membership, which can then be used to determine the class at the optimal probability thresholds (7).

Appendix C: Supplementary methods and results

RF tuning

The RF hyperparameters used to tune the RF algorithms to maximize accuracy were as follows:

- Number of trees (B): Range used: 50-1500
- Number of predictors, randomly chosen from all predictors in the algorithm at each node (mtry): Range used: sqrt(x) to x/2), where x= total number of candidate predictors
- Tree depth, which is maximum number of splits (D): Range used: 1-30 and 0 = no limit
- Node size, which is minimum number of observations at each node (S): Range used: 1-21

The ranges for the hyperparameters were selected based on recommendations in literature (6, 8).

Using the training data, we conducted 10-fold cross validation using grids of the above hyperparameter ranges, the final parameters which maximized accuracy were chosen and these are given in Table 4.6 for all algorithms.

Overall performance

The overall performance was evaluated using a scaled Brier score (BS_{scaled}), which is the Brier score (BS) or mean squared error scaled with the maximum mean squared error in a model that randomly classifies into either CES-D-10 class (BS_{max}): BS_{scaled}=1-

(BS/BS_{max}) (9). Relative efficiency was calculated by comparing the BS_{scaled} for the two algorithms (10, 11).

In the testing data, the full algorithm had a BS_{scaled} of 0.31 (95% CI: 0.22-0.39), that is a 31% reduction in mean squared error as compared to random classification and the reduced algorithm had a BS_{scaled} of 0.20 (95% CI: 0.12-0.27), which means a 20% reduction (10). The relative efficiency comparing the reduced to the full algorithm was 0.65 (95% CI: 0.38-1.01), i.e., a 65% loss in efficiency, indicating better performance of the full algorithm compared to reduced algorithm.

Calibration

Calibration is assessment of the agreement between observed and predicted outcomes. We generated calibration graphs by plotting the predicted probabilities for being in CES-D-10 class=1 (CES-D-10 score >=10) against the observed frequency of CES-D-10 class=1 (11). The Stata package pmcalplot was used to create the calibration graph (12). The calibration graphs are shown in Figure 4.3.

The calibration measure, calibration slope was estimated, with bootstrapped 95% confidence intervals (CI). For the full algorithm, the calibration slope measure was 0.95 (95% CI: 0.72-1.21) for the reduced algorithm and this estimate was close to the ideal calibration slope of 1 for a well-calibrated algorithm (11). Similarly, For the reduced algorithm, the calibration slope measure was 0.97 (95% CI: 0.72-1.21) and this estimate was close to the ideal algorithm, the calibration slope measure was 0.97 (95% CI: 0.72-1.21) and this estimate was close to the ideal calibration slope.

Sr. No.	Algorithms	OOB error	В	mtry	D	S
Prim	Primary analysis					
1	Full algorithm (x = 137)	0.16	600	70	0 (no limit)	1
2	Reduced algorithm (x=46)	0.20	800	20	0 (no limit)	1
Addi	tional analyses					
Α	One visit per individual (x=46)					
1	Full algorithm (x = 137)	0.17	1000	80	10	1
2	Reduced algorithm (x=46)	0.23	100	15	10	5
В	Different CES-D-10 cut-offs					
I	Cut-off - 8					
1	Full algorithm (x = 137)	0.17	100	70	0 (no limit)	5
2	Reduced algorithm (x=46)	0.19	400	10	0 (no limit)	1
II	Cut-off - 13					
1	Full algorithm (x = 137)	0.15	1400	50	0 (no limit)	1
2	Reduced algorithm (x=46)	0.20	1200	20	0 (no limit)	1
III	Cut-off - 15					
1	Full algorithm (x = 137)	0.14	200	30	0 (no limit)	1
2	Reduced algorithm (x=46)	0.17	100	10	20	1
С	With food insecurity variable (x=138)	0.16	100	80	10	1
D	Regression algorithm					
1	Full algorithm (x = 137)	0.24	1400	80	0 (no limit)	1
2	Reduced algorithm (x=46)	0.35	200	15	0 (no limit)	1

Table 4.6: Supplementary table - Final RF hyperparameters for the algorithms

Abbreviations: OOB: Out-Of-Bag sample; B: Number of trees; mtry: Number of predictors chosen randomly at each node; D: tree depth; S: minimum node size; CES-D-10: Center for Epidemiologic Studies Depression Scale-10; EQ-5D-3L: EuroQoL-5Dimension-3Level

Figure 4.3: Supplementary figure - Calibration graphs - A. Full algorithm (x = 137) and

B. Reduced algorithm (x = 46)



A. Full Algorithm (x=137)

Abbreviations: AUC: Area under the Receiver Operating Characteristic curve

Appendix D: Sample generalizability

Table 4.7: Supplementary table - Participant characteristics in visit 1 in the study sample

(n=717) as compared to baseline visit for all CCC participants (n=2008)

Characteristics	Participants (n = 717) $p_{1}(y)$ or modion (IOB)	Participants (n=2008)	
Ace	10(%) or median (IQR)	<u>45 (39, 52)</u>	
Gender - Male	522 (73)	1412 (70)	
Race/Ethnicity	022 (10)	1112 (10)	
Asian	11 (2)	36 (2)	
Black	28 (4)	70 (4)	
White	541 (76)	1401 (70)	
Metis	32 (5)	102 (5)	
First nation	102 (14)	412 (21)	
Hispanic/Latino	7 (1)	29 (1)	
Born outside Canada	64 (9)	181 (9)	
Education - High school and higher	565 (79)	1532 (76)	
Employment - Not employed	525 (73)	1334 (66)	
Monthly income \$1500	543 (76)	1526 (76)	
Revenue Source - Welfare	332 (46)	954 (48)	
Current injection drug use	244 (34)	819 (41)	
Current alcohol use	444 (62)	950 (47)	
Current smoking	534 (75)	1505 (75)	
BMI category - Normal weight (18.5-25 kg/m ²)	312 (44)	798 (40)	
End-stage Liver disease	27 (4)	57 (3)	
HIV clinical stage - A1 (asymptomatic)	248 (35)	438 (22)	
Past AIDS related illness	28 (4)	458 (23)	
Depression diagnosis	68 (10)	742 (37)	
Prescribed antidepressant medications	320 (45)	607 (30)	
HR-QoL using EQ-5D-3L instrument			
Anxiety/depression			
Not anxious or depressed	352 (49)	747 (37)	
Moderately anxious or depressed	302 (42)	818 (41)	
Extremely anxious or depressed	60 (8)	183 (9)	
Current health state (visual analog scale)	70 (56, 80)	70 (50, 80)	

Abbreviations: IQR: Interquartile range; AIDS: Acquired Immunodeficiency Syndrome; HIV: Human Immunodeficiency Virus; BMI: Body Mass Index; HR-QoL: Health related quality of life; EQ-5D-3L: EuroQoL-5Dimension-3Level

References

1. Breiman L. Random Forests. Machine Learning. 2001;45(1):5-32.

2. Fawagreh K, Gaber MM, Elyan E. Random forests: from early developments to recent advancements AU - Fawagreh, Khaled. Systems Science & Control Engineering. 2014;2(1):602-9.

3. Hastie T, Tibshirani R, Friedman JH. The elements of statistical learning: data mining, inference, and prediction. Second edition. ed. New York: Springer; 2009.

4. Mennitt D, Sherrill K, Fristrup K. A geospatial model of ambient sound pressure levels in the contiguous United States. The Journal of the Acoustical Society of America. 2014;135(5):2746-64.

5. Strobl C, Malley J, Tutz G. An introduction to recursive partitioning: rationale, application, and characteristics of classification and regression trees, bagging, and random forests. Psychological methods. 2009;14(4):323-48.

6. Scornet E, Coeurjolly J-Fo, Leclercq-Samson A. Tuning parameters in random forests. ESAIM: Proceedings and Surveys. 2017; 60:144-62.

7. Malley JD, Kruppa J, Dasgupta A, Malley KG, Ziegler A. Probability machines: consistent probability estimation using nonparametric learning machines. Methods Inf Med. 2012;51(1):74-81.

8. Touw WG, Bayjanov JR, Overmars L, Backus L, Boekhorst J, Wels M, et al. Data mining in the Life Sciences with Random Forest: a walk in the park or lost in the jungle? Briefings in bioinformatics. 2013;14(3):315-26.

9. Steyerberg EW, Vickers AJ, Cook NR, Gerds T, Gonen M, Obuchowski N, et al. Assessing the performance of prediction models: a framework for traditional and novel measures. Epidemiology. 2010;21(1):128-38.

10. Wong J, Manderson T, Abrahamowicz M, Buckeridge DL, Tamblyn R. Can Hyperparameter Tuning Improve the Performance of a Super Learner? A Case Study. Epidemiology. 2019;30(4):521-31.

11. Steyerberg EW. Clinical prediction models: a practical approach to development, validation, and updating. Cham, Switzerland: Springer; 2019.

12. Joie E, Kym IES, Emma CM. PMCALPLOT: Stata module to produce calibration plot of prediction model performance. S458486 ed: Boston College Department of Economics; 2018.

5. CHAPTER 5: EFFECT OF DEPRESSIVE SYMPTOMS ON HCV TREATMENT INITIATION

5.1. Preface to Manuscript 2

In this chapter, I aimed to assess the effect of depressive symptoms on HCV treatment initiation. A few studies have addressed depression and other psychiatric illnesses as possible risk factors for reduced linkage to care and treatment in the HCV mono-infection cascade of care (108-110). In a retrospective study of barriers to HCV linkage to care and treatment initiation in Florida, uncontrolled psychiatric illness was found to be a smaller barrier to treatment initiation in the DAA era compared to the IFN era (110). Although these studies assessed depression as a possible barrier for treatment initiation, they did not quantify this effect. Thus, there is limited research regarding the impact that the presence of depressive symptoms may have on initiation of these DAA regimens since their approval in Canada in 2013.

Therefore, in this manuscript, I quantified this effect of depressive symptoms on HCV treatment initiation. I conducted this analysis in both the IFN era (2003-2011) and second generation DAA era (2013-2020) in order to observe possible differences.

This manuscript was published in the journal, Antiviral Therapy in February 2022.

Citation: Marathe G., Moodie E.M., Brouillette M-J., Cox J., Lanièce Delaunay C., Cooper C., Hull M., Gill J., Walmsley S., Pick N., Klein M. B., Canadian Co-Infection Cohort. Are depressive symptoms still a barrier to HCV treatment initiation in the direct acting antiviral era? Antiviral Therapy. February 2022. doi:10.1177/13596535211067610.

This analysis was presented as posters at the following conferences/seminars:

- 1. 17th Annual EBOSS Research Day, McGill University, Virtual March 2021
- 2. 30th Annual Canadian Conference on HIV/AIDS Research, Virtual May 2021
- 3. Canadian Liver Meeting (CLM) 2021, Virtual May 2021
- 4. 5th IDIGH Student Research Day, McGill University, Virtual May 2021
- 5. 11th IAS conference on HIV Science (IAS 2021), Virtual July 2021

5.2. Manuscript 2: Depressive symptoms are no longer a barrier to HCV treatment initiation in the HIV-HCV co-infected population in Canada

Authors and affiliations

Gayatri Marathe ^{1, 2}, Erica E.M. Moodie ¹, Marie-Josée Brouillette ^{2, 3}, Joseph Cox ^{1, 2}, Charlotte Lanièce Delaunay ^{1, 2}, Curtis Cooper ⁴, Mark Hull ⁵, John Gill ⁶, Sharon Walmsley ⁷, Neora Pick ⁸, Marina B. Klein ^{1, 2, 9}, Canadian Co-Infection Cohort

- 1. Department of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada
- 2. McGill University Health Center-Research Institute, Centre for Outcomes Research and Evaluation, Montreal, Quebec, Canada
- 3. Department of Psychiatry, McGill University, Montreal, Quebec, Canada
- 4. Department of Medicine, University of Ottawa, Ottawa, Ontario, Canada
- 5. Centre for Excellence in HIV/AIDS, St. Paul's Hospital, Vancouver, British Columbia, Canada
- 6. Department of Medicine, University of Calgary, Calgary, Alberta, Canada
- 7. Department of Medicine, University of Toronto, Ontario, Canada
- 8. Oak Tree Clinic, BC Women's Hospital, Vancouver, British Columbia, Canada
- 9. CIHR Canadian HIV Trials Network (CTN), Vancouver, British Columbia, Canada

Corresponding author

Marina B. Klein

Division of Infectious Diseases and Chronic Viral Illness Service

McGill University Health Centre

1001 Decarie Boulevard, D02.4110

Montreal, Canada H4A 3J1

Phone: 1-514-843-2090; Fax: 1-514-843-2092

Email address: marina.klein@mcgill.ca

Running head: Depressive symptoms - a HCV treatment barrier?

Abstract

Background

Psychiatric illness was a major barrier for HCV treatment during the Interferon (IFN) treatment era due to neuropsychiatric side effects. While direct acting antivirals (DAA) are better tolerated, patient-level barriers persist. We aimed to assess the effect of depressive symptoms on time to HCV treatment initiation among HIV-HCV co-infected persons during the IFN (2003-2011) and second-generation DAA (2013-2020) eras.

Methods

We used data from the Canadian Co-infection Cohort, a multicentre prospective cohort, and its associated sub-study on Food Security (FS). We predicted Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) classes for depressive symptoms indicative of a depression risk using a random forest classifier and corrected for misclassification using predictive value-based record-level correction. We used marginal structural Cox proportional hazards models with inverse weighting for competing risks (death) to assess the effect of depressive symptoms on treatment initiation among HCV RNA-positive participants.

Results

We included 590 and 1,127 participants in the IFN and DAA eras. The treatment initiation rate increased from 9 (95% confidence interval (CI):7-10) to 21 (95%CI:19-22) per 100 person-years from the IFN to DAA era. Treatment initiation was lower among those with depressive symptoms compared to those without in the IFN era

(hazard ratio: 0.81 (95% CI:0.69-0.95)) and was higher in the DAA era (1.19 (95% CI:1.10-1.27)).

Conclusion

Depressive symptoms no longer appear to be a barrier to HCV treatment initiation in the co-infected population in the DAA era. The higher rate of treatment initiation in individuals with depressive symptoms suggests those previously unable to tolerate IFN are now accessing treatment.

Background

Treatment for Hepatitis C Virus (HCV) has evolved over time, from less effective interferon (IFN)-based regimens to direct acting antiviral (DAA) regimens with cure rates of >95% in real world settings, including among HIV-HCV coinfected patients ¹⁻⁴. IFN-based treatment regimens were long (lasting one year) with weekly injections and had low response rates (on average 30%) and many unpleasant side effects including serious neuropsychiatric outcomes such as mild to severe depression ^{5, 6}. Up to 35% of patients treated with IFN-ribavirin based regimens developed depression ^{7, 8}. Due to these side effects, interferon treatment was relatively contraindicated for patients with current or past major psychiatric illness. Consequently, interferon-based regimens were often not prescribed to those with depression, leading to low treatment rates in this population ⁹. Treatment emergent depression was of particular concerns among people living with HIV as depressive symptoms are associated with decreased adherence to HIV medication, thus modifying the risk-benefit analysis of HCV treatment towards a more conservative approach¹⁰⁻¹³. The second-generation DAA regimens are shorter (8-12 weeks) and have very few adverse effects ¹⁴. Importantly, DAAs have shown few if any psychiatric side effects thus far. Thus, treatment guidelines for HCV have been updated and the presence of current or past psychiatric illness is no longer considered a relative contraindication for treatment ^{15, 16}.

A study in the Canadian Co-infection cohort suggested an almost threefold increase (8 to 28 per 100 person years) in treatment initiation in the HIV-HCV co-infected population after the second-generation DAAs were licensed in Canada in 2013 ¹⁷. However,

treatment initiation rates were lower among people who actively inject drugs, women, and Indigenous populations (5-12 per 100 person-years) compared with other co-infected people ¹⁷. Thus, despite the advantages of the new regimens, treatment initiation is still not the same across subgroups posing a threat to HCV elimination goals¹⁴. Patient- and system-level barriers exist; psychiatric illness, including depression, could be one such barrier.

There is a high prevalence of depression (20-30%) among people living with HIV. Similarly, 24% of those with chronic HCV infection experience depression ¹⁸⁻²¹. Both biological and psychosocial mechanisms are at play ^{18, 22}. Prevalence of depression is reported to be even higher in the co-infected population, which may be due to the co-existence of risk factors and neuropsychiatric effects of both HIV and HCV ²³. The presence of significant depressive symptoms could have an impact on clinical outcomes in patients, even when not meeting diagnostic criteria for major depression.

Studies in both the IFN and DAA eras conducted in people with chronic HCV infection have found psychiatric illness among the barriers to linkage to care, treatment initiation, and adherence ^{9, 24-27}. Whether depressive symptoms continue to prevent treatment initiation in the second generation DAA era is unknown, especially in the co-infected population, which has higher prevalence of depression. The major improvements in HCV treatment safety increases the opportunity to treat people with psychiatric illness and hence we hypothesized improved tolerability would lead to an increase in treatment initiation in this group. Thus, in this study, we evaluated the effect of depressive symptoms

on time to HCV treatment initiation comparing the IFN (2003-2011) and second generation DAA eras (2013-2020) in the HIV-HCV co-infected population in Canada.

Methods

Study population

We used data from the Canadian Co-Infection Cohort (CCC), an open multicentre prospective cohort study, ongoing since 2003. The study has been described in detail elsewhere ²⁸. Briefly, the CCC recruits from 18 HIV urban and semi-urban centres across six Canadian provinces (Quebec, British Columbia, Alberta, Ontario, Nova Scotia, and Saskatchewan). Eligibility criteria include ≥16 years of age, documented HIV infection, and evidence of HCV infection (HCV RNA positive and/or HCV seropositive). The study had recruited 2018 participants as of July 2020. Participants are followed longitudinally, with follow-up visits every six months. Sociodemographic and behavioural data are collected from participants by a standardized self-administered questionnaire at each visit. Clinical data including HIV and HCV treatment dates, medications, co-morbidities, and psychiatric diagnoses are collected via medical chart reviews and HIV and HCV related blood tests performed.

In addition, we used data from an associated sub-study within the CCC, the Food Security and HIV-HCV co-infection study (FS sub-study), to predict the presence of depressive symptoms in the CCC as a whole. Participants for the FS sub-study were recruited from the CCC (n=725) with a maximum of 5 visits integrated into the CCC visits. In the substudy, fully described in detail elsewhere ²⁹, depression screening was performed using

the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) that assesses presence and severity of depressive symptoms in the past week. ³⁰. The scale has 8 items focusing on negative symptoms like feeling restless, fearful, and lonely, feeling bothered by things that usually would not bother you and not being able to concentrate. The remaining 2 items focus on positive symptoms like feeling hopeful about the future and being happy. Each item is measured on a 4-point scale, with reverse scoring for the positive items; score \geq 10 is widely considered to represent the presence of depressive symptoms indicative of being at risk for depression; this cut-off of 10 has been validated in HIV populations in Canada ³⁰⁻³².

Definition of treatment eras and follow-up

The IFN era was defined as the beginning of the CCC on April 28, 2003, until August 1, 2011, when the 1st generation DAA, boceprevir was approved for use by Health Canada. The IFN-free DAA era began at the time that the first 2nd generation DAA, simeprevir, was approved for use by Health Canada November 25, 2013, and continued until end of study period, July 15, 2020. We excluded the period between 2011-2013 in this analysis, because IFN regimens were used widely in combination with the first generation DAA regimens.

Participants were included in the analysis for the IFN era if eligible for treatment i.e., if they tested HCV RNA positive on or after April 28, 2003. Participants were then followed from their first documented positive HCV RNA test in this period (time zero) until treatment initiation or until censoring due to loss to follow-up (no visit for 18 months since last visit),

withdrawal, death, or end of the study period (August 1, 2011). Participants were included in the analysis for the DAA era if eligible for second generation DAA treatment initiation i.e., if they tested HCV RNA positive on or after November 25, 2013. Participants were excluded if they were accessing DAAs through a clinical trial or were treated with IFN. Participants were then followed from their HCV RNA positive test date in this period until second generation DAA treatment initiation or until censoring due to loss to follow-up, withdrawal, death, or end of the study period (July 15, 2020). Participants from the IFN era who remained HCV RNA positive by August 1, 2011, were included in the DAA era analysis if they were still being followed in the CCC and were HCV RNA positive on November 25, 2013.

Measurement

Exposure

The exposure of interest was presence of depressive symptoms indicative of being at risk for major depression, hereafter called depressive symptoms for brevity. CCC participants are not screened for depression during baseline or follow-up. As described earlier, depression screening was however performed in the FS sub-study. Thus, to obtain a measure of depressive symptoms in the full CCC, we developed a random forest (RF) classifier using the CES-D-10 to classify presence/absence of depressive symptoms derived from the FS sub-study as the outcome (target of prediction), and sociodemographic, behavioural, and clinical characteristics from the parent CCC as predictors ³³. The details of the RF classifier development are in Appendix A. Using this RF classifier, the CES-D-10 score ≥10 (presence of depressive symptoms) was predicted

for each CCC visit included in this analysis. We addressed exposure misclassification for the predicted depressive symptoms using the predictive value-based record-level correction method ³⁴. In this method, we applied the positive predictive value (PPV) and negative predictive value (NPV) estimated for the RF algorithm; PPV: 0.74 (95% CI: 0.68-0.80) and NPV: 0.76 (95% CI: 0.69-0.82). The procedure included simulation of corrected exposure at each visit by repeated Bernoulli trials with probability equal to PPV for those classified as CES-D-10 class=1 and 1-NPV for those classified as CES-D-10 class=0 ³⁴.

Outcome

The outcome of interest was time to HCV treatment initiation in each era. The date of treatment initiation in both eras was measured at each visit using exact treatment start and stop dates obtained via medical chart reviews.

Confounders

We considered both baseline and time-varying confounders in this analysis, which were selected a priori based on prior literature, as illustrated in the Directed Acyclic Graph (DAG) shown in figure 1. The baseline confounders included age, gender, race/ethnicity, education level, sexual orientation, previous HCV treatment, immigration status (as a proxy for possible system related factors like medical insurance), marital status, and province (as a proxy for sociodemographic characteristics and changes in treatment policies over time). The time varying confounders measured at each biannual visit included living situation, employment status, monthly income, revenue source, current injection drug use, current alcohol use, current smoking status, incarceration in the past

6 months, advanced fibrosis/cirrhosis (measured by the AST to Platelet Ratio Index (APRI) >= 1.5), detectable HIV viral load (> 50 copies/ml), low CD4 cell count (\leq 250 cells/µl) and antidepressant use. We conducted multiple imputation by chained equations (MICE) to address the missing data in the baseline and time varying confounders ^{35, 36}, as at least one confounder value was missing in > 20% of the included visits in both eras. We created five imputed datasets using linear regression to impute continuous variables, logistic regression for binary variables and multinomial logistic regression for nominal categorical variables.

Statistical analysis

Primary analysis

We estimated the overall treatment initiation rates per 100 person-years in both the IFN and DAA eras. To determine the effect of predicted depressive symptoms on time to treatment initiation, we developed models separately for each era. We fit all models with and without exposure misclassification correction.

First, we developed conventional Cox proportional hazards (Cox PH) model and obtained hazard ratios with 95% confidence intervals (CI), adjusting for baseline and time-varying confounders. The proportionality assumption was assessed by scaled Schoenfeld residuals using a test for zero slope ^{37, 38}. Conventional Cox PH models can yield biased estimators if time-varying covariates act as confounders and mediators simultaneously; that is, if the covariate predicts HCV treatment initiation (outcome) and subsequent presence of depressive symptoms (exposure), but past presence of depressive

symptoms predicts subsequent level of the covariate; DAG in figure 5.1 ³⁹. Thus, we developed marginal structural Cox PH models (MSCM) to address this issue of time-varying confounding ^{39, 40}. In this method, stabilized inverse probability treatment weights (IPTW) were constructed (details in appendix B) ⁴¹. The IPTWs were incorporated in the Cox PH model to obtain the hazard ratio estimates with 95% CIs.

Finally, we needed to consider possible competing risks, specifically deaths due to liver disease, drug overdose or other reasons. In a CCC analysis for the time-period 2013-2017, overall death rates were found to be high, with rates of 23.8 per 1000 person-years (PY) for those aged 20-49 years and 38.3 per 1000 PY among those 50-80 years of age ⁴². A naïve survival analysis would censor individuals at death and assume independence of censoring and event times. This assumption is unlikely to be met in this population ^{43, 44}. To account for death as a competing risk, we calculated inverse probability censoring weights (IPCW) (details in appendix B). We calculated the final weights as the product of IPTW and IPCW and incorporated them in the MSCM model to obtain hazard ratio estimates with 95% CIs.

Secondary analyses

We conducted two planned secondary analyses to assess the robustness of the results. First, we used a restricted subset of participants from the FS sub-study from 2012-2015, majority of which was in the second-generation DAA era, in which measured CES-D-10 scores were available, to measure the effect of exposure misclassification due to the use of predicted depressive symptoms in models. The model development was as in the

primary analysis. Second, we a used a time-fixed exposure at each visit, i.e., baseline predicted CES-D-10 class and developed the conventional Cox models for the IFN and DAA eras separately.

Results

Participant characteristics

The flowcharts for participants included in the final analytical samples are shown in figure 5.2.: 2A - IFN era, and 2B - DAA era. We included 590 and 1,127 of HCV RNA-positive participants in the IFN and DAA eras, respectively. The baseline characteristics of the participants are shown in table 5.1. The proportion of individuals with predicted depressive symptoms, i.e., CES-D-10 scores \geq 10, was high in both the IFN (55%) and DAA (60%) eras at baseline. Participants in both IFN and DAA eras were both predominantly male (77%; 70%) and not employed (70%; 68%). In the DAA era, there were comparatively higher proportions of Indigenous participants (29% vs. 10%), current injection drug users (44% vs. 31%), those recently incarcerated (16% vs 8%) and those with lower levels of education (73% vs. 68% with no post-secondary education) compared with the IFN era, hence more potential barriers to HCV care. A small proportion of the participants in the IFN era (14%) and DAA era (11%) were receiving antidepressants, however whether they were prescribed for depression or for other disorders was not known.

Primary analyses

There were 126 and 566 treatment initiations with median follow-up times of 2.3 years (range: 0.02-8.3 years) and 2.0 years (range: 0.01-6.6 years) in the IFN and DAA eras, respectively. The overall treatment initiation rate increased from 9 (95%CI: 7-10) per 100 person-years (PY) in the IFN era to 21 (95%CI: 19-22) per 100 PY in the DAA era. The primary analyses results are shown in table 5.2. There was no evidence to suggest the proportionality assumption was violated. The hazard ratios were similar across the three primary models in each era. In the IFN era, the MSCM model accounting for competing risks showed lower treatment initiation among those with depressive symptoms compared to those without (hazard ratio (HR): 0.63 (95% CI: 0.43-0.93)). Exposure misclassification correction moved the HR towards the null (0.81 (95% CI: 0.69-0.95)), but still indicated a 19% lower risk of initiation among those with depressive symptoms. In the DAA era, in the MSCM model accounting for competing risks, we observed higher treatment initiation among those with depressive symptoms compared to those without (HR: 1.42 (95% CI: 1.17-1.71)). Again, exposure misclassification correction moved the HR towards the null (1.19 (95% CI: 1.10-1.27)), still indicating a 19% higher risk of initiation among those with depressive symptoms.

Secondary analyses

The secondary analyses results are shown in table 5.3. In the first analysis, in the restricted subset with measured CES-D-10 classes available from the FS sub-study, 485 participants were included with 102 treatment initiations, all in the DAA era. Taking into consideration time-varying confounders and competing risks, the HR for treatment

initiation was 0.82 (95% CI: 0.52-1.29). While this point estimate was comparable to the IFN era estimates, the precision was low. In the second analysis, using time-fixed exposure (baseline depressive symptoms), the results were comparable to the primary analyses, with lower treatment initiation among those with depressive symptoms than those without in the IFN era and higher in the DAA era.

Discussion

In this multicentre prospective cohort study, we observed lower HCV treatment initiation among those with depressive symptoms compared to those without in the IFN era whereas higher initiation among those with depressive symptoms in the DAA era. There was a high prevalence of depressive symptoms (over 50% in both eras); thus, the ability of DAAs to overcome a significant barrier to HCV treatment historically faced by this substantial population should help the goal of eliminating HCV.

Our results are in line with expectations that, during the IFN era, providers would be less likely to prescribe IFN-based regimens to those with depressive symptoms and that neuropsychiatric side-effects of IFN would deter such patients from initiating therapy. A multicentre cohort study of veterans in the US similarly found that those with pre-existing psychiatric conditions (18%) had a much higher odds of not being treated (OR: 9.62; 95% CI: 6.85-13.50) ⁴⁵. In the DAA era, there have been studies examining depression prevalence before and after DAA treatment ^{15,} ⁴⁶⁻⁴⁹. A few studies have addressed depression and other psychiatric illness as

possible risk factors for reduced linkage to care and treatment in the HCV monoinfection cascade of care ²⁴⁻²⁶. Higher Patient Health questionnaire-8 (PHQ-8) scores indicating more severe depression have been found among those not treated for HCV²⁴. In a retrospective study of barriers to HCV linkage to care and treatment initiation in Florida, uncontrolled psychiatric illness was found to be a smaller barrier to treatment initiation in the DAA era compared to the IFN era ²⁶. Though these studies assessed depression as a possible barrier for treatment initiation, they did not explore the effect of depressive symptoms independently. In our study, we present the first quantitative estimates of the effect of depressive symptoms on time to HCV treatment initiation, contrasting uptake in the IFN and second-generation DAA eras among HIV-HCV co-infected people, who may differ from those with HCV mono-infection due to distinct patient- and system-level barriers to care. Unlike other studies, we accounted for potential time-dependent confounding and competing risk of death which could bias estimates of treatment uptake.

An overall increase in HCV treatment initiation rates was observed from the IFN to the DAA era. The percentage of eligible participants that initiated treatment increased from 21% to 50% in this analysis; this is important progress towards reaching the HCV elimination goals in Canada ⁵⁰. Additionally, it was encouraging that this increase occurred in the DAA era in spite of higher prevalence of many possible barriers to treatment such as lower education, higher rates of injection drug use and incarceration and being Indigenous – barriers that may themselves be linked with depression.

Interestingly, we not only observed that treatment initiation increased among those with depressive symptoms, which was anticipated, but also that in the DAA era, initiation was found to be higher among those with depressive symptoms compared those without, which was surprising. One of the possible explanations for this could be that a backlog of patients for whom IFN-based treatments were contraindicated, or were not tolerated, are now accessing treatment, i.e., a warehousing effect. This effect could either persist or plateau over time, which would be expected with a warehousing effect. In our study, we observed that almost 65% of the people with baseline depressive symptoms initiated DAAs between 2015-2017 and comparatively lower number of people (31%) initiated between 2018-2020, which provides some evidence for possible plateauing, however additional data would be needed to confirm this change over time. With the percentage of the eligible participants initiating HCV treatment still being much lower than the WHO elimination goal of 80% and possible plateauing, it is possible that depressive symptoms may have additional impacts on the care cascade beyond medication safety that need to be addressed to further enhance treatment uptake. This might be achieved, for example, by integration of mental health education, screening, and treatment as a part of routine care for HIV-HCV co-infected people.

This study has several strengths. The CCC recruits participants from primary and tertiary care clinics in urban and semi-urban areas across Canada. Thus, our estimates are generalizable to HIV-HCV co-infected patients engaged in care in Canada. We leveraged longitudinally collected data over 15 years, applied robust methods to account for time-dependent confounding, competing risk and conducted multiple

secondary analyses. However, the study does have limitations. The clinically relevant depressive symptoms were predicted via a random forest algorithm, and not measured directly in the cohort. Thus, misclassification was expected and therefore we corrected for potential misclassification in our analysis, resulting in attenuation of our estimates. We predicted the depressive symptoms based on a screening questionnaire, CES-D-10, and not a major depression diagnosis. Thus, this study does not provide an effect estimate for depression but rather for presence of these clinically relevant depressive symptoms.

We demonstrated a high baseline prevalence of depressive symptoms in the HIV-HCV co-infected people and showed a substantial increase in treatment initiation rates with the availability of DAAs. Our study suggests that depressive symptoms are no longer a barrier to HCV treatment initiation in the second-generation DAA era in the HIV-HCV co-infected population in Canada.

Acknowledgements

We would like to acknowledge the participants of the Canadian Co-Infection Cohort (CTN222), the study coordinators and nurses for their assistance with study coordination, participant recruitment, and care, and the Canadian Co-Infection Cohort (CTN222) co-investigators - Drs. Lisa Barrett, Jeff Cohen, Brian Conway, Curtis Cooper, Pierre Côté, Joseph Cox, M. John Gill, Shariq Haider, David Haase, Mark Hull, Valérie Martel-Laferrière, Julio Montaner, Erica E. M. Moodie, Neora Pick, Danielle Rouleau, Aida Sadr, Steve Sanche, Roger Sandre, Mark Tyndall, Marie-Louise Vachon, Sharon Walmsley and Alexander Wong.

Ethics approval and consent to participate

This study was approved by the Research Ethics Board of the McGill University Health Centre (2021-6985). The CCC and the FS Sub-Study were approved by the Research Ethics Board of the McGill University Health Centre (2006-1875, BMB-06-006t, 2013-994) and the research ethics boards of participating institutions. The study was conducted according to the Declaration of Helsinki. Informed consent was obtained from all individual participants included in the study.

Data Availability

The datasets generated and/or analysed during the current study are not publicly available. According to stipulations of the patient consent form signed by all study participants, ethical restrictions imposed by our Institutional Ethics review boards, and

legal restrictions imposed by Canadian law, anonymized data are available upon request by contacting Dr. Marina B. Klein.

Funding

This work was supported by Fonds de recherche du Québec-Santé; Réseau sida/maladies infectieuses, the Canadian Institute for Health Research (CIHR; FDN-143270); and the CIHR Canadian HIV Trials Network (CTN222 & CTN264). GM is supported by the PhD trainee fellowship from the Canadian Network on Hepatitis C. EEMM is supported by a chercheur boursier de mérite award from the Fonds de recherche du Québec-Santé and a Canada Research Chair Tier 1. MBK is supported by a Tier I Canada Research Chair. For the remaining authors none were declared. The funders had no role in the production of this manuscript.

Declaration of Conflicting Interests

MBK reports grants for investigator-initiated studies from ViiV Healthcare, AbbVie, Merck, and Gilead; and consulting fees from ViiV Healthcare, Merck, AbbVie, and Gilead. JC received grants and consulting fees from ViiV Healthcare, Merck, and Gilead and personal fees from Bristol-Myers Squibb. CC has received personal fees for being a member of the national advisory boards of Gilead, Merck, Janssen, and Bristol-Myers Squibb. MH has served as a consultant for Merck, Vertex Pharmaceuticals, Pfizer, Viiv Healthcare, and Ortho-Jansen. MH has also received grants from the National Institute on Drug Abuse, as well as payment for lectures from Merck and Ortho-Janssen. JG has served as ad hoc member on National HIV advisory boards to ViiV healthcare, Gilead, and Merck. SW received grants, consulting fees, lecture fees, nonfinancial support, and

fees for the development of educational presentations from Merck, ViiV Healthcare, GlaxoSmithKline, Pfizer, Gilead, AbbVie, Bristol-Myers Squibb, and Janssen. NP reports honoraria from Gilead and ViiV Healthcare. GM, EEM, MJB, and CLD have no conflict of interest to disclose.

	IFN era (2003-2011)	DAA era (2013-2020)		
Baseline characteristics	Participants (n = 590)	Participants (n = 1127)		
	n (%) or median (IQR)	n (%) or median (IQR)		
Predicted CES-D-10 class - 1: score >=10	325 (55)	678 (60)		
Age (years)	45 (40-50)	45 (38 -51)		
Gender - Male	456 (77)	786 (70)		
Race/ethnicity				
White	472 (80)	722 (64)		
Indigenous (First Nations, Inuit, and	59 (10)	224 (20)		
Metis)		324 (29)		
Asian	13 (2)	18 (2)		
Black	27 (5)	36 (3)		
Hispanic/Latino	9 (2)	20 (2)		
Living situation - Homeless	60 (10)	122 (11)		
Education - High school educated and less	401 (68)	827 (73)		
Employment - Not employed	413 (70)	769 (68)		
Monthly income - <= \$1500 CAD	430 (73)	876 (78)		
Revenue source - Welfare	265 (45)	561 (50)		
Sexual orientation - Heterosexual	418 (71)	792 (70)		
Immigrant to Canada	65 (11)	92 (8)		
Marital status - Single	376 (64)	787 (70)		
Previous IFN-based HCV treatment	96 (16)	154 (14)		
Injection drug use in the past 6 months	184 (31)	490 (44)		
Alcohol use in the past 6 months	313 (53)	543 (48)		
Smoking in the past 6 months	430 (73)	847 (75)		
Incarceration in the past 6 months	49 (8)	176 (16)		
Fibrosis stage (e.g., APRI) >= 1.5	143 (24)	192 (17)		
HIV viral load > 50 copies/ml	209 (35)	358 (32)		
CD4 count - <= 250 cells/uL	153 (26)	266 (24)		
Antidepressant prescribed	81 (14)	120 (11)		

Table 5.1: Baseline characteristics for participants included from IFN and DAA eras

Abbreviations: IFN: Interferon; DAA: Direct acting antivirals; IQR: Interquartile range; CES-D-10: Center for Epidemiologic Studies Depression Scale-10; CAD: Canadian dollar; HCV: Hepatitis C virus; APRI: AST to Platelet Ratio Index; HIV: Human Immunodeficiency Virus; CD4: Cluster of differentiation 4 receptor

Table 5.2: Effect of depressive symptoms on time to HCV treatment initiation in IFN and

DAA eras

Madala	No misclassification correction		Misclassification correction		
Models	HR	95% CI	HR	95% CI	
IFN era (2003-2011)					
Cox proportional hazards model	0.66	0.43-0.99	0.84	0.72-0.99	
MSCM not accounting for competing risks **	0.64	0.43-0.94	0.82	0.70-0.96	
MSCM accounting for competing risks ***	0.63	0.43-0.93	0.81	0.69-0.95	
DAA era (2013-2020)					
Cox proportional hazards model	1.53	1.27-1.84	1.20	1.12-1.29	
MSCM not accounting for competing risks **	1.37	1.14 -1.66	1.17	1.09-1.25	
MSCM accounting for competing risks ***	1.42	1.17-1.71	1.19	1.10-1.27	

* Conventional Cox proportional hazards model - Biased estimate due to time varying confounders acting as mediators

** Marginal structural Cox proportional hazards model adjusting for confounding bias due to time-varying confounders acting as mediators

*** Marginal structural Cox proportional hazards model with inverse probability censoring weights to adjust for death as a competing risk

All models were adjusted for following Baseline confounders: age, gender, race/ethnicity, education level, sexual orientation, previous IFN-based HCV treatment, immigration, marital status, and province; Time varying confounders: living situation, employment, income, revenue source, injection drug use, alcohol use, smoking, incarceration, fibrosis stage (e.g., AST to Platelet Ratio Index (APRI) ≥1.5), HIV viral load, CD4 count and antidepressant use.

Abbreviations: IFN: Interferon; DAA: Direct acting antivirals; HR: Hazard ratio; CI: Confidence interval; MSCM; Marginal structural Cox proportional hazards model

Table 5.3: Secondary analyses using measured CES-D-10 classes in a restricted subset

Models	No misclassi	fication correction	Misclassification correction			
incucio	HR	95% CI	HR	95% CI		
Restricted subset with measured CES-D-10 classes (2012-2015) *						
Cox proportional hazards model	0.88	0.57-1.35	-	-		
MSCM not accounting for competing risks	0.90	0.57-1.40	-	-		
MSCM accounting for competing risks	0.82	0.52-1.29	-	-		
Exposure: Baseline predicted depressive symptoms **						
IFN era (2003-2011)	0.78	0.50-1.23	0.92	0.77-1.08		
DAA era (2013-2020)	1.17	0.97-1.42	1.12	1.05-1.20		

and baseline predicted depressive symptoms

* The subset was from the FS sub-study in which data was collected from 2012-2015 during the DAA era. Models were adjusted for - Baseline confounders: age, gender, race/ethnicity, education level, sexual orientation, previous IFN-based HCV treatment, immigration, marital status, and province; Time varying confounders: living situation, employment, income, revenue source, injection drug use, alcohol use, smoking, incarceration, fibrosis stage (e.g., AST to Platelet Ratio Index (APRI) ≥1.5), HIV viral load, CD4 count and antidepressant use.

** Models were adjusted for baseline confounders: age, gender, race/ethnicity, education level, sexual orientation, previous IFN-based HCV treatment, immigration, marital status, and province

Abbreviations: IFN: Interferon; DAA: Direct acting antivirals; HR: Hazard ratio; CI: Confidence interval; CES-D-10: Center for Epidemiologic Studies Depression Scale-10; MSCM; Marginal structural Cox proportional hazards model

Figure 5.1: Directed Acyclic Graph for effect of depressive symptoms on HCV treatment initiation



Figure 5.2: Flowcharts of participants included in the analytical samples - A. IFN era (2003-2011) and B. DAA era (2013-2020)



Abbreviations: IFN: Interferon; HCV: Hepatitis C virus; DAA: Direct acting antivirals; peg-IFN: Pegylated interferon; CCC: Canadian Co-infection Cohort
References

1. Burstow NJ, Mohamed Z, Gomaa AI, et al. Hepatitis C treatment: where are we now? *International journal of general medicine* 2017; 10: 39-52. 2017/03/04. DOI: 10.2147/ijgm. s127689.

2. Webster DP, Klenerman P and Dusheiko GM. Hepatitis C. *Lancet (London, England)* 2015; 385: 1124-1135. 2015/02/18. DOI: 10.1016/s0140-6736(14)62401-6.

3. Hull M, Shafran S, Wong A, et al. CIHR Canadian HIV Trials Network Coinfection and Concurrent Diseases Core Research Group: 2016 Updated Canadian HIV/Hepatitis C Adult Guidelines for Management and Treatment. *Can J Infect Dis Med Microbiol* 2016; 2016: 4385643. 2016/07/30. DOI: 10.1155/2016/4385643.

4. Wyles DL, Sulkowski MS and Dieterich D. Management of Hepatitis C/HIV Coinfection in the Era of Highly Effective Hepatitis C Virus Direct-Acting Antiviral Therapy. *Clin Infect Dis* 2016; 63 Suppl 1: S3-s11. 2016/07/02. DOI: 10.1093/cid/ciw219.

5. Fried MW. Side effects of therapy of hepatitis C and their management. *Hepatology* 2002; 36: S237-244. 2002/10/31. DOI: 10.1053/jhep.2002.36810.

6. Schaefer M, Engelbrecht MA, Gut O, et al. Interferon alpha (IFNalpha) and psychiatric syndromes: a review. *Progress in neuro-psychopharmacology & biological psychiatry* 2002; 26: 731-746.

7. McHutchison JG, Lawitz EJ, Shiffman ML, et al. Peginterferon alfa-2b or alfa-2a with ribavirin for treatment of hepatitis C infection. *N Engl J Med* 2009; 361: 580-593. 2009/07/25. DOI: 10.1056/NEJMoa0808010.

8. Loftis JM and Hauser P. The phenomenology and treatment of interferon-induced depression. *Journal of Affective Disorders* 2004; 82: 175-190. DOI: https://doi.org/10.1016/j.jad.2004.04.002.

9. Fleming CA, Tumilty S, Murray JE, et al. Challenges in the Treatment of Patients Coinfected with HIV and Hepatitis C Virus: Need for Team Care. *Clinical Infectious Diseases* 2005; 40: S349-S354. DOI: 10.1086/427452.

10. Levintow SN, Pence BW, Powers KA, et al. Depression, antiretroviral therapy initiation, and HIV viral suppression among people who inject drugs in Vietnam. *Journal of Affective Disorders* 2021; 281: 208-215. DOI: https://doi.org/10.1016/j.jad.2020.12.024.

11. Horberg MA, Silverberg MJ, Hurley LB, et al. Effects of Depression and Selective Serotonin Reuptake Inhibitor Use on Adherence to Highly Active Antiretroviral Therapy and on Clinical Outcomes in HIV-Infected Patients. *JAIDS Journal of Acquired Immune Deficiency Syndromes* 2008; 47: 384-390. DOI: 10.1097/QAI.0b013e318160d53e.

12. Gonzalez JS, Batchelder AW, Psaros C, et al. Depression and HIV/AIDS treatment nonadherence: a review and meta-analysis. *Journal of acquired immune deficiency syndromes (1999)* 2011; 58: 181-187. DOI: 10.1097/QAI.0b013e31822d490a.

13. Bengtson AM, Pence BW, Mimiaga MJ, et al. Depressive Symptoms and Engagement in Human Immunodeficiency Virus Care Following Antiretroviral Therapy Initiation. *Clinical Infectious Diseases* 2018; 68: 475-481. DOI: 10.1093/cid/ciy496.

14. *Guidelines for the care and treatment of persons diagnosed with chronic hepatitis C virus infection.* Report no. License: CC BY-NC-SA 3.0 IGO, 2018. Geneva: World Health Organization.

15. Gallach M, Vergara M, da Costa JP, et al. Impact of treatment with direct-acting antivirals on anxiety and depression in chronic hepatitis C. *PLOS one* 2018; 13: e0208112-e0208112. DOI: 10.1371/journal.pone.0208112.

16. Sundberg I, Lannergård A, Ramklint M, et al. Direct-acting antiviral treatment in real world patients with hepatitis C not associated with psychiatric side effects: a prospective observational study. *BMC Psychiatry* 2018; 18: 157. DOI: 10.1186/s12888-018-1735-6.

17. Saeed S, Strumpf EC, Moodie EE, et al. Disparities in direct acting antivirals uptake in HIV-hepatitis C co-infected populations in Canada. *Journal of the International AIDS Society* 2017; 20 2017/11/09. DOI: 10.1002/jia2.25013.

18. Nanni MG, Caruso R, Mitchell AJ, et al. Depression in HIV infected patients: a review. *Current psychiatry reports* 2015; 17: 530. 2014/11/22. DOI: 10.1007/s11920-014-0530-4.

19. Younossi Z, Park H, Henry L, et al. Extrahepatic Manifestations of Hepatitis C: A Meta-analysis of Prevalence, Quality of Life, and Economic Burden. *Gastroenterology* 2016; 150: 1599-1608. DOI: 10.1053/j.gastro.2016.02.039.

20. Bing EG, Burnam MA, Longshore D, et al. Psychiatric disorders, and drug use among human immunodeficiency virus-infected adults in the United States. *Arch Gen Psychiatry* 2001; 58: 721-728. 2001/09/06. DOI: 10.1001/archpsyc.58.8.721.

21. Ciesla JA and Roberts JE. Meta-analysis of the relationship between HIV infection and risk for depressive disorders. *Am J Psychiatry* 2001; 158: 725-730. 2001/05/01. DOI: 10.1176/appi.ajp.158.5.725.

22. Yeoh SW, Holmes ACN, Saling MM, et al. Depression, fatigue, and neurocognitive deficits in chronic hepatitis C. *Hepatology international* 2018 2018/06/23. DOI: 10.1007/s12072-018-9879-5.

23. Fialho R, Pereira M, Rusted J, et al. Depression in HIV and HCV co-infected patients: a systematic review and meta-analysis. *Psychology, health & medicine* 2017; 22: 1089-1104. 2017/01/20. DOI: 10.1080/13548506.2017.1280177.

24. Spradling PR, Zhong Y, Moorman AC, et al. Psychosocial Obstacles to Hepatitis C Treatment Initiation Among Patients in Care: A Hitch in the Cascade of Cure. *Hepatology Communications* 2021; 5: 400-411. DOI: https://doi.org/10.1002/hep4.1632.

25. Nguyen P, Vutien P, Hoang J, et al. Barriers to care for chronic hepatitis C in the direct-acting antiviral era: a single-centre experience. *BMJ Open Gastroenterol* 2017; 4: e000181-e000181. DOI: 10.1136/bmjgast-2017-000181.

26. Malespin M, Harris C, Kanar O, et al. Barriers to treatment of chronic hepatitis C with direct acting antivirals in an urban clinic. 2019; 18: 304-309. DOI: 10.1016/j.aohep.2018.06.001.

27. Chen Y-C, Thio CL, Cox AL, et al. Trends in hepatitis C treatment initiation among HIV/hepatitis C virus-coinfected men engaged in primary care in a multisite community health centre in Maryland: a retrospective cohort study. *BMJ Open* 2019; 9: e027411. DOI: 10.1136/bmjopen-2018-027411.

28. Klein MB, Saeed S, Yang H, et al. Cohort profile: the Canadian HIV-hepatitis C coinfection cohort study. *International journal of epidemiology* 2010; 39: 1162-1169. 2009/09/30. DOI: 10.1093/ije/dyp297.

29. Cox J, Hamelin AM, McLinden T, et al. Food Insecurity in HIV-Hepatitis C Virus Co-Infected Individuals in Canada: The Importance of Co-morbidities. *AIDS and behavior* 2017; 21: 792-802. 2016/02/26. DOI: 10.1007/s10461-016-1326-9.

30. Radloff LS. The CES-D Scale: A Self-Report Depression Scale for Research in the General Population. *Applied Psychological Measurement* 1977; 1: 385-401. DOI: 10.1177/014662167700100306.

31. Andresen EM, Malmgren JA, Carter WB, et al. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). *American journal of preventive medicine* 1994; 10: 77-84. 1994/03/01.

32. Zhang W, O'Brien N, Forrest JI, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. *PLOS one* 2012; 7: e40793. 2012/07/26. DOI: 10.1371/journal.pone.0040793.

33. Breiman L. Random Forests. *Machine Learning* 2001; 45: 5-32. DOI: 10.1023/A:1010933404324.

34. Banack HR, Stokes A, Fox MP, et al. Stratified Probabilistic Bias Analysis for Body Mass Index-related Exposure Misclassification Postmenopausal in Women. 604-613. Epidemiology (Cambridge, Mass) 2018; 29: DOI: 10.1097/EDE.000000000000863.

35. Raghunathan TE, Lepkowski JM, Van Hoewyk J, et al. A Multivariate Technique for Multiply Imputing Missing Values Using a Sequence of Regression Models. *SURVEY METHODOLOGY* 2001; 27: 85-96.

36. van Buuren S. Multiple imputation of discrete and continuous data by fully conditional specification. *Stat Methods Med Res* 2007; 16: 219-242. 2007/07/11. DOI: 10.1177/0962280206074463.

37. Abrahamowicz M, MacKenzie T and Esdaile JM. Time-Dependent Hazard Ratio: Modeling and Hypothesis Testing with Application in Lupus Nephritis. *Journal of the American Statistical Association* 1996; 91: 1432-1439. DOI: 10.2307/2291569.

38. GRAMBSCH PM and THERNEAU TM. Proportional hazards tests and diagnostics based on weighted residuals. *Biometrika* 1994; 81: 515-526. DOI: 10.1093/biomet/81.3.515.

39. Robins JM, Hernan MA and Brumback B. Marginal structural models and causal inference in epidemiology. *Epidemiology (Cambridge, Mass)* 2000; 11: 550-560. 2000/08/24.

40. Hernan MA, Brumback B and Robins JM. Marginal structural models to estimate the causal effect of zidovudine on the survival of HIV-positive men. *Epidemiology (Cambridge, Mass)* 2000; 11: 561-570. 2000/08/24.

41. Moodie EE and Stephens DA. Marginal Structural Models: unbiased estimation for longitudinal studies. *International journal of public health* 2011; 56: 117-119. 2010/10/12. DOI: 10.1007/s00038-010-0198-4.

42. Kronfli N, Bhatnagar SR, Hull MW, et al. Trends in cause-specific mortality in HIV-Hepatitis C co-infection following hepatitis C treatment scale-up. *AIDS (London, England)* 2019 2019/01/17. DOI: 10.1097/qad.00000000002156.

43. Lau B, Cole SR and Gange SJ. Competing risk regression models for epidemiologic data. *American journal of epidemiology* 2009; 170: 244-256. 2009/06/06. DOI: 10.1093/aje/kwp107.

44. Andersen PK, Geskus RB, de Witte T, et al. Competing risks in epidemiology: possibilities and pitfalls. *International journal of epidemiology* 2012; 41: 861-870. 2012/01/19. DOI: 10.1093/ije/dyr213.

45. Bini EJ, Bräu N, Currie S, et al. Prospective multicenter study of eligibility for antiviral therapy among 4,084 U.S. veterans with chronic hepatitis C virus infection. *The American journal of gastroenterology* 2005; 100: 1772-1779.

46. Nardelli S, Riggio O, Rosati D, et al. Hepatitis C virus eradication with directly acting antivirals improves health-related quality of life and psychological symptoms. *World J Gastroenterol* 2019; 25: 6928-6938. 2020/01/08. DOI: 10.3748/wjg. v25.i48.6928.

47. Miarons M, Sánchez-Ulayar A, Sempere G, et al. New direct-acting antivirals for hepatitis C treatment and neuropsychiatric symptoms in psychiatric risk groups. *Eur J Hosp Pharm* 2019; 26: 135-139. 2019/08/21. DOI: 10.1136/ejhpharm-2017-001352.

48. Khalil MA, Shousha HI, EI-Nahaas SM, et al. Depression in patients with chronic hepatitis-C treated with direct-acting antivirals: A real-world prospective observational study. *Journal of Affective Disorders* 2021; 282: 126-132. DOI: 10.1016/j.jad.2020.12.128.

49. Abdel Moez AT, El Hawary YA and Al Balakosy AM. Can successful treatment by direct-acting antivirals improve depression in chronic HCV patients? *Eur J Gastroenterol Hepatol* 2020 2020/06/20. DOI: 10.1097/meg.000000000001790.

50. The Canadian Network on Hepatitis C Blueprint Writing Committee and Working Groups. *Blueprint to inform hepatitis C elimination efforts in Canada.* Montreal, QC.

5.3. Manuscript 2: Appendix

Appendix A: Random Forest algorithm for depressive symptoms prediction

Appendix B: Construction of stabilized inverse probability treatment weights and stabilized inverse probability censoring weights

Appendix A: Random Forest algorithm for depressive symptoms prediction

Depression screening was not performed in the Canadian Co-infection Cohort (CCC) and hence a measure of depressive symptoms was not available. Thus, in our previous work (submitted to the Journal of Psychosomatic Research), we developed an algorithm to predict the presence of depressive symptoms indicative of risk of major depression using data from a sub-study within the CCC, the Food Security and HIV-HCV co-infection study (FS sub-study)^{1, 2}. In the FS sub-study, depression screening was conducted using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10)^{3, 4}. The CES-D-10 is a 10-item Likert scale questionnaire that assesses presence and severity of depressive symptoms in the past week, with a total score range of 0-30. We dichotomized the score at 10 to create the CES-D-10 classes (0/1), as a score \geq 10 is widely considered to be indicative of being at risk for major depression with clinically relevant depressive symptoms.⁴ Both the scale and the dichotomization at 10 have been validated in HIV populations in Canada⁵.

We used the supervised machine learning technique of Random Forests (RF), an ensemble learning approach which uses bootstrap aggregation of multiple decision trees, combining predictions from these many trees ⁶⁻¹⁰. We developed probability machines to estimate the CES-D-10 class probabilities at each visit and then determined the CES-D-10 class at the 0.5 probability threshold ¹¹. The algorithm used 137 candidate predictors available in the CCC, selected based on the literature and subject matter expertise. Predictors fell into one of five major categories: questions related to mental health, health related quality of life, sociodemographic, behavioral, and clinical characteristics. The

performance evaluation characteristics of the algorithm are shown in table 5.4. The algorithm performed well, with an 82% (95% CI: 78-86) chance of distinguishing between the two CES-D-10 classes.

Table 5.4: Supplementary table - Performance evaluation measures for the RF

algorithm

Evaluation measures (95% CI)	RF Algorithm (x=137)
AUC	0.82 (0.78-0.86)
Sensitivity	0.77 (0.70-0.83)
Specificity	0.73 (0.66-0.79)
PPV	0.74 (0.68-0.80)
NPV	0.76 (0.69-0.82)
LR +	2.8 (2.2-3.6)
LR -	0.3 (0.2-0.4)

Abbreviations: CI: Confidence interval; AUC: Area under the Receiver Operating Characteristic curve; PPV: Positive Predictive Value; NPV: Negative Predictive Value; LR +: Positive likelihood ratio; LR -: Negative likelihood ratio

Appendix B: Construction of stabilized inverse probability treatment weights and stabilized inverse probability censoring weights

Stabilized inverse probability treatment weights

We aimed to assess the effect of the predicted depressive symptoms on time to HCV treatment initiation. We thus developed Cox proportional hazards models to obtain this estimate since exact date of treatment initiation was available. However, due to the presence of time-varying covariates, there is a possibility of some of the covariates to be confounders and mediators simultaneously. In this case, adjustment in a conventional Cox PH model can lead to biased estimates. Thus, in addition to the conventional model, we developed marginal structural Cox PH models (MSCM) to address time-varying confounding ^{12, 13}. The weighting in marginal structural models (MSMs) creates a pseudo-population, where the exposure is independent of the measured baseline and time varying confounders, permitting unbiased estimation of the joint effects of time-dependent exposures under the causal assumptions. Stabilized inverse probability treatment weights (IPTW) were constructed. The stabilized IPTW for patient *i* at visit *t* is shown in equation (1) below.

$$w^{s} = \frac{\prod_{t=0}^{T} P(E_{it} = e_{it} | E_{i(t-1)} = e_{i(t-1)}, X_{i0} = x_{i0})}{\prod_{t=0}^{T} P(E_{it} = e_{it} | E_{i(t-1)} = e_{i(t-1)}, X_{i0} = x_{i0}, V_{i(t-1)} = v_{i(t-1)})} \dots (1)$$

The numerator was the estimated probability of observed exposure (E_i) for each time interval given baseline covariates (X_0) and past exposure (E_{t-1}), and the denominator was the estimated probability of observed exposure for the time interval given baseline covariates (X_0), past exposure (E_{t-1}), and time-varying covariates (V_i). These weights were obtained by fitting a logistic regression model to obtain the probability of CES-D-10 class=1 (significant depressive symptoms presence) at each visit. The weights were then truncated at the 99th percentile. Summary statistics for the stabilized IPTWs are given in table 5.5.

Stabilized inverse probability censoring weights

A naïve survival analysis would censor individuals at death and assumes independence of censoring and event times. This assumption would be unlikely to be met in this population. Thus, stabilized inverse probability censoring weights (IPCW) were calculated. The stabilized IPCW for death for patient *i* at visit *t* is shown in equation (2) below.

$$w^{s} = \frac{\prod_{t=0}^{T} P(C_{it}=0|C_{i(t-1)}=0, E_{i(t-1)}=e_{i(t-1)}, X_{i0}=x_{i0})}{\prod_{t=0}^{T} P(C_{it}=0|C_{i(t-1)}=0, E_{i(t-1)}=e_{i(t-1)}, X_{i0}=x_{i0}, V_{i(t-1)}=v_{i(t-1)})} \dots (2)$$

where, *C*=1 if censored, and other variable definitions are same as IPTW above. The numerator was the estimated probability of being uncensored ($C_t = 0$) for each time interval given baseline covariates (X_0), past exposure (E_{t-1}) and past being uncensored ($C_{t-1} = 0$) and the denominator was the estimated probability of being uncensored ($C_{t-1} = 0$) given baseline covariates (X_0), past exposure (E_{t-1}), past being uncensored ($C_{t-1} = 0$) and time-varying covariates (X_0), past exposure (E_{t-1}), past being uncensored ($C_{t-1} = 0$) and time-varying covariates (V_t). These weights were obtained by fitting a logistic regression model to obtain the probability of past being uncensored ($C_{t-1} = 0$) at each visit. The stabilized IPTW and IPCWs will be multiplied to obtain the final weight. The final weight was truncated at the 99th percentile. Summary statistics for the stabilized final weights are provided in table 5.5.

No Misclassification correction			Misclassification correction					
Weights	Mean (SD)	Median (IQR)	Range	Mean (SD)	Median (IQR)	Range		
Primary Analyses								
IFN era (2003-2011)								
Stabilized IPTW	01.01 (0.48)	0.89 (0.75-1.18)	0.12-3.43	1.00 (0.30)	0.98 (0.86-1.10)	0.12-5.63		
Final weight (Stabilized IPTW * Stabilized IPCW)	1.00 (0.49)	0.89 (0.74-1.19)	0.11-3.39	1.00 (0.34)	0.98 (0.83-1.11)	0.11-6.46		
DAA era (2013-2020)								
Stabilized IPTW	1.00 (0.31)	0.95 (0.82-1.14)	0.03-2.23	1.00 (0.21)	0.99 (0.90-1.08)	0.10-4.01		
Final weight (Stabilized IPTW * Stabilized IPCW)	0.99 (0.33)	0.94 (0.82-1.15)	0.03-2.34	0.99 (0.25)	0.98 (0.88-1.08)	0.08-6.75		
Secondary analysis - Restricted subset with CES-D-10 scores available (2012-2015)								
Stabilized IPTW	1.01 (0.35)	0.95 (0.81-1.12)	0.29-2.65	-	-	-		
Final weight (Stabilized IPTW * Stabilized IPCW)	1.01 (0.36)	0.95 (0.79-1.14)	0.11-2.70	-	-	-		

Table 5.5: Supplementary table - Summary of the inverse probability weights

Abbreviations: SD: Standard deviation; IQR: Interquartile range; IFN: Interferon; DAA: Direct acting antivirals; IPTW: Inverse probability treatment weight; IPCW: Inverse probability censoring weight; CES-D-10: Center for Epidemiologic Studies Depression Scale-10

References

1. Klein MB, Saeed S, Yang H, et al. Cohort profile: the Canadian HIV-hepatitis C coinfection cohort study. *International journal of epidemiology* 2010; 39: 1162-1169. 2009/09/30. DOI: 10.1093/ije/dyp297.

2. Cox J and Hamelin A. Prospective investigation of the relationship between food insecurity and health and behavioural outcomes in HIV-HCV co-infection: clues for prevention interventions (CTN 264). A food security & HIV-HCV sub-study of the Canadian Co-Infection Cohort (CTN 222). 2016.

3. Radloff LS. The CES-D Scale: A Self-Report Depression Scale for Research in the General Population. *Applied Psychological Measurement* 1977; 1: 385-401. DOI: 10.1177/014662167700100306.

4. Andresen EM, Malmgren JA, Carter WB, et al. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). *American journal of preventive medicine* 1994; 10: 77-84. 1994/03/01.

Zhang W, O'Brien N, Forrest JI, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. *PLOS one* 2012;
 e40793. 2012/07/26. DOI: 10.1371/journal.pone.0040793.

Breiman L. Random Forests. *Machine Learning* 2001; 45: 5-32. journal article.
 DOI: 10.1023/a:1010933404324.

7. Fawagreh K, Gaber MM and Elyan E. Random forests: from early developments to recent advancements AU - Fawagreh, Khaled. *Systems Science & Control Engineering* 2014; 2: 602-609. DOI: 10.1080/21642583.2014.956265.

8. Hastie T, Tibshirani R and Friedman JH. *The elements of statistical learning: data mining, inference, and prediction*. Second edition. ed. New York: Springer, 2009.

9. Mennitt D, Sherrill K and Fristrup K. A geospatial model of ambient sound pressure levels in the contiguous United States. *The Journal of the Acoustical Society of America* 2014; 135: 2746-2764. 2014/05/13. DOI: 10.1121/1.4870481.

10. Strobl C, Malley J and Tutz G. An introduction to recursive partitioning: rationale, application, and characteristics of classification and regression trees, bagging, and random forests. *Psychological methods* 2009; 14: 323-348. 2009/12/09. DOI: 10.1037/a0016973.

11. Malley JD, Kruppa J, Dasgupta A, et al. Probability machines: consistent probability estimation using nonparametric learning machines. *Methods Inf Med* 2012; 51: 74-81. 2011/09/14. DOI: 10.3414/ME00-01-0052.

12. Robins JM, Hernan MA and Brumback B. Marginal structural models and causal inference in epidemiology. *Epidemiology (Cambridge, Mass)* 2000; 11: 550-560. 2000/08/24.

13. Hernan MA, Brumback B and Robins JM. Marginal structural models to estimate the causal effect of zidovudine on the survival of HIV-positive men. *Epidemiology (Cambridge, Mass)* 2000; 11: 561-570. 2000/08/24.

6. CHAPTER 6: IMPACT OF HCV CURE ON THE PRESENCE OF DEPRESSIVE SYMPTOMS

6.1. Preface to Manuscript 3

In this analysis, I aimed to assess the impact that SVR may have on depressive symptoms over time in people co-infected with HIV-HCV. To assess these long-term impacts of HCV cure, we need to consider both the biological and psychosocial underlying mechanisms of depression in the co-infected population. Since the DAA regimens lead to HCV clearance, we can hypothesize that induction of depression by the biological pathway via immune activation may be alleviated. However, the co-infected population may continue to face challenges including discrimination, socioeconomic burdens, and/or substance use even after HCV cure. This could potentially have an impact on psychosocial and health outcomes like overdoses reducing the benefits of HCV cure. Thus, the broadening of HCV treatment to all people, irrespective of mental health, now provides us with the opportunity to assess the potential impact of HCV treatment on depressive symptoms over time.

A few studies have assessed depressive symptoms before treatment and at SVR ascertainment (83, 85-88). For example, in a study by Moez et. al., the BDI scores were found to be lower (reduced depression severity) at SVR-12 compared to baseline (111). In contrast, in a prospective follow-up study conducted by Khalil et. al., psychiatric assessment using three scales, BDI, symptom checklist 90-R and Structured Clinical Interview for DSM-IV (SCID-IV) was done at baseline and at SVR, and the all scores were shown to have increased post-treatment, with 32% developing moderate to severe

depression (112). All the above studies, however, were conducted in the HCV monoinfected population. Only one study to my knowledge was conducted among HIV-HCV co-infected people; it showed a decline in BDI scores from baseline to 1-8 weeks after end of DAA treatment (113). Additionally, none of the studies examined depression beyond SVR, and it could be possible that depressive symptoms may decrease transiently at the time of cure, but very little is known about post-SVR depressive symptoms trends and persistence in both HCV mono-infected and HIV-HCV co-infected populations.

To address this gap in knowledge, in this manuscript, I assessed the impact of SVR on depressive symptoms in the HIV-HCV co-infected population in the second-generation DAA era (2013-2020).

This manuscript is submitted for publication and is under review at Clinical Infectious Diseases (CID).

This analysis was presented as posters at the following conferences:

- 29th Conference on Retroviruses and Opportunistic Infections (CROI 2022), virtual
 February 2022
- 2. Canadian Liver Meeting (CLM) 2022, Hybrid, Ottawa, Canada May 2022

6.2. Manuscript 3: Impact of HCV cure on depressive symptoms in the HIV-HCV co-infected population in Canada

Gayatri Marathe^{1,2}, Erica EM Moodie¹, Marie-Josée Brouillette^{2,3}, Charlotte Lanièce Delaunay^{1,2}, Joseph Cox^{1,2}, Valérie Martel-Laferrière⁴, John Gill⁵, Curtis Cooper⁶, Neora Pick⁷, Marie-Louise Vachon⁸, Sharon Walmsley⁹, Marina B. Klein^{1,2,10}; Canadian Co-Infection Cohort

- 1. Department of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada
- 2. McGill University Health Center-Research Institute, Centre for Outcomes Research and Evaluation, Montreal, Quebec, Canada
- 3. Department of Psychiatry, McGill University, Montreal, Quebec, Canada
- 4. Département de microbiologie, infectiologie et immunologie, *Centre de Recherche du Centre Hospitalier de l'Université de Montréal, Montréal, Québec, Canada*
- 5. Department of Medicine, University of Calgary, Calgary, Alberta, Canada
- 6. Department of Medicine, University of Ottawa, Ottawa, Ontario, Canada
- 7. Oak Tree Clinic, BC Women's Hospital, Vancouver, British Columbia, Canada
- 8. Centre Hospitalier de l'Université Laval, Québec, Québec, Canada
- 9. Department of Medicine, University of Toronto, Ontario, Canada
- 10. CIHR Canadian HIV Trials Network (CTN), Vancouver, British Columbia, Canada

Key words (up to 5): HIV-HCV co-infection; Depressive symptoms; Direct Acting Antivirals; HCV cure; Sustained virologic response

Running title (Character limit: 40): Impact of SVR on depressive symptoms

Corresponding author:

Marina B. Klein Division of Infectious Diseases and Chronic Viral Illness Service McGill University Health Centre 1001 Decarie Boulevard, D02.4110 Montreal, Canada H4A 3J1 Phone. 1-514-843-2090; Fax: 1-514-843-2092 Email address: marina.klein@mcgill.ca

Alternate corresponding author

Gayatri J. Marathe School of Population and Global Health McGill University 2001 McGill College Avenue Montreal, Canada H3A 1G1 Email address: gayatri.marathe@mail.mcgill.ca

Key points (Word limit: 40 words)

Depressive symptoms are common among HIV-HCV co-infected people. This study suggests that, in the direct acting antivirals era, depressive symptoms declined over time

following HCV cure. Thus, HCV cure leads to benefits beyond reduced risk of liver disease.

Abstract

Background

Depression is common in people living with HIV-HCV, with biological and psychosocial mechanisms at play. Direct acting antivirals (DAA) result in high rates of sustained virologic response (SVR), with minimal side-effects. We assessed the impact of SVR on presence of depressive symptoms in the HIV-HCV co-infected population in Canada during the second-generation DAA era (2013-2020).

Methods

We used data from the Canadian Co-infection Cohort (CCC), a multicentre prospective cohort of people with a HIV and HCV co-infection, and its associated sub-study on food security. Since depression screening was performed only in the sub-study, we predicted Center for Epidemiologic Studies Depression Scale-10 classes in the CCC using a random forest classifier and corrected for misclassification. We included participants who achieved SVR and fit a segmented modified Poisson model using an interrupted time series design, adjusting for time-varying confounders.

Results

We included 470 participants; 58% had predicted depressive symptoms at baseline. The median follow-up was 2.4 years (IQR: 1.0-4.5.) pre-SVR and 1.4 years (IQR: 0.6-2.5) post-SVR. The pre-SVR trend suggested depressive symptoms changed little over time, with no immediate level change at SVR. However, post-SVR trends showed a reduction

of 5% per year (risk ratio: 0.95 (95%CI: 0.94-0.96)) in the prevalence of depressive symptoms.

Conclusions

In the DAA era, predicted depressive symptoms declined over time following SVR. These improvements reflect possible changes in biological pathways and/or better general health. If such improvements in depression symptoms are durable, this provides an additional reason for treatment and early cure of HCV.

Background

Both Hepatitis C virus (HCV) and HIV infections are associated with neuropsychiatric manifestations, mainly depression [1, 2]. Among people living with either HIV or chronic HCV, prevalence of depression ranges from 20-30% [1, 3, 4]. Depression is reported to be even higher in the HIV-HCV co-infected population [5]. Depression mechanisms related to HIV and HCV are both biological and psychosocial. HIV and HCV affect the central nervous system directly, which causes immune activation leading to depression [1, 6]. In addition, pro-inflammatory cytokines like TNF- α and IL-1 and altered neurotransmitter action like dopamine and serotonin play roles in inducing depression [1, 6, 7]. There are also many known psychosocial pathways to depression including social stresses caused by stigma, discrimination, and lack of social and financial support [1, 2]. Among HCV-HIV co-infected persons, ongoing substance use is a common additional risk factor, and may also be affected by presence of depressive symptoms [8].

Depression was a well described major side effects of earlier interferon (IFN)-ribavirin based HCV antiviral treatments, with some studies showing more than 20% of those treated developed depression [9]. This led to those with current or past psychiatric illness often not being prescribed IFN therapy [10]. However, after 2013, second-generation IFN-free direct acting antiviral (DAA) regimens now result in >95% rates of sustained virologic response (SVR), even among HIV-HCV coinfected persons [11, 12]. Importantly, there is no evidence of any significant psychiatric side effects associated with DAA treatment [13-15]. HCV treatment guidelines have thus been updated and depression is no longer a contraindication for treatment [16].

These changes in prescribing practices provide us with the opportunity to assess the potential impact of HCV treatment on depressive symptoms over time. We may expect lower depressive symptoms post-cure via biological pathways due to HCV viral clearance. However, co-infected populations continue to face challenges including discrimination, socioeconomic burden, and/or substance use, increasing risk of overdoses and poor mental health, mitigating the benefits of HCV cure. Thus, it is important to examine longitudinally whether HCV cure leads to change in the level of depressive symptoms and moreover whether such a change persists over time. This will provide evidence for healthcare providers to appropriately monitor and manage depression. Evidence of possible improvement in mental health after cure could encourage individuals hesitant to start treatment to do so. Thus, in this study we evaluated the impact of SVR on presence of depressive symptoms in the HIV-HCV co-infected population in Canada during the second-generation DAA era (2013-2020).

Methods

Study population

We used data from the Canadian Co-Infection Cohort (CCC), an open multicentre prospective cohort study, established in 2003 and described in detail elsewhere [17]. Briefly, the CCC recruits from 18 urban and semi-urban HIV centres across 6 Canadian provinces (Quebec, British Columbia, Alberta, Ontario, Nova Scotia, and Saskatchewan). Eligibility criteria include ≥16 years of age, documented HIV infection, and evidence of HCV infection (HCV RNA positivity and/or HCV seropositivity). As of July 2020, the study had recruited 2018 participants. Participants are followed longitudinally, with visits every

six months. Sociodemographic and behavioural data are collected by a standardized selfadministered questionnaire at each visit. Clinical data including HIV and HCV treatment dates, medications, co-morbidities, and psychiatric diagnoses are collected via medical chart reviews and HIV and HCV related blood tests performed.

We also used data from a sub-study conducted within the CCC, the Food Security and HIV-HCV co-infection study (FS sub-study), to predict the presence of depressive symptoms in the parent CCC. Participants for the FS sub-study were recruited from the CCC (n=725) with a maximum of 5 visits integrated into CCC visits from 2012-2015. In the sub-study, described elsewhere [18], depression screening was performed using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) that assesses presence and severity of depressive symptoms in the past week [19]. A score \geq 10 is widely considered to represent the presence of depressive symptoms indicative of being at risk for depression; this cut-off has been validated in HIV populations in Canada [20].

Measurement

Exposure

The exposure of interest was successful HCV treatment or cure in individuals treated with DAA regimens. Successful treatment or sustained virological response (SVR) was defined as an undetectable viral load (HCV RNA) 12 weeks after the end of treatment. We included participants who were HCV RNA positive, were treated and then achieved SVR during the second-generation (IFN free) DAA era. The second-generation DAA era was defined from when the first second-generation DAA, Simeprevir, was approved for

use by Health Canada November 25, 2013, and continued until end of study period, July 15, 2020.

Outcome

The outcome of interest was presence of depressive symptoms indicative of being at risk for major depression, hereafter referred to as depressive symptoms. CCC participants are not screened for depression as part of usual study procedures (baseline or followup), however depression screening was performed in the FS sub-study. Since the FS sub-study was conducted between 2012 and 2015, we only had such measurements for about 1.5 years in the second-generation DAA era, thus, insufficient data with which to conduct an analysis using measured depressive symptoms. Thus, to obtain a measure of depressive symptoms in the full CCC, we developed a random forest (RF) classifier using the CES-D-10 to classify presence/absence of depressive symptoms derived from the FS sub-study as the outcome (target of prediction), and sociodemographic, behavioural, and clinical characteristics from the parent CCC as predictors [21]. We used the CES-D-10 score cut-off of 10, such that "CES-D-10 class=1" corresponds to a score ≥10 for presence and "CES-D-10 class=0" corresponds to a score < 10 for absence of depressive symptoms indicative of being at risk for depression. The details of the RF classifier development are in appendix A and table 6.4. Using this RF classifier, the CES-D-10 classes were predicted for each CCC visit included in this analysis. We addressed outcome misclassification for the predicted depressive symptoms using the predictive value-based record-level correction method [22]. In this method, we applied the positive predictive value (PPV) and negative predictive value (NPV) estimated for the RF

algorithm; PPV: 0.74 (95% CI: 0.68-0.80) and NPV: 0.76 (95% CI: 0.69-0.82). The procedure included simulation of corrected outcome at each visit by repeated Bernoulli trials with probability equal to PPV for those classified as CES-D-10 class=1 and 1-NPV for those classified as CES-D-10 class=0 [22].

Confounders

We considered time-varying confounders, which were selected a priori [23-25]. These confounders were measured at each biannual visit and included advanced fibrosis/cirrhosis, HIV viral load, CD4 cell count, current injection drug use, current alcohol use, recent incarceration, and antidepressant use. We dichotomized three confounders: advanced fibrosis/cirrhosis (AST to Platelet Ratio Index (APRI) >=1.5 and/or end stage liver disease diagnosis), HIV viral load (at 50 copies/ml), CD4 cell count (at 250 cells/µl). Though using continuous measures may have improved precision, these dichotomizations were chosen to reflect clinical cut-offs for assessment of fibrosis stage and HIV control. In addition, we opted to adjust for HIV viral load directly, rather than antiretroviral therapy status as these two factors are correlated and viral load would be more relevant regarding biological mechanisms underlying HIV and depression [1]. At least one confounder value was missing in 38% of the included visits. We assessed if this missingness was informed by other covariates by using logistic regressions with the missing data indicator as the outcome for each confounder and found missingness to be informed by other covariates in two confounders, recent incarceration, and alcohol use. We used multiple imputation by chained equations (MICE) to address this missing

data in the confounders [26]. We created five imputed datasets using logistic regression for these binary confounder variables.

Statistical analysis

Primary analysis

We used a segmented regression model with interrupted time series (ITS) design to evaluate the impact of SVR on depressive symptoms. In an ITS, a time series of a particular outcome of interest is used to establish an underlying trend, which is 'interrupted' by an exposure at a known point in time, with a clear differentiation between the pre-exposure and post-exposure periods [27]. In this analysis, the extrapolation of the pre-exposure outcome trend acts as a counterfactual for the post-exposure trend for each individual. It is assumed that, since the same individual is observed before and after the exposure, this design accounts for known and unknown confounders that do not vary with time [27, 28]. The pre-exposure period included time from cohort entry when participants were HCV RNA positive to treatment initiation in the second-generation DAA era. The post-exposure period included time after the ascertainment of SVR for each individual. We did not include the time between DAA initiation and SVR ascertainment in the analysis. Based on subject matter knowledge, we hypothesized that depressive symptoms may have an immediate decrease at SVR as well as a decrease over time. The causal diagram can be seen in appendix B-figure 6.3. Subgroup analyses were performed to explore possible difference by sex (male, female), race (white, Indigenous), employment status (employed, not employed), and baseline liver disease (no liver disease, with liver disease).

We used Generalized Estimating Equations (GEEs) with robust standard errors, which account for correlation between repeated measurements on the same participant over time. GEEs are a population-level approach and provides population-averaged estimates of the parameters (as opposed to the individual-level analysis provided by mixed effect models) [29]. We used an exchangeable working correlation structure in these models, which assumes positive correlation between repeated measurements over time for an individual. The segmented modified Poisson model, which involves using a robust variance estimation, was defined as below [27]. We developed these models with and without outcome misclassification correction. The model can be seen in appendix C.

Sensitivity analyses

We conducted planned sensitivity analyses to assess robustness of the results, specifically due to two possible methodological challenges for ITS - lead time bias and non-linear effect. Lead time bias is possible in this analysis as depressive symptoms may change in anticipation of the exposure, SVR [28]. To check for lead time bias, we moved the time axis to set the exposure one year before SVR ascertainment. To address the possibility that the effect may not be linear on the log scale, we developed adjusted models with polynomials (squared and cubic transformations of time) and also with restricted cubic spline with 5 knots - see further details in appendix D [30]. We then used the quasi-likelihood under the independence model criterion (QIC) for model selection among the linear and the non-linear models [31]. Additionally, we conducted a sensitivity analysis exploring depressive symptom trends for those who did not respond to DAAs, by

comparing the trends before and after the date of no-response ascertainment. All primary and sensitivity analyses were performed using Stata v.17 [32].

Results

Participant characteristics

The flowchart for participants in the final analytical sample is shown (figure 6.1). We included 470 participants who achieved SVR in the DAA era. Baseline characteristics are shown in table 6.1. Participants were vulnerable and could face potential barriers to HCV and mental health care. They were predominantly male (68%) and unemployed (68%); 23% were Indigenous, 50% were on welfare as their primary revenue source, 73% had no post-secondary education, and 34% were current injection drug users. At baseline, 58% of the cohort had predicted depressive symptoms indicative of a risk of depression.

Primary analyses

The results of the primary analyses are shown in table 6.2 and illustrated in figure 6.2 A. The median follow-up was 2.4 years (Interquartile range (IQR): 1.0-4.5) pre-SVR and 1.4 years (IQR: 0.6-2.5) post-SVR. After correcting for outcome misclassification, the pre-treatment trends show an adjusted risk ratio (aRR) of 1.01 (95% CI: 1.01-1.02), which indicates little change in the annual rate of predicted depressive symptoms over time prior to treatment. The model does not show any immediate level change at SVR (aRR) of 1.01 (95% CI: 0.97-1.04). However, the post-SVR trends shows a decrease in depressive symptoms over time, of 5% per year (aRR of 0.95 (95% CI: 0.94-0.96)). There were no major differences noted between the various subgroups (sex, race, employment status,

and liver disease) in the pre-treatment initiation and changes at SVR. There was some difference noted in the post-SVR downward trend by sex and race, however sample sizes were limited, precluding definitive conclusions (see appendix E-table 6.6).

Sensitivity analyses

The results of the sensitivity analysis used to assess possible lead time bias are shown in table 6.3 and illustrated in figure 6.2 B. The trends are similar for pre-treatment period as the primary analysis. There is an increase in the immediate level of depressive symptoms prevalence one-year pre-SVR (aRR: 1.06 (1.02-1.10)), showing no evidence of lead time bias. The second sensitivity analysis did not support non-linearity of the effect on the log scale: the linear model was selected based on the lowest values of the QIC statistic. These results of the non-linearity sensitivity analysis are shown in table 6.5 in appendix D. The results for the sensitivity analysis with DAA non-responders are shown in table 6.7 and figure 6.4 in appendix F. In DAA non-responders, the pre-treatment probability trend was stable over time, with no evidence of immediate change at date of no-response ascertainment. However, the probability trend post-no-response indicates a gradual increase in depressive symptoms over time.

Discussion

We measured using segmented regression models the impact of HCV cure on predicted depressive symptoms. While depressive symptoms changed little over time in the lead up to DAA treatment, we observed a gradual decline in prevalence of depressive symptoms over time post-SVR among patients co-infected with HIV. There was no

evidence of immediate change at SVR. The improvement after cure may reflect changes in biological pathways leading to HCV-related depression due to viral clearance and/or improved general physical health.

The use of DAAs has increased and improved HCV treatment among people with a history of depression or with current depressive symptoms. Several studies have assessed health-related quality of life post-SVR with DAAs and have shown a modest improvement after HCV cure [33, 34]. This is in line with our observation of decline of depressive symptoms over time, which are strongly correlated with health-related quality of life [35].

Several studies have compared depressive symptoms at baseline and at SVR-12. In a study by Moez et. al., Beck depression inventory (BDI) scores were found to be lower (reduced depression severity) at SVR-12 compared to baseline [36]. Similar results were observed in a few other studies [14, 15, 37]. In contrast, in a prospective study with psychiatric assessments at baseline and at SVR-12, scores were shown to have increased post-treatment, with 32% developing moderate to severe depression [38]. The authors suggested an explanation for this increase might include biological mechanisms related to increased levels of IFN, the higher percentage of women in the cohort, continued stigma, other co-morbid conditions, and persisting unemployment [38, 39]. Similar results were seen in other studies [40, 41]. All the above studies, however, were conducted in HCV mono-infected populations. Only one study to our knowledge was among HIV-HCV co-infected people which showed a decline in BDI scores from baseline to 1-8 weeks after end of DAA treatment [42]. There are several possible methodological reasons for these conflicting results, like different sample sizes (e.g., n=150 (Moez. et.

al.) vs. n=47 (Khalil et. al.), different depression scales, and varying measurement time points (between 4 and 12 weeks of treatment; not all at SVR, e.g., Egmond et. al.) [36, 38, 41]. Additionally, none of the studies examined depression in the time frame beyond SVR and thus very little is known about post-SVR depressive symptoms trends and persistence in both HCV mono-infected and HIV-HCV co-infected populations.

One major strength of our study is we used longitudinal data collected in numerous, diverse patients. Using a quasi-experimental design, ITS, we were able to obtain robust marginal effect estimates of the impact of SVR on depressive symptoms in the co-infected population. We believe our estimates are generalizable to HIV-HCV co-infected patients engaged in care in Canada, as CCC participants are recruited from primary and tertiary care clinics in urban and semi-urban areas across six provinces in Canada. We also conducted multiple sensitivity analyses and adjusted for time-varying confounders.

Our study, however, does have some limitations. The depressive symptoms were predicted via a RF algorithm, and not measured directly in the cohort. Misclassification was therefore expected, and we corrected for it. We predicted the depressive symptoms based on a screening questionnaire, CES-D-10, and not a major depression diagnosis. Thus, this study does not provide an effect estimate for depression but rather for depressive symptoms that are indicative of a depression risk. Further, we predicted the CES-D-10 classes based on the validated cut-off of 10 and not a CES-D-10 continuous score. The continuous score prediction algorithm could only explain a small portion of the outcome variability. This could have been because the FS sub-study sample was relatively small and did not capture the full range of the continuous scale. We used an

ITS because an appropriate control group was difficult to find. Those not treated may be inherently different from those treated, and in the DAA era, very few of those treated fail to achieve SVR. Finally, the crucial assumption of the ITS design that the extrapolated pre-exposure trend is considered the counterfactual trend, makes it vulnerable to unmeasured time-varying confounding, which we tackled by adding known time-varying confounders; however, some residual confounding could still be possible. Finally, the median post-SVR follow-up was 1.4 years, so the durability of the observed effect is yet to be explored. Persisting psychosocial and economic burdens post-cure such as stigma and discrimination in social, professional as well as healthcare settings could still lead to shame, suffering and lack of disease-related education and recurrence of depressive symptoms over the long-term [43].

In conclusion, following SVR, there appears to be a continuous decline in the presence of depressive symptoms in highly vulnerable patients co-infected with HCV and HIV. This finding suggests that the health benefits of curing HCV extend beyond improving liver disease and provides additional rationale for treating HCV in all chronically infected persons.

Baseline characteristics	Participants (n = 470) n (%) or median (IQR)	
Predicted presence of depressive symptoms (CES-D-10	070 (500/)	
score \geq 10)	272 (58%)	
Age (years)	47 (41-52)	
Gender - Male	321 (68%)	
Self-reported race/ethnicity		
White	323 (69%)	
Indigenous (First Nations, Inuit, and Metis)	107 (23%)	
Asian	14 (3%)	
Black	17 (4%)	
Hispanic/Latinx	7 (2%)	
Living situation - Homeless	48 (10%)	
Education - High school educated and less	344 (73%)	
Employment - Not employed	317 (68%)	
Monthly income - <= \$1500 CAD	356 (76%)	
Revenue source - Welfare	233 (50%)	
Sexual orientation - Heterosexual	319 (68%)	
Immigrant to Canada	42 (9%)	
Marital status - Single	329 (70%)	
Previous IFN-based HCV treatment	69 (15%)	
Injection drug use in the past 6 months	159 (34%)	
Alcohol use in the past 6 months	243 (52%)	
Incarceration in the past 6 months	31 (7%)	
Liver disease - APRI score >= 1.5 and/or liver disease	00 (10%)	
diagnosis	90 (1976)	
Hepatitis B infection	17 (4%)	
HIV viral load - > 50 copies/ml	121 (26%)	
CD4 count - <= 250 cells/uL	107 (23%)	
Antidepressant prescribed in the past 6 months	40 (9%)	

Table 6.1: Baseline characteristics of the included participants (n = 470)

Abbreviations: IQR: Interquartile range; CES-D-10: Center for Epidemiologic Studies Depression Scale-10; CAD: Canadian dollars; IFN: Interferon; HCV: Hepatitis C virus; APRI: AST to Platelet Ratio Index; HIV: Human Immunodeficiency Virus; CD4: Cluster of differentiation 4 receptor
Table 6.2: Impact of SVR on depressive symptoms in the HIV-HCV co-infected population

Sr.		Risk ratios (95% CI)		
No.	Models	Pre-treatment trends	Level change	Post-SVR
		per year	at SVR	trends per year
	Unadjusted models			
Α	No misclassification correction	1.01 (0.99-1.04)	1.01 (0.93-1.09)	0.92 (0.88-0.96)
В	Misclassification correction	1.01 (1.00-1.01)	1.01 (0.97-1.04)	0.95 (0.94-0.96)
II	Adjusted models *			
Α	No misclassification correction	1.02 (0.99-1.04)	1.01 (0.94-1.10)	0.92 (0.88-0.96)
В	Misclassification correction	1.01 (1.01-1.02)	1.01 (0.97-1.04)	0.95 (0.94-0.96)

- primary analysis models with and without outcome misclassification correction (n=470)

* Adjusted for time-varying confounders: Advanced fibrosis/cirrhosis (AST to Platelet Ratio Index (APRI) >=1.5 and/or end stage liver disease diagnosis), detectable HIV viral load (>50 copies/ml), low CD4 cell count (≤250 cells/μl), current injection drug use, current alcohol use, incarceration in the past 6 months and antidepressant use **Abbreviations:** SVR: Sustained virologic response; HIV: Human immunodeficiency virus; HCV: Hepatitis C virus; CI: Confidence interval

Table 6.3: Sensitivity analysis to assess possible lead time bias - models with and without

outcome misclassification correction (n=332)

Sr.		Risk ratios (95% CI)		
No.	Models	Pre-treatment trends	Level change	Post-SVR
		per year	at SVR	trends per year
I	Unadjusted models			
Α	No misclassification correction	1.01 (0.98-1.03)	1.10 (0.99-1.22)	0.93 (0.89-0.98)
В	Misclassification correction	1.00 (0.99-1.01)	1.10 (1.01-1.10)	0.96 (0.95-0.98)
II	Adjusted models *			
Α	No misclassification correction	1.01 (0.98-1.04)	1.10 (0.99-1.23)	0.93 (0.90-0.97)
В	Misclassification correction	1.01 (0.99-1.01)	1.06 (1.02-1.10)	0.96 (0.95-0.98)

* Adjusted for time-varying confounders: Advanced fibrosis/cirrhosis (AST to Platelet Ratio Index (APRI) >=1.5 and/or end stage liver disease diagnosis), detectable HIV viral load (>50 copies/ml), low CD4 cell count (\leq 250 cells/µl), current injection drug use, current alcohol use, incarceration in the past 6 months and antidepressant use **Abbreviations:** CI: Confidence interval; SVR: Sustained virologic response

Figure 6.1: Flowchart of participants included in the analytical sample



Figure description: The Canadian Co-infection Cohort (CCC) had recruited 2018 HIV-HCV co-infected participants (HCV RNA positive/HCV seropositive) until July 2020. In our analysis, we included 503 participants who were treated with IFN-free second-generation Direct Acting Antivirals (DAA) regimens after November 25, 2013, when the first secondgeneration DAA, Simeprevir, was approved for use by Health Canada. Of the participants who were treated, we excluded those who did not achieve sustained virologic response (SVR) (n = 32) and did not have a treatment response date (n = 1). Thus, in the final analytical sample we included a total of 470 participants who had achieved SVR.

Figure 6.2: Impact of sustained virologic response (SVR) on depressive symptoms in the HIV-HCV co-infected population



A: Primary analysis with misclassification correction



158

Figure Description: A. Results of the primary analysis model with outcome misclassification correction - The graph shows that pre-treatment the probability trend for presence of depressive symptoms was stable over time. There was no evidence of immediate change at SVR, however the probability trends post-SVR indicate a gradual decline in depressive symptoms over time. B. Results of the sensitivity analysis model to assess lead time bias with outcome misclassification correction - In this model, we lagged SVR by one year to assess possibility of lead time bias. The graph shows a stable pre-treatment trend like the primary analysis. The increase in the immediate level of depressive symptoms prevalence one-year pre-SVR, provides evidence for no lead time bias in this analysis, meaning depressive symptoms did not seem to improve in anticipation of the cure.

Acknowledgements

We would like to acknowledge the participants of the Canadian Co-Infection Cohort (CTN222), the study coordinators and nurses for their assistance with study coordination, participant recruitment, and care, and the Canadian Co-Infection Cohort (CTN222) co-investigators - Drs. Lisa Barrett, Jeff Cohen, Brian Conway, Curtis Cooper, Pierre Côté, Joseph Cox, M. John Gill, Shariq Haider, David Haase, Mark Hull, Valérie Martel-Laferrière, Julio Montaner, Erica E. M. Moodie, Neora Pick, Danielle Rouleau, Aida Sadr, Steve Sanche, Roger Sandre, Mark Tyndall, Marie-Louise Vachon, Sharon Walmsley and Alexander Wong.

Funding: This work was supported by Fonds de recherche du Québec – Santé; Réseau sida/maladies infectieuses, the Canadian Institute for Health Research (CIHR; FDN-143270); and the CIHR Canadian HIV Trials Network (CTN222 & CTN264). GM and CLD are supported by PhD trainee fellowships from the Canadian Network on Hepatitis C. The Canadian Network on Hepatitis C is funded by a joint initiative of the Canadian Institutes of Health Research (NHC-142832) and the Public Health Agency of Canada. EEMM is supported by a chercheur de mérite award from the Fonds de recherche du Québec-Santé and a Canada Research Chair (Tier 1). CLD received a doctoral training award from the Fonds de recherche du Québec - Santé. VML is supported by Clinical Research Scholars–Junior 1 from the Fonds de recherche du Québec-Santé and a Research Chair. The funders had no role in the production of this manuscript.

Conflicts of interest: JC received grants and consulting fees from ViiV Healthcare, Merck, and Gilead and personal fees from Bristol-Myers Squibb. JG has served as ad hoc member on National HIV advisory boards to ViiV healthcare, Gilead, and Merck. CC has received personal fees for being a member of the national advisory boards of Gilead, Merck, Janssen, and Bristol-Myers Squibb. SW received grants, consulting fees, lecture fees, nonfinancial support, and fees for the development of educational presentations from Merck, ViiV Healthcare, GlaxoSmithKline, Pfizer, Gilead, AbbVie, Bristol-Myers Squibb, and Janssen. NP reports honoraria from Gilead and ViiV Healthcare. MBK reports grants for investigator-initiated studies from ViiV Healthcare, AbbVie, Merck, and Gilead; and consulting fees from ViiV Healthcare, Merck, AbbVie, and Gilead. GM, EEM, MJB, CLD, VML and MLV have no conflicts of interest to disclose.

Ethics approval: This study was approved by the Research Ethics Board of the McGill University Health Centre (2021-6985). The CCC and the FS Sub-Study were approved by the Research Ethics Board of the McGill University Health Centre (2006-1875, BMB-06-006t, 2013-994) and the research ethics boards of participating institutions. The study was conducted according to the Declaration of Helsinki.

Consent to participate: Informed consent was obtained from all individual participants included in the study.

Meetings to be presented at: Conference on Retroviruses and Opportunistic Infections (CROI) 2022 - Virtual - February 12-16, 2022 - Poster presentation; Canadian Liver Meeting (CLM) 2022, Ottawa, Canada - May 13-15, 2022 - Poster presentation.

References

1. Nanni MG, Caruso R, Mitchell AJ, Meggiolaro E, Grassi L. Depression in HIV infected patients: a review. Current psychiatry reports **2015**; 17(1): 530.

2. Yeoh SW, Holmes ACN, Saling MM, Everall IP, Nicoll AJ. Depression, fatigue and neurocognitive deficits in chronic hepatitis C. Hepatology international **2018**.

3. Younossi Z, Park H, Henry L, Adeyemi A, Stepanova M. Extrahepatic Manifestations of Hepatitis C: A Meta-analysis of Prevalence, Quality of Life, and Economic Burden. Gastroenterology **2016**; 150(7): 1599-608.

4. Ciesla JA, Roberts JE. Meta-analysis of the relationship between HIV infection and risk for depressive disorders. American Journal of Psychiatry **2001**; 158(5): 725-30.

5. Fialho R, Pereira M, Rusted J, Whale R. Depression in HIV and HCV co-infected patients: a systematic review and meta-analysis. Psychology, health & medicine **2017**; 22(9): 1089-104.

6. Yarlott L, Heald E, Forton D. Hepatitis C virus infection, and neurological and psychiatric disorders – A review. Journal of Advanced Research **2017**; 8(2): 139-48.

7. Liu CS, Adibfar A, Herrmann N, Gallagher D, Lanctôt KL. Evidence for Inflammation-Associated Depression. Curr Top Behav Neurosci **2017**; 31: 3-30.

 Kalichman SC, Washington C, Kegler C, et al. Continued Substance Use Among People Living With HIV-Hepatitis-C Co-Infection and Receiving Antiretroviral Therapy. Subst Use Misuse **2015**; 50(12): 1536-43.

9. Loftis JM, Hauser P. The phenomenology and treatment of interferon-induced depression. Journal of Affective Disorders **2004**; 82(2): 175-90.

10. Fleming CA, Tumilty S, Murray JE, Nunes D. Challenges in the Treatment of Patients Coinfected with HIV and Hepatitis C Virus: Need for Team Care. Clinical Infectious Diseases **2005**; 40(Supplement_5): S349-S54.

11. Burstow NJ, Mohamed Z, Gomaa AI, et al. Hepatitis C treatment: where are we now? International journal of general medicine **2017**; 10: 39-52.

12. Wyles DL, Sulkowski MS, Dieterich D. Management of Hepatitis C/HIV Coinfection in the Era of Highly Effective Hepatitis C Virus Direct-Acting Antiviral Therapy. Clinical Infectious Diseases **2016**; 63 Suppl 1: S3-s11.

13. Sackey B, Shults JG, Moore TA, Rogers R, Mehvar M, King JG. Evaluating psychiatric outcomes associated with direct-acting antiviral treatment in veterans with hepatitis C infection. The mental health clinician **2018**; 8(3): 116-21.

14. Gallach M, Vergara M, da Costa JP, et al. Impact of treatment with direct-acting antivirals on anxiety and depression in chronic hepatitis C. PLoS One **2018**; 13(12): e0208112.

15. Sundberg I, Lannergård A, Ramklint M, Cunningham JL. Direct-acting antiviral treatment in real world patients with hepatitis C not associated with psychiatric side effects: a prospective observational study. BMC Psychiatry **2018**; 18(1): 157.

16. Sanjeev Sockalingam, M.D. ,, Kathleen Sheehan, M.D., D.Phil. ,, Jordan J. Feld, M.D., M.P.H. ,, Hemant Shah, M.D., M.Sc.C.H. Psychiatric Care During Hepatitis C Treatment: The Changing Role of Psychiatrists in the Era of Direct-Acting Antivirals. American Journal of Psychiatry **2015**; 172(6): 512-6.

17. Klein MB, Saeed S, Yang H, et al. Cohort profile: the Canadian HIV-hepatitis C coinfection cohort study. International journal of epidemiology **2010**; 39(5): 1162-9.

18. Cox J, Hamelin AM, McLinden T, et al. Food Insecurity in HIV-Hepatitis C Virus Co-infected Individuals in Canada: The Importance of Co-morbidities. AIDS and behavior **2017**; 21(3): 792-802.

 Andresen EM, Malmgren JA, Carter WB, Patrick DL. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). American journal of preventive medicine **1994**; 10(2): 77-84.
 Zhang W, O'Brien N, Forrest JI, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. PLOS ONE **2012**; 7(7): e40793.

21. Breiman L. Random Forests. Machine Learning **2001**; 45(1): 5-32.

22. Banack HR, Stokes A, Fox MP, et al. Stratified Probabilistic Bias Analysis for Body Mass Index-related Exposure Misclassification in Postmenopausal Women. Epidemiology **2018**; 29(5): 604-13.

23. Aibibula W, Cox J, Hamelin A-M, et al. Association between depressive symptoms, CD4 count and HIV viral suppression among HIV-HCV co-infected people. AIDS Care **2018**; 30(5): 643-9.

24. Conner KR, Pinquart M, Duberstein PR. Meta-analysis of depression and substance use and impairment among intravenous drug users (IDUs). Addiction **2008**; 103(4): 524-34.

25. Huang X, Liu X, Yu Y. Depression and Chronic Liver Diseases: Are There Shared Underlying Mechanisms? Front Mol Neurosci **2017**; 10: 134.

26. Raghunathan TE, Lepkowski JM, Van Hoewyk J, Solenberger P. A Multivariate Technique for Multiply Imputing Missing Values Using a Sequence of Regression Models. Survey Methodology **2001**; 27(Part 1): 85-96.

27. Bernal JL, Cummins S, Gasparrini A. Interrupted time series regression for the evaluation of public health interventions: a tutorial. International journal of epidemiology **2017**; 46(1): 348-55.

28. Saeed S, Moodie EEM, Strumpf EC, Klein MB. Segmented generalized mixed effect models to evaluate health outcomes. International journal of public health **2018**; 63(4): 547-51.

29. Wang M. Generalized Estimating Equations in Longitudinal Data Analysis: A Review and Recent Developments. Advances in Statistics **2014**; 2014: 303728.

30. Gauthier J, Wu QV, Gooley TA. Cubic splines to model relationships between continuous variables and outcomes: a guide for clinicians. Bone Marrow Transplantation **2020**; 55(4): 675-80.

31. Cui J. QIC program and model selection in GEE analyses. Stata Journal 2007;7(2): 209-20.

32. StataCorp. Stata Statistical Software: Release 17. College Station, TX: StataCorp LLC, **2021**.

33. Saeed S, Moodie EEM, Strumpf E, et al. Real-world impact of direct acting antiviral therapy on health-related quality of life in HIV/Hepatitis C co-infected individuals. J Viral Hepat **2018**; 25(12): 1507-14.

34. Younossi ZM, Stepanova M, Afdhal N, et al. Improvement of health-related quality of life and work productivity in chronic hepatitis C patients with early and advanced fibrosis treated with ledipasvir and sofosbuvir. Journal of Hepatology **2015**; 63(2): 337-45.

35. Gaynes BN, Burns BJ, Tweed DL, Erickson P. Depression and health-related quality of life. J Nerv Ment Dis **2002**; 190(12): 799-806.

36. Abdel Moez AT, El Hawary YA, Al Balakosy AM. Can successful treatment by direct-acting antivirals improve depression in chronic HCV patients? European Journal of Gastroenterology & Hepatology **2021**; 33(5): 727-30.

37. Durcan E, Hatemi I, Sonsuz A, Canbakan B, Ozdemir S, Tuncer M. The effect of direct antiviral treatment on the depression, anxiety, fatigue and quality-of-life in chronic hepatitis C patients. European Journal of Gastroenterology & Hepatology **2020**; 32(2): 246-50.

38. Khalil MA, Shousha HI, EI-Nahaas SM, Negm MI, Kamal K, Madbouly NM. Depression in patients with chronic hepatitis-C treated with direct-acting antivirals: A real-world prospective observational study. Journal of Affective Disorders **2021**; 282: 126-32.

39. Wedemeyer H, Khera T, Strunz B, Björkström NK. Reversal of Immunity After Clearance of Chronic HCV Infection-All Reset? Front Immunol **2020**; 11: 571166.

40. Miarons M, Sánchez-Ulayar A, Sempere G, Marín S, Castellví JM. New directacting antivirals for hepatitis C treatment and neuropsychiatric symptoms in psychiatric risk groups. European Journal of Hospital Pharmacy Science and Practice **2019**; 26(3): 135-9.

41. Egmond E, Mariño Z, Navines R, et al. Incidence of depression in patients with hepatitis C treated with direct-acting antivirals. Braz J Psychiatry **2020**; 42(1): 72-6.

42. Lundgren L, Kattakuzhy S, Price A, et al. Abstract number 695: Mental Health Impact of HCV Treatment in HIV/HCV Patients: DAA vs IFN-Based Therapy. In: Conference on Retroviruses and Opportunistic Infections (CROI). Seattle, Washington, 2015.

43. Cho H-J, Park E. Illness Experience of Patients with Chronic Hepatitis C Participating in Clinical Trials. Osong Public Health and Research Perspectives **2016**; 7(6): 394-9.

6.3. Manuscript 3: Appendix

Appendix A: Random Forest algorithm for depressive symptoms prediction

- Appendix B: Causal diagram
- Appendix C: Segmented modified Poisson model
- Appendix D: Sensitivity analysis non-linearity
- **Appendix E:** Subgroup analyses

Appendix F: DAA non-responders

Appendix A: Random Forest algorithm for depressive symptoms prediction

Depression screening was not performed in the Canadian Co-infection Cohort (CCC) and hence a measure of depressive symptoms was not available. Thus, in previous work, we developed an algorithm to predict the presence of depressive symptoms indicative of risk of depression using data from a sub-study within the CCC, the Food Security and HIV-HCV co-infection study (FS sub-study) [1, 2]. In the FS sub-study, depression screening was conducted using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) [3, 4]. The CES-D-10 is a 10-item Likert scale questionnaire that assesses presence and severity of depressive symptoms in the past week, with a total score range of 0-30. We dichotomized the score at 10 to create the CES-D-10 classes (0/1), as a score \geq 10 is widely considered to be indicative of being at risk for major depression with clinically relevant depressive symptoms [4]. Both the scale and the dichotomization at 10 have been validated in HIV populations in Canada [5].

We used the supervised machine learning technique of Random Forests (RF), an ensemble learning approach which uses bootstrap aggregation of multiple decision trees, combining predictions from these many trees [6-10]. We developed probability machines to estimate the CES-D-10 class probabilities at each visit and then determined the CES-D-10 class [11]. The algorithm used 137 candidate predictors available in the CCC, selected based on the literature and subject matter expertise. Predictors fell into one of five major categories: questions related to mental health, health related quality of life, sociodemographic, behavioral, and clinical characteristics. To develop the algorithm, we split the analytical sample into training and testing data, using the recommended 80:20 split. The algorithm was then developed by 10-fold cross-validation using only the training

data and RF hyperparameters (i.e., various RF settings like number of decision trees) were tuned to maximize accuracy. The final tuned algorithms were implemented in the testing data, which was not used in the development stage. To assess the ability to distinguish between classes (discrimination), we plotted receiver operating characteristic (ROC) and estimated the area under the ROC curve (AUC) We used the default probability threshold of 0.50 for classification and at this threshold, sensitivity, specificity, positive predictive value (PPV), negative predictive value (NPV), positive likelihood ratio (LR+) and negative likelihood ratio (LR-) measures were then estimated with the 95% confidence intervals (CI). The performance evaluation characteristics of the algorithm are shown in table S1. The algorithm performed well, with an 82% (95% CI: 78-86) chance of distinguishing between the two CES-D-10 classes.

Evaluation measures (95% Cl)	RF Algorithm (x=137)
AUC	0.82 (0.78-0.86)
Sensitivity	0.77 (0.70-0.83)
Specificity	0.73 (0.66-0.79)
PPV	0.74 (0.68-0.80)
NPV	0.76 (0.69-0.82)
LR +	2.8 (2.2-3.6)
LR -	0.3 (0.2-0.4)

Table 6.4: Supplementary table - Performance evaluation measures for the RF algorithm

Abbreviations: CI: Confidence interval; AUC: Area under the Receiver Operating Characteristic curve; PPV: Positive predictive value; NPV: Negative predictive value; LR +: Positive likelihood ratio; LR -: Negative likelihood ratio

Appendix B: Causal diagram

Figure 6.3: Supplementary figure - Causal Diagram



Figure Description: A: Exposure - SVR, Y: Outcome - Depressive symptoms; Z: Time varying confounders - advanced fibrosis/cirrhosis, HIV viral load, low CD4 cell count, current injection drug use, current alcohol use, recent incarceration, and antidepressant use

Appendix C: Segmented modified Poisson model

$log (P (CES-D-10 = 1) = \beta_0 + \beta_1 time + \beta_2 pre-post + \beta_3 time * pre-post + \beta_4 x$

time = time in years pre-SVR and post- SVR

pre-post = indicator of pre and post SVR, pre-post = 0 if pre-SVR and pre-post = 1 if post-

SVR

x = time-varying confounders

 $\beta_0 = \log \text{ probability of CES-D-10 class} = 1$

 β_1 = Pre-SVR CES-D-10 class 1 probability trends

 β_2 = immediate level change in probability trend post-SVR (level change)

 β_3 = impact of DAAs on post-SVR probability trends (slope change)

Appendix D: Sensitivity analysis - non-linearity

In order to assess the possible non-linearity of the relationship between the outcome, presence of depressive symptoms and time, we developed two types of non-linear models.

1. With cubic polynomials: In this model, we included squared and cubic polynomials of time and their interaction with pre-post to explore this possible non-linear relationship. The model we used is as follows:

$$log (P(CES - D - 10 = 1))$$

 $= \beta_0 + \beta_1 pre - post + \beta_2 time + \beta_3 pre - post * time + \beta_4 time^2$

+ $\beta_5 time^2 * pre - post + \beta_6 time^3 + \beta_7 time^3 * pre - post + + \beta_8 x$

2. With restricted cubic splines: We also developed a non-linear model using a spline function for time, which are piecewise polynomials, i.e., polynomials within intervals of the variable, time in our case [12]. We included restricted cubic splines or natural splines because they are smooth and can better fit a curve than linear splines. Restricted cubic splines also have benefits compared to cubic splines as they are linear at the tails and estimate fewer parameter [12]. We developed the restricted cubic spline with 5 knots for time, positioned at the 5th (-7.1 years), 27.5th (-2.2 years), 50th (-0.2 years), 72.5th (0.7 years) and 97.5th (3.4 years) percentile. We used mkspline in Stata v.17 for this analysis [13].

We then selected the model using the quasi-likelihood under the independence model criterion (QIC), a statistic applicable to Generalized estimating equations (GEEs) [14-16]. The model selection is based on the lowest QIC, and thus, the linear models were selected in this analysis.

Models	QIC
Linear	885.9
Cubic polynomial	893.3
Restricted cubic splines	892.8
Selected model (Lowest QIC)	Linear

Appendix E: Subgroup analyses

Table 6.6: Supplementary table - Subgroup analyses - models without outcome

misclassification correction

		Adjusted risk ratios (95% CI)		
Subgroups	No. of participants	Pre-treatment trends per year	Level change at SVR	Post-SVR trends per year
Gender				
Male	321	1.02 (0.99-1.04)	0.99 (0.90-1.11)	0.95 (0.89-0.99)
Female	144	1.03 (0.99-1.06)	1.04 (0.92-1.17)	0.86 (0.80-0.93)
Race				
White	323	1.01 (0.99-1.04)	0.96 (0.87-1.1)	0.94 (0.89-0.98)
Indigenous	107	1.06 (1.01-1.09)	1.10 (0.96-1.26)	0.83 (0.75-0.93)
Employment				
Not employed	317	1.01 (0.99-1.02)	1.04 (0.96-1.13)	0.92 (0.88-0.97)
Employed	129	1.12 (1.06-1.20)	0.85 (0.68-1.05)	0.87 (0.78-0.98)
Liver disease				
No liver disease	343	1.01 (1.00-1.04)	1.04 (0.95-1.14)	0.92 (0.87-0.97)
With liver disease	90	1.04 (0.97-1.11)	0.92 (0.76-1.10)	0.89 (0.81-0.98)

Appendix F: DAA non-responders

Table 6.7: Supplementary table - Sensitivity analysis model for DAA non-responders

No. of	Adjusted risk ratios (95% CI)			
narticinants	Pre-treatment trends	Level change at	Post-SVR trends	
	per year	SVR	per year	
19 (245 visits)	1.00 (0.95-1.10)	0.98 (0.65-1.50)	1.20 (1.01-1.35)	

Figure 6.4: Supplementary figure - Impact of non-response on depressive symptoms in

DAA non-responders



Figure Description: The graph shows that pre-treatment the probability trend for presence of depressive symptoms was stable over time. There was no evidence of immediate change at date of no-response ascertainment. However, the probability trend post-no-response indicates a gradual increase in depressive symptoms over time.

References

- 1. Klein MB, Saeed S, Yang H, et al. Cohort profile: the Canadian HIV-hepatitis C coinfection cohort study. International journal of epidemiology **2010**; 39(5): 1162-9.
- Cox J, Hamelin A. Prospective investigation of the relationship between food insecurity and health and behavioural outcomes in HIV-HCV co-infection: clues for prevention interventions (CTN 264). A food security & HIV-HCV sub-study of the Canadian Co-Infection Cohort (CTN 222). . 2016.
- 3. Radloff LS. The CES-D Scale: A Self-Report Depression Scale for Research in the General Population. Applied Psychological Measurement **1977**; 1(3): 385-401.
- Andresen EM, Malmgren JA, Carter WB, Patrick DL. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). American journal of preventive medicine **1994**; 10(2): 77-84.
- Zhang W, O'Brien N, Forrest JI, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. PloS one 2012; 7(7): e40793.
- 6. Breiman L. Random Forests. Machine Learning **2001**; 45(1): 5-32.
- Fawagreh K, Gaber MM, Elyan E. Random forests: from early developments to recent advancements AU - Fawagreh, Khaled. Systems Science & Control Engineering 2014; 2(1): 602-9.
- 8. Hastie T, Tibshirani R, Friedman JH. The elements of statistical learning : data mining, inference, and prediction. Second edition. ed. New York: Springer, **2009**.

- Mennitt D, Sherrill K, Fristrup K. A geospatial model of ambient sound pressure levels in the contiguous United States. The Journal of the Acoustical Society of America 2014; 135(5): 2746-64.
- 10. Strobl C, Malley J, Tutz G. An introduction to recursive partitioning: rationale, application, and characteristics of classification and regression trees, bagging, and random forests. Psychological methods **2009**; 14(4): 323-48.
- Malley JD, Kruppa J, Dasgupta A, Malley KG, Ziegler A. Probability machines: consistent probability estimation using nonparametric learning machines. Methods Inf Med **2012**; 51(1): 74-81.
- 12. Harrell FE. Regression modeling strategies : with applications to linear models, logistic regression, and survival analysis. New York: Springer, **2001**.
- StataCorp. Stata Statistical Software: Release 17. College Station, TX: StataCorp LLC, 2021.
- 14. Cui J. QIC program and model selection in GEE analyses. Stata Journal 2007;7(2): 209-20.
- Wedderburn RWM. Quasi-Likelihood Functions, Generalized Linear Models, and the Gauss-Newton Method. Biometrika **1974**; 61(3): 439-47.
- Pan W. Akaike's information criterion in generalized estimating equations.
 Biometrics 2001; 57(1): 120-5.

7. CHAPTER 7: EFFECT OF DEPRESSIVE SYMPTOMS ON HEALTH SERVICES UTILIZATION

7.1. Preface to Manuscript 4

In this chapter, I aimed to assess the effect of depressive symptoms on HSU. HSU is defined as the "quantification of the use of services by persons for the purpose of preventing and curing health problems, promoting maintenance of health and wellbeing, or obtaining information about one's health status and prognosis" (114). It can be measured in several ways, including number of services used in a given period of time or percentage of a population using a certain service in a given time (114). Generally, a higher HSU is an indication of lower health status and leads to higher individual as well as societal economic burden (115).

There is evidence that suggests HSU is high among PLWH as compared to those without an HIV infection and this is even higher among those co-infected with HCV (116-118). This could be explained by increased co-morbidities, including psychiatric illness. Additionally, the literature regarding health services among people with chronic HCV suggests that cure or SVR may lead to lower HSU among people infected with HCV alone as well as co-infected with HIV (119, 120). However, studies quantifying the impact on HSU of specific psychiatric illnesses including depression are scarce. It would also be of interest to assess this impact of depressive symptoms on HSU in the light of the available treatment options and cure for HCV in the co-infected population in the second-generation DAA era. With Manuscript 3 indicating a gradual decline in depressive symptoms over time after SVR, it would also be interesting to study whether HCV cure impacts the relationship between depressive symptoms and HSU in the HIV-HCV co-infected population.

In this manuscript, I assessed the effect of depressive symptoms and SVR on HSU in the co-infected population in the second-generation DAA era (2013-2020).

This manuscript will be submitted for publication as a short report at Epidemiology.

This analysis is accepted for poster presentation at the following conferences/seminars:

 24th International AIDS conference (AIDS 2022), Hybrid, Montreal Canada - July 2022

7.2. Manuscript 4: Effect of depressive symptoms on health services utilization in the HIV and Hepatitis C co-infected population in Canada

Title: Effect of depressive symptoms on health services utilization in the HIV and Hepatitis C co-infected population in Canada

Gayatri Marathe^{1,2}, Erica EM Moodie¹, Marie-Josée Brouillette^{2,3}, Charlotte Lanièce Delaunay^{1,2}, Marina B. Klein^{1,2,4}; Canadian Co-Infection Cohort

- 1. Department of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada
- 2. McGill University Health Center-Research Institute, Centre for Outcomes Research and Evaluation, Montreal, Quebec, Canada
- 3. Department of Psychiatry, McGill University, Montreal, Quebec, Canada
- 4. CIHR Canadian HIV Trials Network (CTN), Vancouver, British Columbia, Canada

Corresponding author:

Marina B. Klein

Division of Infectious Diseases and Chronic Viral Illness Service

McGill University Health Centre

1001 Decarie Boulevard, D02.4110

Montreal, Canada H4A 3J1

Phone. 1-514-843-2090; Fax: 1-514-843-2092

Email address: marina.klein@mcgill.ca

Running head: Depressive symptoms and health services utilization

Sources of Support: This work was supported by Fonds de recherche du Québec – Santé; Réseau sida/maladies infectieuses, the Canadian Institute for Health Research (CIHR; FDN-143270); and the CIHR Canadian HIV Trials Network (CTN222 & CTN264). GM and CLD are supported by the PhD trainee fellowship from the Canadian Network on Hepatitis C. EEMM is supported by a chercheur de mérite award from the Fonds de recherche du Québec-Santé and a Canada Research Chair (Tier 1). MBK is supported by a Canada Research Chair (Tier I). For the remaining authors none were declared.

Conflicts of interest: MBK reports grants for investigator-initiated studies from ViiV Healthcare, AbbVie, Merck, and Gilead; and consulting fees from ViiV Healthcare, Merck, AbbVie, and Gilead. GM, EEM, MJB, and CLD have no conflict of interest to disclose.

Acknowledgements: We would like to acknowledge the participants of the Canadian Co-Infection Cohort (CTN222), the study coordinators and nurses for their assistance with study coordination, participant recruitment, and care, and the Canadian Co-Infection Cohort (CTN222) co-investigators - Drs. Lisa Barrett, Jeff Cohen, Brian Conway, Curtis Cooper, Pierre Côté, Joseph Cox, M. John Gill, Shariq Haider, David Haase, Mark Hull, Valérie Martel-Laferrière, Julio Montaner, Erica E. M. Moodie, Neora Pick, Danielle Rouleau, Aida Sadr, Steve Sanche, Roger Sandre, Mark Tyndall, Marie-Louise Vachon, Sharon Walmsley and Alexander Wong.

Ethics approval: This study was approved by the Research Ethics Board of the McGill University Health Centre (2021-6985). The CCC and the FS Sub-Study were approved by the Research Ethics Board of the McGill University Health Centre (2006-1875, BMB-06-006t, 2013-994) and the research ethics boards of participating institutions. The study was conducted according to the Declaration of Helsinki.

Consent to participate: Informed consent was obtained from all individual participants included in the study

Abstract

People living with HCV and HIV experience many co-morbid conditions including depression, contributing to high health services utilization (HSU). Successful HCV treatment may however have an impact. We examined the relationship between depressive symptoms and HSU and whether it was affected by sustained virologic response (SVR) in the HIV-HCV co-infected population. We used data from the Canadian Co-infection Cohort and predicted depressive symptoms using a random forest classifier and corrected for misclassification. Outcomes were number of inpatient and outpatient visits in the past 6 months. We restricted attention to the direct acting antiviral era and included all HCV-RNA+ participants. We fitted a zero-inflated negative binomial regression model. We included 1,153 participants, 530 were treated and 504 (95%) achieved SVR. Among those without SVR, inpatient visits were 16% higher and outpatient visits were 5% higher among those with depressive symptoms than those without, while among those with SVR, this association disappeared. SVR itself was associated with 24% fewer inpatient visits, but no change in outpatient visits. In conclusion, depressive symptoms were associated with a modest increase in HSU among those remaining HCV infected, indicating it contributes to poorer health outcomes. SVR was associated with substantial reduction in inpatient visits and an attenuated impact of depressive symptoms.

Background

Chronic HCV infection can result in end stage liver disease such as cirrhosis, its complications and hepatocellular carcinoma. It is also associated with many extrahepatic manifestations such as cryoglobulinemia, type 2 diabetes, and chronic kidney disease¹. People living with HIV are also disproportionately affected by cardiovascular disease, chronic kidney disease, osteoporosis, and cancer². People co-infected with HCV and HIV, thus, are faced with several co-morbid conditions which may result in increased health services utilization (HSU), i.e., the use of health care services for prevention, treatment, and management of health problems. Such problems may increase both inpatient hospitalizations and outpatient medical visits such as physician, emergency room or walk-in visits³⁻⁸. Greater HSU is an indicator of lower health status and has health economic implications.

Neuropsychiatric manifestations, particularly depression^{9,10} represent a common and potentially burdensome co-morbidity associated with HCV and HIV infections. Among people living with HIV, studies have shown a prevalence of depression of between 20-30% and as high as 24% among those with chronic HCV ^{1,9,11,12}. Depression prevalence is reported to be even higher in the in the co-infected population¹³. Mechanisms may be both biological via immune activation within the central nervous system and psychosocial through social stresses caused by stigma, discrimination, lack of support, and substance use^{9,14-17}.

Studies assessing HSU and depression have shown evidence of an increase in the use of health services and hospitalizations among those with depression¹⁸⁻²⁰. In a retrospective cohort study of people living with HIV in British Columbia, the interaction between co-infection with HCV and mental health disorders was assessed in relation to acute care hospitalizations. The presence of both co-infection and mental health disorders led to a larger number of hospitalizations than presence of either one ²¹. There are however limited studies looking at depressive symptoms specifically and their effect on HSU in the co-infected population. HCV cure has been shown to have a number of health benefits both with respect to liver disease and related mortality and non-liver related comorbidities which could reduce HSU. Additionally, in the Canadian Co-infection Cohort (CCC), we have shown that SVR could lead to a gradual decline in depressive symptoms over time. However, it has not been examined if SVR may hence also have an impact on this relationship between depressive symptoms and HSU in the co-infected population.

Thus, in this study, we examined the effect of presence of depressive symptoms on HSU, both inpatient and outpatient healthcare visits and how it is affected by SVR in the HIV-HCV co-infected population in Canada.

Methods

We used data from the CCC, an open multicentre prospective cohort study, ongoing since 2003. The study has been described in detail elsewhere²². Briefly, the CCC recruits from 18 HIV centres across six Canadian provinces. Eligibility criteria include \geq 16 years of age,

documented HIV infection, and evidence of HCV infection (HCV RNA positive and/or HCV seropositive). The study had recruited 2018 participants as of July 2020. Participants are followed longitudinally every 6 months and sociodemographic, behavioural, and clinical data are collected by a standardized self-administered questionnaires and chart reviews. We also used data from an associated sub-study, the Food Security and HIV-HCV co-infection study (FS sub-study). Participants for the FS sub-study were recruited from the CCC (n=725) between 20212-15. The sub-study is fully described in detail elsewhere²³, Depression screening was performed in the sub-study using the Center for Epidemiologic Studies Depression Scale-10 (CES-D-10) to assess the presence and severity of depressive symptoms in the past week²⁴. We obtained a measure of depressive symptoms in the full CCC using a random forest (RF) classifier. The CES-D-10 class (cut-off \geq 10 for presence of depressive symptoms indicative of a risk of depression) was predicted for each CCC visit ^{25,26}.

In the final analytical sample, we included all participants who were HCV RNA positive in the second generation DAA era, i.e., after November 25, 2013, when the first 2nd generation DAA, Simeprevir, was approved for use by Health Canada. We followed the HCV RNA positive participants until death, withdrawal, or end of study period (July 2020). We considered participants to have achieved SVR if they were treated with 2nd generation DAAs and had undetectable HCV viral load at 12 weeks post-end of treatment.

We developed zero-inflated negative binomial (ZINB) models and obtained the incidence rate ratios (IRR) with 95% confidence intervals (CI). The exposure was the RF predicted

CES-D-10 class for presence of depressive symptoms at each included CCC visit. We addressed possible exposure misclassification using a predictive value-based recordlevel correction method²⁷. We applied the positive predictive value (PPV) and negative predictive value (NPV) estimated for the RF algorithm; PPV: 0.70 (95% CI: 0.63-0.76) and NPV: 0.69 (95% CI: 0.62-0.76). The two HSU outcomes were: 1. Self-reported number of inpatient visits (overnight hospital stays and emergency room visits) in the past 6 months and 2. Self-reported number of outpatient visits (general practitioner, HIV clinic, specialist, and walk-in clinic visits) in the past 6 months. ZINB models accounted for the over-dispersion and excess zeroes in these HSU outcomes (Figure 7.1). We used a clustered sandwich estimator for standard errors in order to account the intragroup correlation. We included an interaction term for those who achieved SVR. We adjusted for confounders including - baseline: age, gender, race/ethnicity, education level, immigration, and province, employment, monthly income, injection drug use, incarceration, fibrosis stage, HIV viral load and CD4 count; time-varying: HCV RNA status. We conducted multiple imputation by chained equations (MICE) to address the missing data in the confounders and created five imputed datasets ^{28,29}.

Results

We included 1,153 CCC participants who were HCV RNA positive. Of the 530 participants who initiated treatment, 504 (95%) achieved SVR. The participants were predominantly male (70%) with a median baseline age of 45 years (interquartile range (IQR): 38-51); 29% were Indigenous, 66% were unemployed, 73% had no post-secondary education, and 38% were current injection drug users. Thus, the participants were vulnerable and

possibly could face numerous potential barriers to HCV treatment and mental health care³⁰. At baseline, 61% had predicted CES-D-10 scores \geq 10. The median number of inpatient visits in the past 6 months was 0 (interquartile range (IQR): 0-1) and for outpatient visits was 3 (IQR: 1-6) (see Figure 7.1).

The model results are shown in table 7.1. The models indicate that among those who had not achieved SVR, the number of inpatient visits were 17% higher in those with depressive symptoms compared to those without (IRR: 1.17; 95% CI: 1.06-1.29) and outpatient visits were 5% higher (IRR: 1.05; 95% CI: 1.02-1.08). In contrast, among those who achieved SVR, the models showed no difference in inpatient and outpatient visits with respect to predicted depressive symptoms. SVR was associated with a considerable reduction in inpatient visits with 24% fewer visits (IRR: 0.76; 95% CI: 0.70-0.84) compared to those without SVR; however, no effect of SVR was observed on the number of outpatient visits (IRR: 1.00; 95% CI: 0.97-1.02).

Discussion

Depressive symptoms were associated with a modest increase in HSU, indicating a greater burden on both individuals and the health systems. SVR was associated with a substantial reduction in inpatient visits, and SVR appeared to attenuate the effect of depressive symptoms on HSU.

Depressive symptoms have been associated with increased HSU in the co-infected population in previous studies. Similarly, SVR has been reported to reduce HSU among

people infected with HCV alone as well as those co-infected with HCV and HIV ^{31,32}. From our studies, it appears however that the overall improvement in health due to SVR may not only directly lead to lower healthcare use for serious HCV associated comorbidities, but also for depressive symptoms themselves (e.g., lowering of severity of depressive symptoms ³³). Thus, indirectly, a better state of general health/reduced hospitalisation after SVR may also help people experiencing depressive symptoms to manage their depressive symptoms without needing to access acute healthcare resources. This possible lowering of burden on health systems due to HCV cure could be indicative of major system-level economic benefits, beyond the patient level improved outcomes.

Our results are thus, consistent with these prior findings, but in addition provide evidence that there could be a modifying effect of SVR in the relationship between depressive symptoms and HSU. Interestingly, SVR reduced inpatient visits including hospitalizations, but not outpatient visits. One possible explanation in people with HIV-HCV co-infection could be continued engagement in care for HIV and for other co-morbid conditions that are not affected by HCV cure.

This study has several strengths. We used longitudinal data collected over the last 18 years in many diverse patients. Our estimates are generalizable to HIV-HCV coinfected patients engaged in care in Canada because the CCC recruits participants from primary and tertiary care clinics in urban and semi-urban areas across six provinces in Canada and includes 16% of the co-infected population in Canada. The study provides a real-world learning example of skewed distributions as seen in the HSU outcome, and
modeling methods that can be used in case of excess zeroes and overdispersion. The study, however, does have some limitations. The depressive symptoms were predicted via a random forest algorithm, and not measured directly in the cohort. Thus, misclassification was expected and therefore we corrected for potential misclassification in our analysis. We predicted the depressive symptoms based on a screening questionnaire, CES-D-10, and not a major depression diagnosis. Thus, this study does not provide an effect estimate for depression but rather for depressive symptoms that are indicative of a depression risk.

In conclusion, our study shows that depressive symptoms are associated with increased HSU and if treated for HCV this could be reduced, further supporting the benefits of universal treatment of HCV among people co-infected with HCV and HIV.

Figure 7.1: Distribution of number of self-reported inpatient and outpatient visits in the past 6 months recorded at each participant visit (n = 1,153)



Histogram of inpatient visits in the past 6 months

Abbreviations: IQR: Interquartile range 193

Table 7.1: Effect of depressive symptoms and SVR on HSU in the HIV-HCV co-infected population; all IRRs are relative to those without depressive symptoms and without (or prior to) SVR

Outcomes	Depressive symptoms - Among those who had not achieved SVR		SVR		Depressive symptoms - Among those who achieved SVR	
	IRR	95% CI	IRR	95% CI	IRR	95% CI
Inpatient visits	1.17	1.06-1.29	0.76	0.70-0.84	1.03	0.89-1.19
Outpatient visits	1.05	1.02-1.08	1.00	0.97-1.02	1.01	0.97-1.05

Abbreviations: SVR: Sustained virologic response; HSU: Health services utilization; IRR: Incidence rate ratio; CI: Confidence interval

References

- Younossi Z, Park H, Henry L, Adeyemi A, Stepanova M. Extrahepatic Manifestations of Hepatitis C: A Meta-analysis of Prevalence, Quality of Life, and Economic Burden. *Gastroenterology* 2016;**150**(7):1599-1608.
- Lerner AM, Eisinger RW, Fauci AS. Comorbidities in Persons With HIV: The Lingering Challenge. JAMA 2020;323(1):19-20.
- 3. Norton BL, Park L, McGrath LJ, et al. Health care utilization in HIV-infected patients: assessing the burden of hepatitis C virus coinfection. *AIDS Patient Care STDS* 2012;**26**(9):541-5.
- Ma H, Villalobos CF, St-Jean M, et al. The impact of HCV co-infection status on healthcare-related utilization among people living with HIV in British Columbia, Canada: a retrospective cohort study. *BMC Health Serv Res* 2018;**18**(1):319.
- 5. Baum MK, Jayaweera DT, Duan R, et al. Quality of life, symptomatology and healthcare utilization in HIV/HCV co-infected drug users in Miami. *J Addict Dis* 2008;**27**(2):37-48.
- Linas BP, Wang B, Smurzynski M, et al. The impact of HIV/HCV co-infection on health care utilization and disability: results of the ACTG Longitudinal Linked Randomized Trials (ALLRT) Cohort. J Viral Hepat 2011;18(7):506-12.
- Crowell TA, Gebo KA, Balagopal A, et al. Impact of hepatitis coinfection on hospitalization rates and causes in a multicenter cohort of persons living with HIV. *J Acquir Immune Defic Syndr* 2014;65(4):429-37.

- 8. Crowell TA, Berry SA, Fleishman JA, et al. Impact of hepatitis coinfection on healthcare utilization among persons living with HIV. *J Acquir Immune Defic Syndr* 2015;**68**(4):425-31.
- Nanni MG, Caruso R, Mitchell AJ, Meggiolaro E, Grassi L. Depression in HIV infected patients: a review. *Curr Psychiatry Rep* 2015;**17**(1):530.
- 10. Yeoh SW, Holmes ACN, Saling MM, Everall IP, Nicoll AJ. Depression, fatigue and neurocognitive deficits in chronic hepatitis C. *Hepatol Int* 2018.
- 11. Bing EG, Burnam MA, Longshore D, et al. Psychiatric disorders and drug use among human immunodeficiency virus-infected adults in the United States. *Arch Gen Psychiatry* 2001;**58**(8):721-8.
- 12. Ciesla JA, Roberts JE. Meta-analysis of the relationship between HIV infection and risk for depressive disorders. *Am J Psychiatry* 2001;**158**(5):725-30.
- Fialho R, Pereira M, Rusted J, Whale R. Depression in HIV and HCV co-infected patients: a systematic review and meta-analysis. *Psychol Health Med* 2017;**22**(9):1089-1104.
- Yarlott L, Heald E, Forton D. Hepatitis C virus infection, and neurological and psychiatric disorders – A review. *Journal of Advanced Research* 2017;8(2):139-148.
- 15. Strazza M, Pirrone V, Wigdahl B, Nonnemacher MR. Breaking down the barrier: the effects of HIV-1 on the blood-brain barrier. *Brain Res* 2011;**1399**:96-115.
- 16. Wilkinson J, Radkowski M, Laskus T. Hepatitis C virus neuroinvasion: identification of infected cells. *J Virol* 2009;**83**(3):1312-9.

- Kalichman SC, Washington C, Kegler C, et al. Continued Substance Use Among People Living With HIV-Hepatitis-C Co-Infection and Receiving Antiretroviral Therapy. Substance use & misuse 2015;50(12):1536-1543.
- 18. Guo J, Kong D, Fang L, Zhu Y, Zhang B. Depressive symptoms and health service utilization among Chinese middle-aged and older adults: a national population-based longitudinal survey. *International Journal of Mental Health Systems* 2021;**15**(1):2.
- Kong D, Li M, Wang J, Davitt JK, Dong X. The Relationship Between Depressive Symptoms and Health Services Utilization in U.S. Chinese Older Adults. *Gerontologist* 2019;**59**(3):447-455.
- 20. Tusa N, Koponen H, Kautiainen H, et al. The profiles of health care utilization among a non-depressed population and patients with depressive symptoms with and without clinical depression. *Scand J Prim Health Care* 2019;**37**(3):312-318.
- 21. St-Jean M, Tafessu H, Closson K, et al. The syndemic effect of HIV/HCV coinfection and mental health disorders on acute care hospitalization rate among people living with HIV/AIDS: a population-based retrospective cohort study. *Can J Public Health* 2019;**110**(6):779-791.
- 22. Klein MB, Saeed S, Yang H, et al. Cohort profile: the Canadian HIV-hepatitis C coinfection cohort study. *Int J Epidemiol* 2010;**39**(5):1162-9.
- Cox J, Hamelin AM, McLinden T, et al. Food Insecurity in HIV-Hepatitis C Virus Co-infected Individuals in Canada: The Importance of Co-morbidities. *AIDS Behav* 2017;**21**(3):792-802.

- 24. Radloff LS. The CES-D Scale: A Self-Report Depression Scale for Research in the General Population. *Applied Psychological Measurement* 1977;**1**(3):385-401.
- 25. Breiman L. Random Forests. *Machine Learning* 2001;**45**(1):5-32.
- Fawagreh K, Gaber MM, Elyan E. Random forests: from early developments to recent advancements. *Systems Science & Control Engineering* 2014;**2**(1):602-609.
- Banack HR, Stokes A, Fox MP, et al. Stratified Probabilistic Bias Analysis for Body Mass Index-related Exposure Misclassification in Postmenopausal Women. *Epidemiology (Cambridge, Mass.)* 2018;**29**(5):604-613.
- Raghunathan TE, Lepkowski JM, Van Hoewyk J, Solenberger P. A Multivariate Technique for Multiply Imputing Missing Values Using a Sequence of Regression Models. SURVEY METHODOLOGY 2001;27 (Part 1):85-96.
- 29. van Buuren S. Multiple imputation of discrete and continuous data by fully conditional specification. *Stat Methods Med Res* 2007;**16**(3):219-42.
- 30. Saeed S, Strumpf EC, Moodie EE, et al. Disparities in direct acting antivirals uptake in HIV-hepatitis C co-infected populations in Canada. *J Int AIDS Soc* 2017;**20**(3).
- 31. Yeung MW, Young J, Moodie E, et al. Changes in quality of life, healthcare use, and substance use in HIV/hepatitis C coinfected patients after hepatitis C therapy: a prospective cohort study. *HIV Clin Trials* 2015;**16**(3):100-10.
- Smith-Palmer J, Cerri K, Valentine W. Achieving sustained virologic response in hepatitis C: a systematic review of the clinical, economic and quality of life benefits.
 BMC Infectious Diseases 2015;15(1):19.

 Marathe G., Moodie E.E.M. et al. Impact of HCV cure on depressive symptoms in the HIV-HCV coinfected population [CROI Abstract 535]. In Special Issue: Abstracts From the 2022 Conference on Retroviruses and Opportunistic Infections. Top Antiv Med. 2022;30(1s):206.

8. CHAPTER 8: DISCUSSION

8.1. Summary of findings

Depression is common in the HIV and HCV co-infected population and was a major side effect of the IFN-based HCV treatment regimen. Thus, treatment was relatively contraindicated in people with past or current psychiatric illness. With advent of the safer and more effective DAA regimens, HCV treatment is now more accessible. However, in spite of the overall increase in treatment uptake, there are still many patient- and systemlevel barriers. Depression is one such barrier which could have an effect on clinical and other outcomes in people living with HIV and HCV. Thus, in this dissertation, I investigated depressive symptoms in the HIV and HCV co-infected population in Canada.

In Manuscript 1, I aimed to predict presence of depressive symptoms indicative of being at risk for major depression in the CCC, since depression screening was only done in a subset of participants in the FS sub-study. I developed two random forest algorithms using predictor data from the CCC and the CES-D-10 screening tool data from the FS sub-study - one with all candidate predictors and the second with a reduced subset of predictors. Both algorithms had good accuracy. I found that the most important predictors were the health-related quality of life indicators, employment, HIV clinical stage, revenue source, BMI, and education. This is the first study which used machine learning algorithms for predicting presence of depressive symptoms in the HIV-HCV co-infected population. The reduced algorithm that used fewer than half of the predictors of the full algorithm still had acceptable accuracy. Thus, it could be useful in other studies without direct measures of depressive symptoms, after external validation to rule out overfitting.

I was able to use these algorithms to predict the CES-D-10 class at each CCC visit; this was then used as a measure of depressive symptoms in the next thesis objectives. I corrected for possible misclassification using a predictive value-based record level correction method, using the estimated positive and negative predictive values of the algorithm.

The aim of Manuscript 2 was to assess the effect of depressive symptoms on HCV treatment initiation in the IFN (2003-2011) and the second-generation DAA (2013-2020) eras. I observed an overall increase in HCV treatment initiation rates from the IFN to the DAA era and the percentage of eligible participants initiating HCV treatment increased from 21% to 50%. There was lower HCV treatment initiation among those with depressive symptoms compared to those without in the IFN era, but higher initiation among those with depressive symptoms in the DAA era.

In Manuscript 3, my goal was to evaluate the impact of SVR on presence of depressive symptoms over time. I observed that depressive symptoms changed very little over time in the lead up to DAA treatment and there was no evidence of immediate improvement at SVR. There was a gradual decline in depressive symptoms over time post-SVR among patients co-infected with HIV. The improvement after cure may reflect changes in biological pathways and/or improved general physical health.

Finally, Manuscript 4 aimed to assess the effect of depressive symptoms and SVR on HSU in the Canadian HIV-HCV co-infected population. I found that depressive symptoms

were associated with a modest increase in HSU, indicating that these symptoms contribute to poor health outcomes. SVR was associated with a substantial reduction in inpatient visits. Additionally, SVR appeared to attenuate the effect of depressive symptoms on HSU. This analysis provides a real-world learning example of modeling methods that can be used in case of excess zeroes and overdispersion to address the highly skewed distributions typical of HSU.

8.2. Strengths and limitations

The major strength of this thesis is that the CCC has high quality longitudinal data which is generalizable to the HIV-HCV co-infected population engaged in care in Canada. This is because the study recruits from a variety of clinical settings (outreach, primary, and tertiary care clinics in urban and semi-urban areas across the country) and includes approximately 16% of the HIV-HCV co-infected population in Canada. The CCC began in 2003 and thus I was able to leverage 19 years of longitudinal data and answer important questions with advanced analytical methods.

In Manuscript 1, I was able to use data from a sub-study within the CCC about food security to develop a prediction algorithm for depressive symptoms in the CCC, as the FS sub-study sample was generalizable to the parent CCC (see Table 4.7). I used the ensemble machine learning method of random forests, which is non-parametric, highly accurate, and relatively robust to outliers and noise. In addition, the predictor importance plots provided some insight regarding predictors that play a major role classification.

Random forests have safeguards against overfitting and thus improve chances of applicability beyond the data used to develop the classifier. Nevertheless, external validation is needed before application in other cohorts and research to mitigate the risk of overfitting.

However, there are also several limitations for Manuscript 1. I used the screening tool of CES-D-10 to assess the presence of depressive symptoms, which may be indicative of being at risk for major depression, as we did not have clinical diagnosis via a gold standard interview such as the SCID-5. Thus, I cannot make any of the conclusions in this thesis for depression, but for the specific construct of depressive symptoms that is measured by the CES-D-10 scale, as described in the section 2.7 of chapter 2. Further, I categorized the CES-D-10 to create the binary classes, and thus may have lost some information by not predicting the unit CES-D-10 scores and the validity of the cut-off of 10 could not be assessed directly in this sample, although this threshold has been assessed in HIV populations in Canada in a previous study (99). In fact, I did develop a regression algorithms to predict the continuous CES-D-10 scores in additional analysis E, but with a R-squared between 0.3-0.5, these algorithms could only explain a small portion of the variability in the outcome and were not able to predict the CES-D-10 score at a unit level with high accuracy. The sample size in this study was small for prediction models. Finally, some predictors described in other studies such as childhood trauma and food insecurity among others, were not available for the full CCC.

In the next thesis manuscripts, the clinically relevant depressive symptoms were predicted using the full random forest algorithm from Manuscript 1, and not measured directly in the cohort. Thus, misclassification was expected and therefore I needed to correct for potential misclassification in all three manuscripts, for which I used a recordlevel correction method using positive and negative predictive values of the algorithm.

The major strength of Manuscript 2 was that I applied robust methods to account for time-dependent confounding, competing risks and conducted multiple secondary analyses. Manuscript 4 provides a real-world example of methods taking into consideration the skewness of the data. A limitation of both Manuscripts 2 and 4 is that they do not provide an effect estimate for major depression but rather for these (predicted) clinically relevant depressive symptoms and further research with clinical diagnosis of depression using a structured interview like SCID-5, which is the gold standard for major depression diagnosis, is crucial.

In Manuscript 3, using the quasi-experimental analytic design of ITS, I was able to obtain robust marginal effect estimates of the impact of SVR on depressive symptoms in the co-infected population. I conducted multiple sensitivity analyses to account for possible issues such as lead time bias and non-linearity and adjusted for time-varying confounders in the model. However, the crucial assumption of the ITS design that the extrapolated pre-exposure trend is considered the counterfactual trend, makes it vulnerable to unmeasured time-varying confounding. With no available appropriate comparison group, I tackled this issue by adding known time-varying

confounders to the model; however, some residual confounding could still be possible. Another limitation to consider is that the median post-SVR follow-up time was 1.7 years, so the durability of the observed effect still needs to be explored with additional data.

8.3. Implications and future directions

I investigated several different questions in this thesis which are important in strengthening the available literature base regarding depressive symptoms in this highly vulnerable population of HIV-HCV co-infected individuals. First, I found that it is possible to use machine learning algorithms like the ones I developed in this thesis to accurately predict depressive symptom classes based on screening tools like the CES-D-10. Depression being a common neuropsychiatric disorder in the co-infected population, it is important to understand its impact overall as well as on specific outcomes. I used predictors, especially in the reduced algorithm, which may be routinely collected in longitudinal or cross-sectional studies in this population. Thus, this provides a basis for use of such algorithms in research studies that may not have previously screened for depression, but have rich demographic, behavioural, and clinical data. These studies can retrospectively predict presence of depressive symptoms and explore their impact in this study population in different settings. In terms of future directions, it is important to externally validate the algorithms in other datasets which have the same measured predictors, in order to rule out overfitting and assess generalizability.

As described in the introduction, depression has been part of the discussion in this population not only due to its high prevalence but also as a side effect of the IFN-based HCV treatments. The overall increase in treatment uptake was observed from the IFN to the DAA era. This is a step in the right direction, however further improvement is essential to improve treatment uptake and reach the elimination goals set by the Canadian blueprint for HCV elimination and WHO (2, 29). I found that depression may no longer be a barrier to HCV treatment initiation in the second generation DAA era in Canada in the HCV-HIV co-infected population. This study is the first to quantify this effect of depressive symptoms on HCV treatment initiation in the co-infected population and also show the difference between the DAA era and IFN era. It was observed that people with depressive symptoms were initiating treatment more than those without in the DAA era. As described earlier, one of the explanations for this improvement in treatment initiation among people with depressive symptoms could be that patients for whom IFN-based treatments were contraindicated, or were not tolerated, are now accessing treatment, i.e., a warehousing effect.

The warehousing effect has been described in the HCV literature as one where the near-term prospect of availability of DAAs with better cure rates and improved side-effects led to deferring treatment at a later date when DAAs became widely available (121). Thus, individuals with past or current psychiatric illness, as well as those harder to treat or more susceptible to side effects with IFN like those with cirrhosis and hepatocellular carcinoma, were deferred treatment. Thus, this warehousing effect is likely to have played a role in this observed increase in

treatment initiation among those with depressive symptoms following the availability of DAA. In an attempt to further understand if this effect could persist or plateau over time, in a crude analysis, we looked at treatment initiations among those with baseline depressive symptoms comparing early and the later years of the second-generation DAA era in the CCC. We observed that 65% of the people with baseline depressive symptoms initiated DAAs between 2015-2017 and comparatively lower number of people (31%) initiated between 2018-2020, which provides some evidence for possible plateauing over time. However, overall treatment rates have also been reported to have fallen once those waiting for treatment were cured (122). Hence, further analysis with extended follow-up is important to assess the durability of the observed effect using methodology for incorporating a time-varying exposure definition for depressive symptoms.



Figure 8.1: Patient perceived barriers and facilitators to DAA treatment uptake

Reproduced from Int J Drug Policy 96, Amoako A. et.al. Patient and provider perceived barriers and facilitators to direct acting antiviral hepatitis C treatment among priority populations in high income countries: A knowledge synthesis, 103247 (2021)

Depression and other mental health issues are still widely described as patient-level barriers to linkage to care in this population. There can be many reasons for this, including lack of diagnosis and treatment and that real-world health systems are unable to deal with multiple co-morbid conditions. A recent qualitative study by Amoako et.al., which included published and grey literature (post-2014), examined patient perceived barriers and facilitators of DAA treatment uptake in key priority populations including PWID, gbMSM, and Indigenous people (123). The various barriers and facilitators found are shown in figure 8.1. The patient level barriers still included competing social, psychological, and mental health issues. They discussed the literature showing that high toxicity and sideeffects of IFN-based therapy still casts a shadow leading some patients as well as providers to fear initiating treatment in the presence of competing health issues. In an article by Ortiz-Paredes et. al., they conducted focus groups to determine barriers and facilitators for DAA uptake in the HIV co-infected population in Canada and found that multidisciplinary care fostering a strong supportive network and intrinsically motivated patients along with HCV education emerged as key facilitators (124).

There is growing discussion in the HIV and HCV literature about syndemics, which is defined as a "population-level clustering of social and health problems", i.e., health conditions, and social, behavioral and structural determinants interact synergistically, leading to excess burden of disease in the affected population (125). Syndemics were described by Tsai et. al. (2017) as "a multi-level phenomena with three underlying concepts, disease concentration, disease interaction, and large-scale social forces" (126). The consideration of syndemics in relation to HIV and HCV include mental health

and substance use; efforts need to be taken in terms of integration of care in order to deal with the individual as well as joint effects of these conditions (127-129). Thus, despite evidence from my analysis in Manuscript 2 of depressive symptoms no longer being a barrier to treatment initiation quantitatively, depression could still have an impact overall on linkage to care post-cure and on other health outcomes. Screening, and possible treatment after clinical diagnosis are therefore essential for improving treatment uptake and linkage to care. This might be achieved by integration of mental health education, screening, and treatment as a part of routine care for HIV-HCV co-infected people.

In the next manuscripts, I found that SVR seems to have a direct effect on depressive symptoms over time leading to a gradual decline and that depressive symptoms lead to an increase in HSU, but this is reduced among those who had achieved SVR. Thus, SVR or HCV cure are shown here to have led to improvement in patient outcomes beyond liver outcomes and mortality. This provides further rationale for increasing treatment access and supporting adherence to treatment in all people chronically infected with HCV. Further research will be essential using longitudinal data over a longer follow-up post-SVR to assess the durability of these effects. Additionally, outcomes such as liver disease, mental health issues including depression and overall HSU add tremendous economic burden on both individuals and health systems (115, 130-133). DAAs are infamously known for their access-limiting high costs (134), although prices have steadily declined. Many studies have, however, shown DAAs to be cost-effective and sometimes cost saving in a number of settings and populations with different co-morbid conditions (135-137). As this thesis provides further evidence of advantages of DAA beyond

improved liver disease outcomes, we need to consider possible lowering of mental illness and other HSU-related costs in future work assessing the cost-effectiveness of treating all chronically infected persons with DAA regimens.

8.4. Conclusions

With the advent of the DAA regimens, the HCV landscape changed dramatically with possible elimination on the horizon. Depression was a major side effect and barrier to HCV treatment in the IFN era, so my aim was to understand the current role of depressive symptoms in the co-infected population in this DAA era. I observed that, first, depressive symptoms are no longer a barrier to HCV treatment initiation in the DAA era and second, the health benefits of curing HCV extend beyond improving liver disease, to improving mental health symptoms and reducing health service use which provides further rationale for treating HCV in all chronically infected persons.

REFERENCES

1. Global progress report on HIV, viral hepatitis and sexually transmitted infections,2021. Accountability for the global health sector strategies 2016–2021: actions for impact. Geneva: World Health Organization; 2021. Report No.: Licence: CC BY-NC-SA 3.0 IGO.

2. Guidelines for the care and treatment of persons diagnosed with chronic hepatitis C virus infection. Geneva: World Health Organization; 2018. Report No.: License: CC BY-NC-SA 3.0 IGO.

3. Nanni MG, Caruso R, Mitchell AJ, Meggiolaro E, Grassi L. Depression in HIV infected patients: a review. Current psychiatry reports. 2015;17(1):530.

4. Younossi Z, Park H, Henry L, Adeyemi A, Stepanova M. Extrahepatic Manifestations of Hepatitis C: A Meta-analysis of Prevalence, Quality of Life, and Economic Burden. Gastroenterology. 2016;150(7):1599-608.

5. Fialho R, Pereira M, Rusted J, Whale R. Depression in HIV and HCV co-infected patients: a systematic review and meta-analysis. Psychology, health & medicine. 2017;22(9):1089-104.

6. Loftis JM, Hauser P. The phenomenology and treatment of interferon-induced depression. Journal of Affective Disorders. 2004;82(2):175-90.

7. Gallach M, Vergara M, da Costa JP, Miquel M, Casas M, Sanchez-Delgado J, et al. Impact of treatment with direct-acting antivirals on anxiety and depression in chronic hepatitis C. PLOS ONE. 2018;13(12):e0208112-e.

8. Sundberg I, Lannergard A, Ramklint M, Cunningham JL. Direct-acting antiviral treatment in real world patients with hepatitis C not associated with psychiatric side effects: a prospective observational study. BMC Psychiatry. 2018;18(1):157.

9. Saeed S, Strumpf EC, Moodie EE, Young J, Nitulescu R, Cox J, et al. Disparities in direct acting antivirals uptake in HIV-hepatitis C co-infected populations in Canada. Journal of the International AIDS Society. 2017;20(3).

10. Ma H, Villalobos CF, St-Jean M, Eyawo O, Lavergne MR, Ti L, et al. The impact of HCV co-infection status on healthcare-related utilization among people living with HIV in British Columbia, Canada: a retrospective cohort study. BMC Health Services Research. 2018;18(1):319.

11. Fleury M-J, Grenier G, Bamvita J-M, Caron J. Determinants and patterns of service utilization and recourse to professionals for mental health reasons. BMC Health Services Research. 2014;14(1):161.

12. Chiu M, Lebenbaum M, Cheng J, de Oliveira C, Kurdyak P. The direct healthcare costs associated with psychological distress and major depression: A population-based cohort study in Ontario, Canada. PLOS ONE. 2017;12(9):e0184268-e.

13. Wong WWL, Haines A, Bremner KE, Yao Z, Calzavara A, Mitsakakis N, et al. Health care costs associated with chronic hepatitis C virus infection in Ontario, Canada: a retrospective cohort study. CMAJ Open. 2021;9(1):E167.

14. Estimates of HIV Incidence, Prevalence and Canada's Progress On Meeting The 90-90-90 HIV Targets, 2018. Public Health Agency of Canada; 2020. Report No.: ISBN: 978-0-660-36587-9.

15. Khan P, Parry S. Epidemiology and Natural History of HIV. Tutorial Topics in Infection for the Combined Infection Training Programme. <u>https://oxford.universitypressscholarship.com/view/10.1093/oso/9780198801740.001.00</u> 01/isbn-9780198801740-book-part-66: Oxford University Press. ; 2019.

16. Günthard HF, Aberg JA, Eron JJ, Hoy JF, Telenti A, Benson CA, et al. Antiretroviral treatment of adult HIV infection: 2014 recommendations of the International Antiviral Society-USA Panel. JAMA. 2014;312(4):410-25.

17. Bavinton BR, Pinto AN, Phanuphak N, Grinsztejn B, Prestage GP, Zablotska-Manos IB, et al. Viral suppression and HIV transmission in serodiscordant male couples: an international, prospective, observational, cohort study. The Lancet HIV. 2018;5(8):e438-e47.

18. Thompson MA, Aberg JA, Hoy JF, Telenti A, Benson C, Cahn P, et al. Antiretroviral treatment of adult HIV infection: 2012 recommendations of the International Antiviral Society-USA panel. JAMA. 2012;308(4):387-402.

19. Trickey A, May MT, Vehreschild J-J, Obel N, Gill MJ, Crane HM, et al. Survival of HIV-positive patients starting antiretroviral therapy between 1996 and 2013: a collaborative analysis of cohort studies. The Lancet HIV. 2017;4(8):e349-e56.

20. Indigenous Services Canada. Annual Report to Parliament. 2020.

21. Melhuish A, Lewthwaite P. Natural history of HIV and AIDS. Medicine. 2018;46(6):356-61.

22. Engelman A, Cherepanov P. The structural biology of HIV-1: mechanistic and therapeutic insights. Nat Rev Microbiol. 2012;10(4):279-90.

 Marcus JL, Leyden WA, Alexeeff SE, Anderson AN, Hechter RC, Hu H, et al. Comparison of Overall and Comorbidity-Free Life Expectancy Between Insured Adults With and Without HIV Infection, 2000-2016. JAMA Network Open. 2020;3(6):e207954-e.
 Lerner AM, Eisinger RW, Fauci AS. Comorbidities in Persons With HIV: The Lingering Challenge. JAMA. 2020;323(1):19-20.

25. Polaris Observatory HCV Collaborators. Global prevalence and genotype distribution of hepatitis C virus infection in 2015: a modelling study. The Lancet Gastroenterology and Hepatology. 2017;2(3):161-76.

26. Spearman CW, Dusheiko GM, Hellard M, Sonderup M. Hepatitis C. Lancet (London, England). 2019;394(10207):1451-66.

27. Trubnikov M, Yan P, Archibald C. Estimated prevalence of Hepatitis C Virus infection in Canada, 2011. Canada Communicable Disease Report. 2014;40(19):429-36.

28. Alter MJ. Epidemiology of hepatitis C virus infection. World J Gastroenterol. 2007;13(17):2436-41.

29. The Canadian Network on Hepatitis C Blueprint Writing Committee and Working Groups. Blueprint to inform hepatitis C elimination efforts in Canada. Montreal, QC.

30. Høj SB, Minoyan N, Artenie AA, Grebely J, Bruneau J. The role of prevention strategies in achieving HCV elimination in Canada: what are the remaining challenges? Canadian Liver Journal. 2018;1(2):4-13.

Shepard CW, Finelli L, Alter MJ. Global epidemiology of hepatitis C virus infection.
 The Lancet Infectious Diseases. 2005;5(9):558-67.

32. Terrault NA, Dodge JL, Murphy EL, Tavis JE, Kiss A, Levin TR, et al. Sexual transmission of hepatitis C virus among monogamous heterosexual couples: The HCV partners study. Hepatology. 2013;57(3):881-9.

33. Ti L, Lima V, Hull M, Nosyk B, Joy J, Montaner J, et al. Hepatitis C testing in Canada: don't leave baby boomers behind. CMAJ : Canadian Medical Association journal = journal de l'Association medicale canadienne. 2017;189(25):E870-E1.

34. Smith BD, Morgan RL, Beckett GA, Falck-Ytter Y, Holtzman D, Teo CG, et al. Recommendations for the identification of chronic hepatitis C virus infection among persons born during 1945-1965. MMWR Recomm Rep. 2012;61(Rr-4):1-32.

35. Spittal PM, Pearce ME, Chavoshi N, Christian WM, Moniruzzaman A, Teegee M, et al. The Cedar Project: high incidence of HCV infections in a longitudinal study of young Aboriginal people who use drugs in two Canadian cities. BMC Public Health. 2012;12:632.

36. Gordon J, Bocking N, Pouteau K, Farrell T, Ryan G, Kelly L. First Nations hepatitis C virus infections: Six-year retrospective study of on-reserve rates of newly reported infections in northwestern Ontario. Can Fam Physician. 2017;63(11):e488-e94.

37. Alazard-Dany N, Denolly S, Boson B, Cosset FL. Overview of HCV Life Cycle with a Special Focus on Current and Possible Future Antiviral Targets. Viruses. 2019;11(1).

38. Hajarizadeh B, Grebely J, Dore GJ. Epidemiology and natural history of HCV infection. Nature reviews Gastroenterology & hepatology. 2013;10(9):553-62.

39. Maheshwari A, Ray S, Thuluvath PJ. Acute hepatitis C. The Lancet. 2008;372(9635):321-32.

40. Kish T, Aziz A, Sorio M. Hepatitis C in a New Era: A Review of Current Therapies. Pharmacy and Therapeutics. 2017;42(5):316-29.

41. Axley P, Ahmed Z, Ravi S, Singal AK. Hepatitis C Virus and Hepatocellular Carcinoma: A Narrative Review. J Clin Transl Hepatol. 2018;6(1):79-84.

42. Younossi ZM, Birerdinc A, Henry L. Hepatitis C infection: A multi-faceted systemic disease with clinical, patient reported and economic consequences. Journal of Hepatology. 2016;65(1 Suppl):S109-s19.

43. Messina JP, Humphreys I, Flaxman A, Brown A, Cooke GS, Pybus OG, et al. Global distribution and prevalence of hepatitis C virus genotypes. Hepatology (Baltimore, Md). 2015;61(1):77-87.

44. Platt L, Easterbrook P, Gower E, McDonald B, Sabin K, McGowan C, et al. Prevalence and burden of HCV co-infection in people living with HIV: a global systematic review and meta-analysis. Lancet Infect Dis. 2016;16(7):797-808.

45. Remis R. Estimating the number of persons co-infected with hepatitis c virus and human immunodeficiency virus in Canada. Health Canada. 2001.

46. Hull M, Shafran S, Wong A, Tseng A, Giguere P, Barrett L, et al. CIHR Canadian HIV Trials Network Coinfection and Concurrent Diseases Core Research Group: 2016 Updated Canadian HIV/Hepatitis C Adult Guidelines for Management and Treatment. The Canadian journal of infectious diseases & medical microbiology = Journal canadien des maladies infectieuses et de la microbiologie medicale. 2016;2016:4385643.

47. Gobran ST, Ancuta P, Shoukry NH. A Tale of Two Viruses: Immunological Insights Into HCV/HIV Coinfection. Front Immunol. 2021;12:726419.

48. Kim AY, Schulze zur Wiesch J, Kuntzen T, Timm J, Kaufmann DE, Duncan JE, et al. Impaired hepatitis C virus-specific T cell responses and recurrent hepatitis C virus in HIV coinfection. PLoS Med. 2006;3(12):e492-e.

49. Ganesan M, Poluektova LY, Kharbanda KK, Osna NA. Human immunodeficiency virus and hepatotropic viruses co-morbidities as the inducers of liver injury progression. World journal of gastroenterology. 2019;25(4):398-410.

50. Burstow NJ, Mohamed Z, Gomaa AI, Sonderup MW, Cook NA, Waked I, et al. Hepatitis C treatment: where are we now? International journal of general medicine. 2017;10:39-52.

51. Webster DP, Klenerman P, Dusheiko GM. Hepatitis C. Lancet (London, England). 2015;385(9973):1124-35.

52. Hearn B, Delbello D, Lawler J, Ng M, Harty A, Dieterich DT. Hepatitis C Virus Treatment in HIV-Coinfected Patients: No Longer Different From Monoinfection Treatment. Gastroenterol Hepatol (N Y). 2014;10(11):706-15.

53. Wyles DL, Sulkowski MS, Dieterich D. Management of Hepatitis C/HIV Coinfection in the Era of Highly Effective Hepatitis C Virus Direct-Acting Antiviral Therapy. Clinical Infectious Diseases. 2016;63 Suppl 1:S3-s11.

54. Eaton WW, Johns Hopkins Bloomberg School of Public Health. Department of Mental H. Public mental health. New York, NY: Oxford University Press; 2012.

55. American Psychiatric Association. Depressive Disorders. Diagnostic and Statistical Manual of Mental Disorders DSM-5. fifth edition ed2013.

56. Ballard ED, Yarrington JS, Farmer CA, Lener MS, Kadriu B, Lally N, et al. Parsing the heterogeneity of depression: An exploratory factor analysis across commonly used depression rating scales. J Affect Disord. 2018;231:51-7.

57. Villas Boas GR, Boerngen de Lacerda R, Paes MM, Gubert P, Almeida WLdC, Rescia VC, et al. Molecular aspects of depression: A review from neurobiology to treatment. European Journal of Pharmacology. 2019;851:99-121.

58. Snoek FJ, Bremmer MA, Hermanns N. Constructs of depression and distress in diabetes: time for an appraisal. Lancet Diabetes Endocrinol. 2015;3(6):450-60.

59. Kessler RC, Bromet EJ. The epidemiology of depression across cultures. Annu Rev Public Health. 2013;34:119-38.

60. Hardeveld F, Spijker J, De Graaf R, Nolen WA, Beekman AT. Prevalence and predictors of recurrence of major depressive disorder in the adult population. Acta Psychiatr Scand. 2010;122(3):184-91.

61. Torpey DC, Klein DN. Chronic depression: update on classification and treatment. Current psychiatry reports. 2008;10(6):458-64.

62. Kaster MP, Moretti M, Cunha MP, Rodrigues ALS. Novel approaches for the management of depressive disorders. European Journal of Pharmacology. 2016;771:236-40.

63. Ferrari F, Villa RF. The Neurobiology of Depression: an Integrated Overview from Biological Theories to Clinical Evidence. Molecular Neurobiology. 2017;54(7):4847-65.

64. Global, regional, and national burden of 12 mental disorders in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. The Lancet Psychiatry. 2022;9(2):137-50.

65. Santomauro DF, Mantilla Herrera AM, Shadid J, Zheng P, Ashbaugh C, Pigott DM, et al. Global prevalence and burden of depressive and anxiety disorders in 204 countries and territories in 2020 due to the COVID-19 pandemic. The Lancet. 2021;398(10312):1700-12.

66. Dobson KG, Vigod SN, Mustard C, Smith PM. Trends in the prevalence of depression and anxiety disorders among Canadian working-age adults between 2000 and 2016. Health Rep. 2020;31(12):12-23.

67. Skorikov VB, VanderVoort DJ. Relationships Between The Underlying Constructs Of The Beck Depression Inventory And The Center For Epidemiological Studies Depression Scale. Educational and Psychological Measurement. 2003;63(2):319-35.

68. Binning JF. construct. Encyclopedia Britannica2016.

69. Malhi GS, Mann JJ. Depression. Lancet (London, England). 2018;392(10161):2299-312.

70. Thombs BD, Kwakkenbos L, Levis AW, Benedetti A. Addressing overestimation of the prevalence of depression based on self-report screening questionnaires. Canadian Medical Association Journal. 2018;190(2):E44.

71. Mojtabai R. Clinician-identified depression in community settings: concordance with structured-interview diagnoses. Psychother Psychosom. 2013;82(3):161-9.

72. Kupfer DJ. The pharmacological management of depression. Dialogues in clinical neuroscience. 2005;7(3):191-205.

73. Cuijpers P, van Straten A, Warmerdam L, Andersson G. Psychotherapy versus the combination of psychotherapy and pharmacotherapy in the treatment of depression: a meta-analysis. Depress Anxiety. 2009;26(3):279-88.

74. Yeoh SW, Holmes ACN, Saling MM, Everall IP, Nicoll AJ. Depression, fatigue and neurocognitive deficits in chronic hepatitis C. Hepatology international. 2018.

75. Ciesla JA, Roberts JE. Meta-analysis of the relationship between HIV infection and risk for depressive disorders. American Journal of Psychiatry. 2001;158(5):725-30.

76. Kalichman SC, Washington C, Kegler C, Grebler T, Kalichman MO, Cherry C, et al. Continued Substance Use Among People Living With HIV-Hepatitis-C Co-Infection and Receiving Antiretroviral Therapy. Subst Use Misuse. 2015;50(12):1536-43.

77. Yarlott L, Heald E, Forton D. Hepatitis C virus infection, and neurological and psychiatric disorders – A review. Journal of Advanced Research. 2017;8(2):139-48.

78. Liu CS, Adibfar A, Herrmann N, Gallagher D, Lanctôt KL. Evidence for Inflammation-Associated Depression. Curr Top Behav Neurosci. 2017;31:3-30.

79. Dantzer R, Capuron L. Inflammation-associated depression : evidence, mechanisms and implications: Springer; 2017.

80. Sockalingam S, Links PS, Abbey SE. Suicide risk in hepatitis C and during interferon-alpha therapy: a review and clinical update. Journal of Viral Hepatitis. 2011;18(3):153-60.

81. Fleming CA, Tumilty S, Murray JE, Nunes D. Challenges in the Treatment of Patients Coinfected with HIV and Hepatitis C Virus: Need for Team Care. Clinical Infectious Diseases. 2005;40(Supplement_5):S349-S54.

82. Sackey B, Shults JG, Moore TA, Rogers R, Mehvar M, King JG. Evaluating psychiatric outcomes associated with direct-acting antiviral treatment in veterans with hepatitis C infection. The mental health clinician. 2018;8(3):116-21.

83. Sundberg I, Lannergård A, Ramklint M, Cunningham JL. Direct-acting antiviral treatment in real world patients with hepatitis C not associated with psychiatric side effects: a prospective observational study. BMC Psychiatry. 2018;18(1):157.

84. Sanjeev Sockalingam, M.D. ,, Kathleen Sheehan, M.D., D.Phil. ,, Jordan J. Feld, M.D., M.P.H. ,, Hemant Shah, M.D., M.Sc.C.H. Psychiatric Care During Hepatitis C Treatment: The Changing Role of Psychiatrists in the Era of Direct-Acting Antivirals. American Journal of Psychiatry. 2015;172(6):512-6.

85. Gallach M, Vergara M, da Costa JP, Miquel M, Casas M, Sanchez-Delgado J, et al. Impact of treatment with direct-acting antivirals on anxiety and depression in chronic hepatitis C. PLoS One. 2018;13(12):e0208112.

86. Durcan E, Hatemi I, Sonsuz A, Canbakan B, Ozdemir S, Tuncer M. The effect of direct antiviral treatment on the depression, anxiety, fatigue and quality-of-life in chronic hepatitis C patients. European Journal of Gastroenterology & Hepatology. 2020;32(2):246-50.

87. Miarons M, Sánchez-Ulayar A, Sempere G, Marín S, Castellví JM. New directacting antivirals for hepatitis C treatment and neuropsychiatric symptoms in psychiatric risk groups. Eur J Hosp Pharm. 2019;26(3):135-9.

88. Egmond E, Mariño Z, Navines R, Oriolo G, Pla A, Bartres C, et al. Incidence of depression in patients with hepatitis C treated with direct-acting antivirals. Braz J Psychiatry. 2020;42(1):72-6.

89. Klein MB, Saeed S, Yang H, Cohen J, Conway B, Cooper C, et al. Cohort profile: the Canadian HIV-hepatitis C co-infection cohort study. International journal of epidemiology. 2010;39(5):1162-9.

90. Canadian Co-infection Cohort. About the Cohort 2022 [Available from: https://cocostudy.ca/the-study/about-the-cohort/.

91. Tyndall M. An emergency response to the opioid overdose crisis in Canada: a regulated opioid distribution program. Canadian Medical Association Journal. 2018;190(2):E35.

92. EuroQol--a new facility for the measurement of health-related quality of life. Health policy (Amsterdam, Netherlands). 1990;16(3):199-208.

93. Brooks R. EuroQol: the current state of play. Health policy (Amsterdam, Netherlands). 1996;37(1):53-72.

94. Cox J, Hamelin AM, McLinden T, Moodie EE, Anema A, Rollet-Kurhajec KC, et al. Food Insecurity in HIV-Hepatitis C Virus Co-infected Individuals in Canada: The Importance of Co-morbidities. AIDS and behavior. 2017;21(3):792-802.

95. Canadian Co-infection Cohort. Food security - CTN-264 2021 [Available from: https://cocostudy.ca/associated-studies/food-security-ctn-264/.

96. Aibibula W, Cox J, Hamelin A-M, Moodie EEM, Anema A, Klein Marina B, et al. Association between depressive symptoms, CD4 count and HIV viral suppression among HIV-HCV co-infected people. AIDS Care. 2018;30(5):643-9.

97. Canadian Community Health Survey, Cycle 2.2, Nutrition (2004): Income-Related Household Food Security in Canada. Health Canada, Ottawa, Canada; 2007.

98. Andresen EM, Malmgren JA, Carter WB, Patrick DL. Screening for depression in well older adults: evaluation of a short form of the CES-D (Center for Epidemiologic Studies Depression Scale). American journal of preventive medicine. 1994;10(2):77-84.

99. Zhang W, O'Brien N, Forrest JI, Salters KA, Patterson TL, Montaner JS, et al. Validating a shortened depression scale (10 item CES-D) among HIV-positive people in British Columbia, Canada. PLOS ONE. 2012;7(7):e40793.

100. Wong J, Motulsky A, Eguale T, Buckeridge DL, Abrahamowicz M, Tamblyn R. Treatment Indications for Antidepressants Prescribed in Primary Care in Quebec, Canada, 2006-2015. JAMA. 2016;315(20):2230-2.

101. Anagnostopoulos A, Ledergerber B, Jaccard R, Shaw SA, Stoeckle M, Bernasconi E, et al. Frequency of and Risk Factors for Depression among Participants in the Swiss HIV Cohort Study (SHCS). PLOS ONE. 2015;10(10):e0140943.

102. Dwyer DB, Falkai P, Koutsouleris N. Machine Learning Approaches for Clinical Psychology and Psychiatry. Annual Review of Clinical Psychology. 2018;14(1):91-118.

103. Shatte ABR, Hutchinson DM, Teague SJ. Machine learning in mental health: a scoping review of methods and applications. Psychological Medicine. 2019;49(9):1426-48.

104. Graham S, Depp C, Lee EE, Nebeker C, Tu X, Kim HC, et al. Artificial Intelligence for Mental Health and Mental Illnesses: an Overview. Current psychiatry reports. 2019;21(11):116.

105. Sau A, Bhakta I. Artificial Neural Network (ANN) Model to Predict Depression among Geriatric Population at a Slum in Kolkata, India. J Clin Diagn Res. 2017;11(5):Vc01-vc4.

106. Jin H, Wu S, Di Capua P. Development of a Clinical Forecasting Model to Predict Comorbid Depression Among Diabetes Patients and an Application in Depression Screening Policy Making. Prev Chronic Dis. 2015;12:E142.

107. Banack HR, Stokes A, Fox MP, Hovey KM, Cespedes Feliciano EM, LeBlanc ES, et al. Stratified Probabilistic Bias Analysis for Body Mass Index-related Exposure Misclassification in Postmenopausal Women. Epidemiology. 2018;29(5):604-13.

108. Spradling PR, Zhong Y, Moorman AC, Rupp LB, Lu M, Gordon SC, et al. Psychosocial Obstacles to Hepatitis C Treatment Initiation Among Patients in Care: A Hitch in the Cascade of Cure. Hepatology Communications. 2021;5(3):400-11.

109. Nguyen P, Vutien P, Hoang J, Trinh S, Le A, Yasukawa LA, et al. Barriers to care for chronic hepatitis C in the direct-acting antiviral era: a single-centre experience. BMJ Open Gastroenterol. 2017;4(1):e000181-e.

110. Malespin M, Harris C, Kanar O, Jackman K, Smotherman C, Johnston A, et al. Barriers to treatment of chronic hepatitis C with direct acting antivirals in an urban clinic. 2019;18(2):304-9.

111. Abdel Moez AT, El Hawary YA, Al Balakosy AM. Can successful treatment by direct-acting antivirals improve depression in chronic HCV patients? European Journal of Gastroenterology & Hepatology. 2021;33(5):727-30.

112. Khalil MA, Shousha HI, El-Nahaas SM, Negm MI, Kamal K, Madbouly NM. Depression in patients with chronic hepatitis-C treated with direct-acting antivirals: A realworld prospective observational study. Journal of Affective Disorders. 2021;282:126-32. 113. Lundgren L, Kattakuzhy S, Price A, Seamon C, Nelson A, Kohli A, et al., editors. Abstract number 695: Mental Health Impact of HCV Treatment in HIV/HCV Patients: DAA vs IFN-Based Therapy. Conference on Retroviruses and Opportunistic Infections (CROI); 2015; Seattle, Washington.

114. Carrasquillo O. Health Care Utilization. In: Gellman MD, Turner JR, editors.
Encyclopedia of Behavioral Medicine. New York, NY: Springer New York; 2013. p. 90910.

 Andersen R, Newman JF. Societal and Individual Determinants of Medical Care Utilization in the United States. Milbank Q. 2005;83(4):10.1111/j.468-0009.2005.00428.x.
 Norton BL, Park L, McGrath LJ, Proeschold Bell RJ, Muir AJ, Naggie S. Health care utilization in HIV-infected patients: assessing the burden of hepatitis C virus coinfection. AIDS patient care and STDs. 2012;26(9):541-5.

117. Ma H, Villalobos CF, St-Jean M, Eyawo O, Lavergne MR, Ti L, et al. The impact of HCV co-infection status on healthcare-related utilization among people living with HIV in British Columbia, Canada: a retrospective cohort study. BMC Health Serv Res. 2018;18(1):319.

118. Baum MK, Jayaweera DT, Duan R, Sales S, Lai S, Rafie C, et al. Quality of life, symptomatology and healthcare utilization in HIV/HCV co-infected drug users in Miami. J Addict Dis. 2008;27(2):37-48.

119. Yeung MW, Young J, Moodie E, Rollet-Kurhajec KC, Schwartzman K, Greenaway C, et al. Changes in quality of life, healthcare use, and substance use in HIV/hepatitis C coinfected patients after hepatitis C therapy: a prospective cohort study. HIV Clinical Trials. 2015;16(3):100-10.

120. Smith-Palmer J, Cerri K, Valentine W. Achieving sustained virologic response in hepatitis C: a systematic review of the clinical, economic and quality of life benefits. BMC Infectious Diseases. 2015;15(1):19.

121. Alberti A, Colombo M, Craxì A, Rizzetto M. The dilemma for patients with chronic hepatitis C: Treat now or warehouse? Digestive and Liver Disease. 2014;46(1):27-9.

122. Tadrous M, Mason K, Dodd, Z., Guyton M, Powis J, McCormack D, et al. Prescribing trends in direct-acting antivirals for the treatment of hepatitis C in Ontario, Canada. Canadian Liver Journal. 2021;4(1):51-8.

123. Amoako A, Ortiz-Paredes D, Engler K, Lebouché B, Klein MB. Patient and provider perceived barriers and facilitators to direct acting antiviral hepatitis C treatment among priority populations in high income countries: A knowledge synthesis. The International journal on drug policy. 2021;96:103247.

124. Ortiz-Paredes D, Amoako A, Lessard D, Engler K, Lebouché B, Klein M. Barriers and facilitators related to HCV treatment uptake among HIV coinfected populations in Canada: Patients and treatment provider perceptions. Canadian Liver Journal. 2021.

125. Singer M, Bulled N, Ostrach B, Mendenhall E. Syndemics and the biosocial conception of health. The Lancet. 2017;389(10072):941-50.

126. Tsai AC, Mendenhall E, Trostle JA, Kawachi I. Co-occurring epidemics, syndemics, and population health. The Lancet. 2017;389(10072):978-82.

127. Butt ZA, Shrestha N, Wong S, Kuo M, Gesink D, Gilbert M, et al. A syndemic approach to assess the effect of substance use and social disparities on the evolution of HIV/HCV infections in British Columbia. PLOS ONE. 2017;12(8):e0183609.

128. Reece R, Dugdale C, Touzard-Romo F, Noska A, Flanigan T, Rich JD. Care at the Crossroads: Navigating the HIV, HCV, and Substance Abuse Syndemic. Fed Pract. 2014;31:37S-40S.

129. Winetsky D, Burack D, Antoniou P, Garcia B, Gordon P, Scherer M. Psychosocial Factors and the Care Cascade for Hepatitis C Treatment Colocated at a Syringe Service Program. The Journal of infectious diseases. 2020;222(Suppl 5):S392-S400.

130. Stephens T, Joubert N. The economic burden of mental health problems in Canada. Chronic Dis Can. 2001;22(1):18-23.

131. Agboola AA, Esan OT, Afolabi OT, Soyinka TA, Oluwaranti AO, Adetayo A. Economic burden of the therapeutic management of mental illnesses and its effect on household purchasing power. PLOS ONE. 2018;13(9):e0202396.

132. Trautmann S, Rehm J, Wittchen H-U. The economic costs of mental disorders: Do our societies react appropriately to the burden of mental disorders? EMBO Rep. 2016;17(9):1245-9.

133. Desai AP, Mohan P, Nokes B, Sheth D, Knapp S, Boustani M, et al. Increasing Economic Burden in Hospitalized Patients With Cirrhosis: Analysis of a National Database. Clinical and Translational Gastroenterology. 2019;10(7).

134. Barber MJ, Gotham D, Khwairakpam G, Hill A. Price of a hepatitis C cure: Cost of production and current prices for direct-acting antivirals in 50 countries. Journal of Virus Eradication. 2020;6(3):100001.

135. Stevens ER, Nucifora KA, Hagan H, Jordan AE, Uyei J, Khan B, et al. Costeffectiveness of Direct Antiviral Agents for Hepatitis C Virus Infection and a Combined Intervention of Syringe Access and Medication-assisted Therapy for Opioid Use Disorders in an Injection Drug Use Population. Clinical infectious diseases : an official publication of the Infectious Diseases Society of America. 2020;70(12):2652-62.
136. Scott N, McBryde ES, Thompson A, Doyle JS, Hellard ME. Treatment scale-up to achieve global HCV incidence and mortality elimination targets: a cost-effectiveness model. Gut. 2017;66(8):1507.

137. Chhatwal J, He T, Hur C, Lopez-Olivo MA. Direct-Acting Antiviral Agents for Patients With Hepatitis C Virus Genotype 1 Infection Are Cost-Saving. Clin Gastroenterol Hepatol. 2017;15(6):827-37.e8.